ADULT LITERACY in Iowa

Results of the State Adult Literacy Survey



ADULT LITERACY in Iowa

Results of the State Adult Literacy Survey

Lynn B. Jenkins and Irwin S. Kirsch



FEBRUARY 1994

Ordering Information

For information on ordering copies of this book, contact:

John Hartwig, Consultant Iowa Department of Education Grimes State Office Building Des Moines, Iowa 50319-0146

or fax:

515-281-6544

or call: 515-281-3636

For information on adult literacy programs in Iowa, contact:

Iowa Department of Education Division of Community Colleges Grimes State Office Building Des Moines, Iowa 50319-0146 or call: Miriam Temple, 515-281-3640 Donald Wederquist, 515-281-3671

John Hartwig, 515-281-3636

Educational Testing Service is an equal opportunity, affirmative action employer. Educational Testing Service, ETS, and (fs) are registered trademarks of Educational Testing Service.



CONTENTS

Foreword	xiii
Preface	xv
Executive Summary	xvii
Introduction	xix
Profiles of Adult Literacy in Iowa	xx
Education and Training	xxiii
Employment, Economic Status, and Civic Responsibility	xxiv
Language Use and Literacy Practices	xxv
Reflections on the Results	xxvi
	Ţ
Defining and Measuring Literacy	5
Conducting the Survey	9
Reporting the Results	11
A Note on Interpretations	15
About This Report	17
Section I: Profiles of Adult Literacy in Iowa	19
Results for the Total Population in Iowa,	
the Midwest, and the Nation	22
Level 1	22
Level 2	25
Level 3	29
Level 4	29
Level 5	30
Results for Adults in Different Age Groups	31
Results for Adults Born in the United States	
and Those Born in Other Countries	35
Results for foreign-born adults by the number	
of years lived in the United States	38

Results for Adults in Different Racial/Ethnic Groups		41
Results for adults in different racial/ethnic groups, by country of birth		45
Results for Adults by the Number of Years		
Lived in Iowa		45
Results for Adults by Their Likelihood of Moving Out of		
Iowa in the Next Five Years		51
Results for Adults with Physical or Mental Conditions		52
Results for Males and Females		55
Summary		58
Section II: Education and Training		61
Educational Background		63
Highest level of education attained in the United States		64
Average years of schooling completed by various		
population groups		68
Level of education attained before coming to the United States		69
Participation in a GED or high school equivalency program		73
Current educational enrollment		76
Participation in Adult Education and Training		76
Enrollment in a basic skills program		78
Main reason for not enrolling in a basic skills program		79
Effect of state literacy rate on employers' relocation decisions		79
Opinion as to employers' obligation to provide literacy education to employees		82
Summary		83
Section III: Employment, Economic Status, and		OF
Employment		00
Labor force status		01
		00
Weeks worked		91
Foonomio Status		94
Weekhuwages		95
Annual household income		90
Sources of nonwage income		97
Povertu status	•••	97
Civic Responsibility	• • •	101
Voting		104
Summan	•••	105
Summary		107

.

iv Contents

Section IV: Language Use and Literacy Practices	109
Language Use	111
Language learned before starting school	111
Language usually spoken now	113
Language use in various contexts	114
Self-reported proficiency in the English language	115
Literacy Practices	117
Reliance on print and nonprint sources of information	117
Frequency of newspaper reading	119
Aspects of newspaper reading	119
Magazine and book reading practices	122
Frequency of library use	123
Amount of television watching	124
Personal and job-related use of prose materials	125
Personal and job-related use of documents	128
Personal use of mathematics	130
Summary	131
Section V: Interpreting the Literacy Scales	133
Building the Literacy Tasks	135
Defining the Literacy Levels	137
Interpreting the Literacy Levels	139
Prose literacy	139
Document literacy	150
Quantitative literacy	159
Estimating Performance Across the Literacy Levels	167
Appendices	171
Appendix A: Variable Definitions	173
Appendix B: Technical Notes	183
Appendix C: Participants in the Development Process	199
Acknowledgments	202

 $Contents \ldots . v$

30

.

Figures and Tables

36

Introduction

Table 1	Characteristics of the Iowa, Midwest, and National Samples	10
Figure 1	Difficulty Values of Selected Tasks Along the Prose, Document, and Quantitative Literacy Scales	13
Figure 2	Description of the Prose, Document, and Quantitative Literacy Levels	14

Section I: Profiles of Adult Literacy in Iowa

Figure 1.1	Prose, Document, and Quantitative Literacy Levels and Average Proficiencies: Results for Iowa, the Midwest, and the Nation	23
Table 1.1	Prose, Document, and Quantitative Literacy Levels and Average Proficiencies: Results for Iowa, the Midwest, and the Nation	24
Table 1.2P	Characteristics of the Population, by Prose Literacy Level: Results for Iowa	26
Table 1.2D	Characteristics of the Population, by Document Literacy Level: Results for Iowa	27
Table 1.2Q	Characteristics of the Population, by Quantitative Literacy Level: Results for Iowa	28
Table 1.3P	Prose Literacy Levels and Average Proficiencies, by Age: Results for Iowa, the Midwest, and the Nation	32
Table 1.3D	Document Literacy Levels and Average Proficiencies, by Age: Results for Iowa, the Midwest, and the Nation	33
Table 1.3Q	Quantitative Literacy Levels and Average Proficiencies, by Age: Results for Iowa, the Midwest, and the Nation	34
Table 1.4P	Prose Literacy Levels and Average Proficiencies, by Country of Birth: Results for Iowa, the Midwest, and the Nation	35

- Table 1.4D
 Document Literacy Levels and Average Proficiencies, by Country of Birth: Results for Iowa, the Midwest, and the Nation

vi Contents

Table 1.5P	Prose Literacy Levels and Average Proficiencies of Foreign-born Adults, by Years Lived in the United States: Results for Iowa, the Midwest, and the Nation	38
Table 1.5D	Document Literacy Levels and Average Proficiencies of Foreign-born Adults, by Years Lived in the United States: Results for Iowa, the Midwest, and the Nation	39
Table 1.5Q	Quantitative Literacy Levels and Average Proficiencies of Foreign-born Adults, by Years Lived in the United States: Results for Iowa, the Midwest, and the Nation	40
Table 1.6P	Prose Literacy Levels and Average Proficiencies, by Race/Ethnicity: Results for Iowa, the Midwest, and the Nation	42
Table 1.6D	Document Literacy Levels and Average Proficiencies, by Race/Ethnicity: Results for Iowa, the Midwest, and the Nation	43
Table 1.6Q	Quantitative Literacy Levels and Average Proficiencies, by Race/Ethnicity: Results for Iowa, the Midwest, and the Nation	44
Table 1.7	Country of Birth, by Race/Ethnicity: Results for Iowa, the Midwest, and the Nation	46
Table 1.8P	Average Prose Literacy Proficiencies, by Country of Birth and Race/Ethnicity: Results for Iowa, the Midwest, and the Nation	47
Table 1.8D	Average Document Literacy Proficiencies, by Country of Birth and Race/Ethnicity: Results for Iowa, the Midwest, and the Nation	48
Table 1.8Q	Average Quantitative Literacy Proficiencies, by Country of Birth and Race/Ethnicity: Results for Iowa, the Midwest, and the Nation	49
Table 1.9	Prose, Document, and Quantitative Literacy Levels and Average Proficiencies, by Years Lived in Iowa: Besults for Iowa	50
Table 1.10	Prose, Document, and Quantitative Literacy Levels and Average Proficiencies, by Likelihood of Moving Out of Iowa in the Next Five Years: Results for Iowa	51
Table 1.11P	Prose Literacy Levels and Average Proficiencies, by Disability Status: Results for Iowa, the Midwest, and the Nation	52
Table 1.11D	Document Literacy Levels and Average Proficiencies, by Disability Status: Results for Iowa, the Midwest, and the Nation	53
Table 1.11Q	Quantitative Literacy Levels and Average Proficiencies, by Disability Status: Results for Iowa, the Midwest, and the Nation	54

•-

1.80

18

Table 1.12P	Prose Literacy Levels and Average Proficiencies, by Sex: Results for Iowa, the Midwest, and the Nation	55
Table 1.12D	Document Literacy Levels and Average Proficiencies, by Sex: Results for Iowa, the Midwest, and the Nation	56
Table 1.12Q	Quantitative Literacy Levels and Average Proficiencies, by Sex: Results for Iowa, the Midwest, and the Nation	57

Section II: Education and Training

Table 2.1P	Prose Literacy Levels and Average Proficiencies, by Level of Education in the United States: Results for Iowa, the Midwest, and the Nation	65
Table 2.1D	Document Literacy Levels and Average Proficiencies, by Level of Education in the United States: Results for Iowa, the Midwest, and the Nation	66
Table 2.1Q	Quantitative Literacy Levels and Average Proficiencies, by Level of Education in the United States: Results for Iowa, the Midwest, and the Nation	67
Table 2.2	Average Years of Schooling Completed in the United States by Various Population Groups: Results for Iowa, the Midwest, and the Nation	70
Table 2.3	Average Literacy Proficiencies of Foreign-born Adults, by Highest Level of Education Attained Before Coming to the U.S.: Results for Iowa, the Midwest, and the Nation	72
Table 2.4	Highest Level of Education Attained Before Coming to the U.S., by Years Lived in the U.S.: Results for Iowa, the Midwest, and the Nation	73
Table 2.5	Average Literacy Proficiencies of Dropouts, by Participation in a GED Program: Results for Iowa, the Midwest, and the Nation	74
Table 2.6	Among School Dropouts, Participation in a GED Program, by Age: Results for Iowa	75
Table 2.7	Average Literacy Proficiencies, by Current Educational Enrollment and Goals: Results for Iowa, the Midwest, and the Nation	77
Table 2.8	Literacy Levels and Average Proficiencies, by Enrollment in a Basic Skills Program: Results for Iowa	78
Table 2.9	Literacy Levels and Average Proficiencies, by Reasons for Not Participating in a Basic Skills Program.	

for Not Participating in a Basic Skills Program: Results for Iowa

Table 2.10

80



Table 2.11	Literacy Levels and Average Proficiencies, by Opinion as to Employers' Obligation to Provide Literacy Education: Results for Iowa	82
Section III:	Employment, Economic Status, and Civic Responsibility	
Table 3.1P	Prose Literacy Levels and Average Proficiencies, by Labor Force Status: Results for Iowa, the Midwest, and the Nation	88
Table 3.1D	Document Literacy Levels and Average Proficiencies, by Labor Force Status: Results for Iowa, the Midwest, and the Nation	89
Table 3.1Q	Quantitative Literacy Levels and Average Proficiencies, by Labor Force Status: Results for Iowa, the Midwest, and the Nation	90
Table 3.2P	Prose Literacy Levels and Average Proficiencies, by Occupational Category: Results for Iowa, the Midwest, and the Nation	92
Table 3.2D	Document Literacy Levels and Average Proficiencies, by Occupational Category: Results for Iowa, the Midwest, and the Nation	93
Table 3.2Q	Quantitative Literacy Levels and Average Proficiencies, by Occupational Category: Results for Iowa, the Midwest, and the Nation	94
Table 3.3	Average Number of Weeks Worked in the Past 12 Months, by Literacy Level: Results for Iowa, the Midwest, and the Nation	95
Table 3.4	Median Weekly Wages, by Literacy Level: Results for Iowa, the Midwest, and the Nation	96
Table 3.5	Median Annual Household Income, by Literacy Level: Results for Iowa, the Midwest, and the Nation	98
Table 3.6P	Prose Literacy Levels and Average Proficiencies, by Sources of Nonwage Income and Support: Results for Iowa, the Midwest, and the Nation	99
Table 3.6D	Document Literacy Levels and Average Proficiencies, by Sources of Nonwage Income and Support: Results for Iowa, the Midwest, and the Nation	100
Table 3.6Q	Quantitative Literacy Levels and Average Proficiencies, by Sources of Nonwage Income and Support: Results for Iowa, the Midwest, and the Nation	101
Table 3.7P	Prose Literacy Levels and Average Proficiencies, by Poverty Status: Results for Iowa, the Midwest, and the Nation	102
Table 3.7D	Document Literacy Levels and Average Proficiencies, by Poverty Status: Results for Iowa, the Midwest, and the Nation	103

Contents ix

•-

Table 3.7Q	Quantitative Literacy Levels and Average Proficiencies, by Poverty Status: Results for Iowa, the Midwest, and the Nation
Table 3.8P	Prose Literacy Levels and Average Proficiencies, by Voting in Recent Elections: Results for Iowa, the Midwest, and the Nation
Table 3.8D	Document Literacy Levels and Average Proficiencies, by Voting in Recent Elections: Results for Iowa, the Midwest, and the Nation
Table 3.8Q	Quantitative Literacy Levels and Average Proficiencies, by Voting in Recent Elections: Results for Iowa, the Midwest, and the Nation
Section IV:	Language Use and Literacy Practices
Table 4.1	Average Literacy Proficiencies, by Languages Learned Before Starting School: Results for Iowa, the Midwest, and the Nation
Table 4.2	Average Literacy Proficiencies of Adults Who Learned a Non-English Language, by Language Spoken Now: Results for Iowa, the Midwest, and the Nation
Table 4.3	Use of English or Another Language in Various Contexts by Adults Who Learned a Non-English Language: Results for Iowa
Table 4.4	Average Literacy Proficiencies, by Self-reported English Literacy: Results for Iowa
Table 4.5	Average Literacy Proficiencies, by Reliance on Various Sources of Information About Current Events: Results for Iowa
Table 4.6	Average Literacy Proficiencies, by Frequency of Newspaper Reading: Results for Iowa
Table 4.7	Average Literacy Proficiencies of Adults Who Read the Newspaper Regularly, by Parts Read: Results for Iowa
Table 4.8	Average Literacy Proficiencies, by Magazine and Book Reading Practices: Results for Iowa
Table 4.9	Average Literacy Proficiencies, by Frequency of Library Use: Results for Iowa
T.L. 410	

- Table 4.10Average Literacy Proficiencies, by Amount of Television
Usually Watched Each Day: Results for Iowa125Table 4.11Types of Prose Materials Used for Personal or Job-related

x Contents

.

Table 4.12	Average Prose Proficiency, by Types of Prose Materials Used for Personal or Job-related Reading and Writing: Results for Iowa	127
Table 4.13	Types of Documents Used for Personal or Job-related Reading and Writing: Results for Iowa	128
Table 4.14	Average Document Proficiency, by Types of Documents Used for Personal or Job-related Reading and Writing: Results for Iowa	129
Table 4.15	Quantitative Literacy Levels and Average Proficiencies, by Frequency of Arithmetic or Mathematics Use: Results for Iowa	130

Section V: Interpreting the Literacy Scales

Figure 5.1	Probabilities of Successful Performance on Two Prose Tasks by Individuals at Selected Points on the Prose Scale	138
Figure 5.2	Average Probabilities of Successful Performance by Individuals with Selected Proficiency Scores on the Tasks in Each Literacy Level	168

Appendices

Table B.1	Response Rates for the National and State	
	Household Samples	191

Contents xi

•-

-

FOREWORD



• owa's aim is to develop well-educated and well-trained people who are equal to any in the world. In partnership with business and industry, education, and other entities, Iowa's literacy initiative established an important benchmark that can lead to well-paying jobs, a strong and diverse economy, and the full participation of all citizens in the benefits and opportunities Iowa provides. The key to achieving and maintaining a superior work force is first discovering where we are and then setting appropriate achievement goals. The findings presented in this report begin the journey; the commitment of Iowa's citizens will lead the way to a bright and productive future for the state.

> Ted Stilwill Acting Director Iowa Department of Education

> > Foreword xiii

PREFACE



erhaps never before have so many people from so many different sectors of our society been concerned about adult literacy. Numerous reports published in the last decade have indicated that a large portion of the United States population lacks adequate literacy skills, and many employers say they cannot find enough workers with the reading, writing, mathematical, and other competencies required in the workplace. Changing economic, demographic, and labor-market forces may exacerbate the problem in the future.

Whether the gap between our nation's literacy resources and its literacy needs will widen remains an open question; the evidence to prove or discredit such predictions is scarce. Many believe, however, that we must respond to the literacy challenge if we are to preserve our nation's economic vitality and ensure that every individual has a full range of opportunities for personal fulfillment and participation in society.

This view was reaffirmed at the historic education summit in Charlottesville, Virginia, where the nation's governors — including Governor Clinton — met with President Bush to establish a set of national education goals for the twenty-first century. As adopted in 1990 by members of the National Governors' Association, one of the six goals states:

> By the year 2000, every adult American will be literate and will possess the knowledge and skills necessary to compete in a global economy and exercise the rights and responsibilities of citizenship.

But how should this ambitious goal be pursued? In the past, whenever the population's skills were questioned, critics tended to focus on the educational system and insist that school reforms were needed if the nation were to escape serious social and economic consequences. Yet, many who need to improve their literacy skills have already left school. In fact, it is estimated that almost 80 percent of the work force for the year 2000 is already employed. Clearly, then, the schools alone cannot address our nation's literacy needs. A broader response is necessary.

Preface xv

To initiate such a response, we need more than localized reports or anecdotes from employers, public leaders, or the press. Accurate and detailed information is essential. Surprisingly, though, we lack answers to even the most basic questions, including how many individuals have limited literacy skills, who are they, and how severe are their problems.

In 1988, Congress asked the U.S. Department of Education to address this need by reporting on the nature and extent of adult literacy in this nation. In response, the Department's National Center for Education Statistics and Division of Adult Education and Literacy called for a national household survey of adult literacy. A contract was awarded to Educational Testing Service and a subcontract to Westat, Inc., to design and conduct the National Adult Literacy Survey. To give states an opportunity to explore the literacy skills of their own populations, all 50 states were invited to participate in the State Adult Literacy Survey, a concurrent study that would provide state-level results.

During the first eight months of 1992, trained staff visited thousands of households across the nation to interview adults age 16 and older. In Iowa, approximately 1,250 adults were surveyed, randomly selected to represent the 2.1 million adults in the state. In all, some 26,000 adults were surveyed, representing more than 191 million individuals nationwide. Each respondent was asked to spend about an hour performing diverse literacy tasks and answering questions about his or her background, education, work experiences, and reading practices.

Together, the results of the state and national surveys represent the most comprehensive database ever available on adult literacy in this nation. In an effort to disseminate the results to a wide and diverse audience, the findings are being issued in a series of reports. This report on the Iowa study profiles the literacy skills of state residents and explores connections between literacy and various factors. Reports are also available on each of the other 11 states that participated in the State Adult Literacy Survey.

Readers who seek additional information may wish to read Adult Literacy in America: A First Look at the Results of the National Adult Literacy Survey, or one of the forthcoming reports on literacy and education, literacy in the labor force, literacy among older adults and among prisoners, literacy and culture, and literacy practices.

Our hope is that this report and its companions will be a valuable resource

to those who are concerned about literacy in Iowa, and who are addressing the needs that are so plainly revealed in these data.





EXECUTIVE SUMMARY





EXECUTIVE SUMMARY

Adult Literacy in Iowa

his executive summary presents a portrait of adult literacy in Iowa based on the results of the State Adult Literacy Survey, an important research project in which 12 states assessed the literacy skills of their adult populations. The project, conducted in 1992, is a component of the National Adult Literacy Survey, a large-scale study funded by the U.S. Department of Education and administered by Educational Testing Service.

Introduction

In recent years, numerous studies have been conducted to evaluate programs that serve Iowa's adults, including the adult basic education population, GED graduates, and participants in community college continuing education programs.¹ The Iowa State Adult Literacy Survey adds to this important and growing body of research on adult literacy and education in this state.

Many past studies of adult literacy have tried to count the number of "illiterates" in this nation, thereby treating literacy as a condition that individuals either do or do not have. We believe that such efforts are inherently arbitrary and misleading. They are also damaging, in that they fail to acknowledge both the complexity of the literacy problem and the range of solutions needed to address it.

The Iowa State Adult Literacy Survey, like the National Adult Literacy Survey of which it is a part, is based on a different definition of literacy and therefore follows a different approach to measuring it. The aim of this survey is

Executive Summary xix

¹ Iowa Department of Education. (1991, April). A Study of the Impact of Iowa Community College Continuing Education Programs. Research on Adult Basic Education. (1990, February). What Motivates Adults to Participate in the Federal Adult Basic Education Program? CASAS. (1993, September). Iowa's Adult Basic Education Programs: A Survey of Learner Demographics and Preliminary Skill Levels. Iowa Department of Education. (1992, March). What Has Happened to Iowa's GED Graduates? Executive Summary. Iowa Department of Education. (1993, May). Performance Indicators of Program Quality for Iowa's Adult Basic Education Programs.

to characterize adults' literacy skills in English based on their performance on diverse tasks that reflect the types of materials and demands they encounter in their daily lives.

To gather information on the literacy skills of adults in Iowa, trained staff interviewed selected individuals aged 16 and older during the first eight months of 1992. These participants were randomly chosen to represent the adult population in the state as a whole. In total, approximately 1,250 adults in Iowa were surveyed, representing approximately 2.1 million adults statewide.

Each survey participant was asked to spend approximately an hour responding to a series of varied literacy tasks as well as questions about his or her demographic characteristics, educational background, employment, income, reading practices, and other areas related to literacy. Based on their responses to the survey tasks, adults received proficiency scores along three scales, each ranging from 0 to 500. The score points along these scales reflect varying degrees of skill in prose, document, and quantitative literacy. To provide a way to examine the distribution of performance within various subpopulations of interest, five levels of proficiency were defined along each scale: Level 1 (0 to 225), Level 2 (226 to 275), Level 3 (276 to 325), Level 4 (326 to 375), and Level 5 (376 to 500).

The full report offers a comprehensive look at the results of the Iowa survey. It describes the average literacy proficiencies and the levels of proficiency demonstrated by adults in this state, compared with individuals in the region and nation, and explores connections between literacy and an array of variables. Some of the major findings are highlighted in the pages that follow.

Profiles of Adult Literacy in Iowa

• Fourteen to 16 percent of the adults in Iowa demonstrated skills in the lowest level of prose, document, and quantitative proficiencies (Level 1). Though all adults in this level displayed limited skills, their characteristics are diverse. Many adults in this level were successful in performing simple, routine tasks involving brief and uncomplicated texts and documents. For example, they were able to total the entries on a deposit slip, locate the time or place of a meeting on a form, and identify a piece of specific information in a brief news article. Others did not perform these types of

tasks successfully, however, and some had such limited skills that they were unable to respond to much of the survey.

xx Executive Summary

- The composition of the Level 1 population differs in some important respects from the state population as a whole. For example, 38 percent of the Iowa residents who performed in Level 1 on the quantitative literacy scale had zero to eight years of education, compared with 7 percent of adults statewide. Respondents who demonstrated skills in Level 1 were much less likely to have completed high school or a General Educational Development (GED) certificate or attended a postsecondary institution (33 percent) than adults in the state population as a whole (77 percent). Half the Iowa respondents in Level 1 were age 65 or older, and almost 40 percent had physical or mental conditions that kept them from participating fully in work, school, housework, or other activities.
- Twenty-two to 27 percent of the Iowa respondents performed in the next higher level of proficiency (Level 2) on each literacy scale. While their skills were more varied than those of individuals in Level 1, their repertoires were still quite limited. They were generally able to locate information in text, to make low-level inferences using printed materials, and to integrate easily identifiable pieces of information. Further, they demonstrated the ability to perform quantitative tasks that involve a single operation where the numbers are either stated or can be easily found in text. For example, adults in this level were able to calculate the total cost of a purchase or determine the difference in price between two items. They could also locate a particular intersection on a street map and enter background information on a simple form.
- Individuals in Levels 1 and 2 were sometimes, but not consistently, able to respond correctly to the more challenging literacy tasks in the assessment —

those requiring higher level reading and problem-solving skills. In particular, they appeared to have considerable difficulty with tasks that required them to integrate or synthesize information from complex or lengthy texts or to perform quantitative tasks in which the individual had to set up the problem and then carry out two or more sequential operations.

• Thirty-six to 37 percent of the survey participants in Iowa performed in Level 3 on each literacy scale. Respondents with skills in this level on the prose and document scales integrated information from relatively long or dense text or from documents. Those in Level 3 on the quantitative scale demonstrated an ability to determine the appropriate arithmetic operation based on information contained in the directive, and to identify the quantities needed to perform that operation.

- Twenty-one to 27 percent of the respondents in Iowa scored in the two highest levels of prose, document, and quantitative literacy (Levels 4 and 5). These adults consistently demonstrated the ability to perform the most challenging tasks in this assessment, many of which involved long and complex documents and text passages. They were more likely than individuals in the state population as a whole to have completed high school or a GED or to have attended a postsecondary institution.
- The average prose, document, and quantitative proficiencies of adults in Iowa were comparable to those of adults living in the Midwest region and were significantly (13 to 16 points) higher than those of adults nationwide. In all three populations — the state, region, and nation — average scores were either in the high end of the Level 2 range (226 to 275) or the low end of the Level 3 range (276 to 325).
- Older adults were more likely than middle-aged and younger adults to demonstrate limited literacy skills. On the prose scale, for example, average scores rise from 290 among Iowa's 16- to 18-year-olds to 303 among 35- to 44-year-olds before declining across the older age groups (to 275 among 55- to 64-year-olds).
- The vast majority of Iowa residents were born in the United States or one of its territories. In the national population, native-born adults performed far better in the assessment, on average, than did individuals born outside the United States. Foreign-born adults who had lived in this country for more than a decade outperformed more recent immigrants.
- Ninety-six percent of the Iowa population is White. The numbers of adults in other racial/ethnic groups are too small to provide reliable proficiency estimates. Nationwide, however, African American and Latino adults were more likely than White adults to perform in the lowest two literacy levels and less likely to attain the two highest levels. The average proficiencies of Latino individuals who were born in this country were higher than those of African American adults.
- Approximately three-quarters of the adults in Iowa reported having lived in the state for more than 20 years. There are no significant differences in literary shills are supported by the state of th

literacy skills, on average, among adults who had lived in Iowa for varying lengths of time. Nearly three-quarters of the state's adults said it was unlikely that they would move out of the state in the next five years, while 18 percent reported that it was somewhat likely and 10 percent said it was likely. Again, there are no significant differences in performance between adults who believed they would move out of the state and those who did not.

xxii Executive Summary

- Iowa residents who reported having physical or mental conditions that keep them from participating fully in work or other activities were more likely than adults in the population as a whole to perform in the lowest levels on each literacy scale and less likely to reach the highest levels.
- In the Iowa population, there were no significant differences in the average literacy scores of men and women. Nationwide, however, men displayed higher average document and quantitative proficiencies than women.

Education and Training

- Iowa residents with relatively few years of education demonstrated lower average literacy proficiencies than those who completed high school or some postsecondary education. In fact, scores rise steadily across the entire range of education levels. The average prose proficiency of those who completed
 9 to 12 years of schooling was 242, for example, compared with 283 for those who earned a high school diploma but went no further, and 333 for those who had completed a four-year degree.
- There were no statistically significant differences between the average literacy scores of GED and high school graduates.
- Differences in the average years of schooling completed by adults in various subpopulations tend to parallel the observed differences in literacy proficiencies. Though not all the differences are statistically significant, average years of schooling tend to increase from the youngest age group to

the middle groups and then to decline across the older groups. Further, the more education respondents' parents had completed, the more education they themselves were likely to have completed — and the higher their literacy proficiencies were likely to be.

- Roughly one-third of the school dropouts in Iowa reported having participated in a GED or high school equivalency program. On each literacy scale, the average scores of program participants were approximately 50 points higher than those of dropouts who had not taken part in a GED program. The vast majority of program participants in Iowa were between the ages of 25 and 54.
- Eleven percent of the adults in Iowa were enrolled in school or college at the time of the survey, and they had higher literacy proficiencies, on average, than adults who were not enrolled in an academic program. Thirty-eight percent of those enrolled in a program stated that their goal was a four-year college degree.

- Six percent of the survey respondents in Iowa said they were currently or previously enrolled in a course to improve their basic skills. These individuals performed as well as those who had not enrolled in such a course.
- Thirty-eight percent of the Iowa adults said they would not enroll in a basic skills program because they did not think they needed to improve their skills. Their average scores were higher than those of adults who cited other reasons for not enrolling. One-quarter of the Iowa respondents said they would not enroll because they did not have time, and another 12 percent said they lacked information about basic skills programs.
- Three-quarters of Iowa's survey participants agreed with the view that a state's literacy rate affects an out-of-state employer's decision to establish a location there. Their scores were, on average, higher than those of adults who disagreed. Fifty-eight percent of Iowa's adults believed that employers are obligated to provide literacy education to employees who need it. They performed similarly to adults who did not share this view.

Employment, Economic Status, and Civic Responsibility

- Employed adults were less likely than adults who were unemployed or out of the labor force to perform in the lowest levels on each literacy scale and more likely to attain the highest levels. Across the three scales, 25 to 33 percent of the employed adults in Iowa performed in Levels 1 and 2, compared with 45 percent of the unemployed adults and roughly two-thirds of respondents who were out of the labor force. Conversely, employed adults were more likely than unemployed adults and those not in the labor force to attain Levels 4 and 5.
- Iowa residents who reported being in professional, technical, or managerial positions in their current or most recent jobs had higher average literacy scores than those in other types of occupations. On the prose scale, for example, they had an average proficiency score of 330, compared with scores of 309 for those in sales or clerical positions, 286 for those in craft or service occupations, and 276 for those in labor, assembly, fishing, or farming positions.
- On each literacy scale, adults who performed in the higher levels had worked

more weeks in the past year, on average, than individuals in the lower levels. Among Iowa residents, those in the three highest literacy levels reported working an average of 37 to 45 weeks in the past year, compared with only 13 to 14 weeks for individuals performing in Level 1, and 27 to 31 weeks for those in Level 2.

xxiv Executive Summary

- Across the scales, Iowa adults with proficiencies in Levels 1 and 2 reported median weekly earnings of \$228 to \$261. In contrast, those in Level 4 earned about \$391 to \$419, while those in Level 5 earned between \$504 and \$550 each week. Similarly, the median annual household income reported by adults in the highest proficiency levels was far higher than that of adults in the lowest levels.
- Approximately half the Iowa residents who were classified as either poor or near poor demonstrated skills in the two lowest levels on each literacy scale; in contrast, 25 to 31 percent of those designated not poor performed in these levels. As a result, the average literacy scores of poor and near poor adults are considerably lower than the scores of adults who were not in poverty.
- Among Iowa residents, voting practices appear to be related to literacy proficiency. On all three scales, the average literacy proficiencies of state residents who said they had voted in a recent election are higher than those of nonvoters.

Language Use and Literacy Practices

 The vast majority of Iowa residents (96 percent) reported that English was the only language they learned before beginning school. Nationwide, individuals who learned a language other than English as a child, either in addition to or in place of English, displayed lower average proficiencies than adults who reported having learned only English.

- Virtually all survey respondents in Iowa (98 to 99 percent) said they understand, speak, and read English well or very well; a slightly smaller proportion described themselves as writing (96 percent) well or very well. In each dimension of literacy, the average proficiencies of adults who said they do not write English well are approximately 60 points lower than those of individuals who said they write well or very well.
- Ninety-seven percent of the survey respondents in Iowa reported getting some or a lot of information about current events, public affairs, or government from nonprint media — that is, from television or radio. A smaller percentage (86 percent) said they get much of their information from print media, such as newspapers or magazines. Those who get some or a lot of information from print media earned higher average scores in the assessment than those who do not.

Executive Summary xxv

- Slightly more than half (56 percent) of the adults in the state said they read a newspaper every day, while another 35 percent said they do so at least once a week. Four percent reported never reading a newspaper. There are no significant differences in literacy proficiency between newspaper readers and nonreaders in Iowa.
- Fourteen percent of the Iowa respondents said they do not read any magazines in English on a regular basis. Their average literacy scores were considerably lower than the scores of those who read a few magazines regularly. Similarly, 17 percent of the adults in Iowa said they had not read any books in English in the past six months, and their average scores were considerably lower than those of adults who had read at least one book.
- One-third of the adults in Iowa reported that they never use a library, while 19 percent said they do so monthly and 20 percent said they do so either weekly or daily. In general, individuals who reported frequent use of the library outperformed less frequent users.
- Virtually all (98 percent) of the adults in Iowa reported watching some television every day, although 23 percent said they spend no more than an hour on this activity. Approximately one-third of the state's residents reported watching four or more hours of television each day. Individuals who watch the most television demonstrated lower average proficiencies than individuals who watch relatively little television.
- There are very large differences in prose proficiency between Iowa residents who read and write prose frequently, either for their personal use or for their jobs, and those who do not. Similarly, the average document proficiencies of individuals who use documents at least a few times a week are far higher than the scores of individuals who do not use these materials often. Finally, adults who said they frequently use mathematics tend to display better quantitative skills than those who rarely or never do so.

Reflections on the Results

In reflecting on the results of this study, many readers will undoubtedly seek an answer to a fundamental question: Are the outcomes satisfactory? That is, are

the distributions of prose, document, and quantitative proficiency observed in this survey adequate to ensure individual opportunities for all adults, to increase worker productivity, or to strengthen America's competitiveness around the world?

xxvi Executive Summary

Because it is impossible to say precisely what literacy skills are essential for individuals to succeed in this or any other society, the results of the State and National Adult Literacy Surveys provide no firm answers to such questions. As the authors examined the survey data and deliberated on the results with members of the advisory committees, however, several observations and concerns emerged.

Perhaps the most salient finding of this study is that nationwide, surprisingly large percentages of adults performed in the lowest levels (Levels 1 and 2) of prose, document, and quantitative literacy. In and of itself, this may not indicate a serious problem. After all, the majority of adults who demonstrated limited skills described themselves as reading or writing English well, and relatively few said they get a lot of assistance from others in performing everyday literacy tasks. Perhaps these individuals are able to meet most of the literacy demands they encounter currently at work, at home, and in their communities.

Yet, some argue that lower literacy skills mean a lower quality of life and more limited employment opportunities. As noted in a recent report from the American Society for Training and Development, "The association between skills and opportunity for individual Americans is powerful and growing. . . . Individuals with poor skills do not have much to bargain with; they are condemned to low earnings and limited choices."²

The data from this survey appear to support such views. On each of the literacy scales, adults who were unemployed or out of the labor force demonstrated far more limited skills than those who were employed, and those who earned low wages displayed far lower proficiencies than those who earned high wages. Adults who rarely or never read displayed lower average proficiencies than those who were at least occasional readers. Moreover, the average literacy scores of individuals who received food stamps and who were poor or near poor were much lower than those of their more affluent peers.

Literacy is not the only factor that contributes to how we live our lives, however. Some adults who were out of work or who earned low wages performed relatively well in the assessment, while some full-time workers or adults who earned high wages did relatively poorly. Thus, having advanced literacy skills is not necessarily associated with individual opportunities.

Still, literacy can be thought of as a currency in this society. Just as adults with little money have difficulty meeting their basic needs, those with limited literacy skills are likely to find it more challenging to pursue their goals — whether these involve job advancement, consumer decision making,

Executive Summary xxvii

² A.J. Carnevale and L.J. Gainer. (1989). The Learning Enterprise. Washington, D.C.: U.S. Department of Labor, Employment and Training Administration.

citizenship, or other aspects of their lives. Even if adults who performed in the lowest literacy levels are not experiencing difficulties at present, they may be at risk as the nation's economy and social fabric continue to change.

Beyond these personal consequences, what implications are there for society when so many individuals display limited skills? The answer to this question is elusive. Still, it seems apparent that a nation in which large numbers of citizens display limited literacy skills has fewer resources with which to meet its goals and objectives, whether these are social, political, civic, or economic.

If large percentages of adults had to do little more than sign their names on forms or locate single facts in newspapers or tables, then the levels of literacy seen in this survey might not warrant concern. We live in a nation, however, where both the volume and variety of written information are growing and where increasing numbers of citizens are expected to be able to read, understand, and use these materials.

Historians remind us that during the last 200 years, our nation's literacy skills have increased dramatically in response to new requirements and expanded opportunities for social and economic growth. Today we are a better educated and more literate society than at any time in our history.³ Yet, there have also been periods of imbalance — times when demands seemed to surpass levels of attainment.

In recent years, our society has grown more technologically advanced and the roles of formal institutions have expanded. As this has occurred, many have argued that there is a greater need for all individuals to become more literate and for a larger proportion to develop advanced skills.⁴ Growing numbers of individuals are expected to be able to attend to multiple features of information in lengthy and sometimes complex displays, to compare and contrast information, to integrate information from various parts of a text or document, to generate ideas and information based on what they read, and to apply arithmetic operations sequentially to solve a problem.

The results from this and other surveys, however, indicate that many adults do not demonstrate these levels of proficiency. Further, the continuing process of demographic, social, and economic change within this country could lead to a more divided society along both racial and socioeconomic lines.

³L.C. Stedman and C.F. Kaestle. (1991). "Literacy and Reading Performance in the United States from 1880 to the Present," in C.F. Kaestle et al., *Literacy in the United States: Readers and Reading Since 1880.* New Haven, CT: Yale University Press. T. Snyder (ed.). (1993). 120 Years of American Education: A Statistical Portrait. Washington, DC: National Center for Education Statistics.

⁴ U.S. Department of Labor. (1992, April). Learning a Living: A Blueprint for High Performance. Washington, DC: The Secretary's Commission on Achieving Necessary Skills (SCANS). R.L. Venezky, C.F. Kaestle, and A. Sum. (1987, January). The Subtle Danger: Reflections on the Literacy Abilities of America's Young Adults. Princeton, NJ: Educational Testing Service.

xxviii Executive Summary

Already there is evidence of a widening division. According to the report *America's Choice: High Skills or Low Wages!*, over the past 15 years the gap in earnings between professionals and clerical workers has grown from 47 to 86 percent, while the gap between white collar workers and skilled tradespeople has risen from 2 to 37 percent. At the same time, earnings for college educated males 24 to 34 years of age have increased by 10 percent, while earnings for those with high school diplomas have declined by 9 percent. Moreover, the poverty rate for African American families is nearly three times that for White families.⁵ One child in five is born into poverty, and for minority populations this rate approaches one in two.

In 1990, President Bush and the nation's governors, including Governor Clinton, adopted the goal that *all* of America's adults be literate by the year 2000. The responsibility for meeting this objective must, in the end, be shared among individuals, groups, and organizations throughout our society. Programs that serve adult learners cannot be expected to solve the literacy problem alone, and neither can the schools. Other institutions — ranging from the largest and most complex government agency, to large and small businesses, to the family — all have a role to play in ensuring that adults who need or wish to improve their literacy skills have the opportunity to do so. It is also important that individuals themselves come to realize the value of literacy in their lives and to recognize the benefits associated with having better skills. Only then will more adults in this nation develop the literacy resources they need to function in society, to achieve their goals, and to develop their knowledge and potential.

⁵ National Center on Education and the Economy. (1990, June). America's Choice: High Skills or Low Wages! The Report of The Commission on the Skills of the American Workforce. p. 20.

Executive Summary xxix

INTRODUCTION



INTRODUCTION



As a nation, we place a high value on literacy. This was affirmed at the historic education summit in Charlottesville, Virginia, when the nation's governors — including Governor Clinton — met with President Bush to define a set of national education goals that would guide the country into the twenty-first century. As adopted in 1990, the fifth goal states:

By the year 2000, every adult American will be literate and will possess the knowledge and skills necessary to compete in a global economy and exercise the rights and responsibilities of citizenship.

To gauge our progress toward meeting this ambitious goal, it is necessary first to have accurate and detailed information about our current status. The National Adult Literacy Survey (NALS) and the concurrent State Adult Literacy Survey (SALS) were designed to provide this essential information on the literacy skills of America's adults. The surveys grew out of the Adult Education Amendments of 1988, in which the U.S. Congress called on the Department of Education to report on the definition of literacy and on the nature and extent of literacy among America's adults. In response, the Department's National Center for Education Statistics (NCES) and the Division of Adult Education and Literacy planned a national household survey of adult literacy. In September 1989, NCES awarded a four-year contract to Educational Testing Service to design and administer the survey and to analyze and report the results. A subcontract was given to Westat, Inc., for sampling and field operations. Over the next few years, an extensive process was undertaken to develop a working definition of literacy for the study, construct survey instruments that would measure adults' proficiencies and gather important background information, analyze the survey data, and report on the results.

While the National Adult Literacy Survey would, by design, provide information on the literacy skills of America's adults nationwide, and on the performance of those living in various regions of the country, it would not enable individual states to describe the literacy proficiencies of adults living

Introduction 3

within their borders. Accordingly, each of the 50 states was invited to participate in a concurrent project, the State Adult Literacy Survey, designed to provide state-level results comparable to those of the national survey. Many states expressed an interest, and the following decided to participate in the concurrent study.

California	Louisiana	Pennsylvania
Illinois	New Jersey	Texas
Indiana	New York	Washington
Iowa	Ohio	

To permit comparisons of the state and national results, the survey instruments administered to the state and national samples were virtually identical; the only difference was that the state survey instruments included a small number of additional background questions. Further, the data for the national and state surveys were gathered at the same time. Florida also participated in the survey, but its data collection was unavoidably delayed until 1993.

During the first eight months of 1992, approximately 1,000 adults age 16 to 64 were surveyed in each state that participated in the State Adult Literacy Survey, in addition to the more than 14,000 adults age 16 and older who were surveyed nationwide as part of the National Adult Literacy Survey. In total, then, more than 26,000 individuals across the country participated in the state and national studies. Respondents spent, on average, more than an hour performing a series of diverse literacy tasks and answering a set of background questions on various topics. The results offer the most detailed portrait ever available of adult literacy in the United States.

In recent years, numerous studies have been conducted to evaluate programs that serve Iowa's adults, including the adult basic education population, GED graduates, and participants in community college continuing education programs.¹ The Iowa State Adult Literacy Survey adds to this important and growing body of research on adult literacy and education in this state.

The remainder of this introduction summarizes the definition of literacy for the national and state surveys, the framework used in designing the survey instruments, the populations assessed, the survey administration, and the

methods for reporting the results.

4.... Introduction

¹ Iowa Department of Education. (1991, April). A Study of the Impact of Iowa Community College Continuing Education Programs. Research on Adult Basic Education. (1990, February). What Motivates Adults to Participate in the Federal Adult Basic Education Program? CASAS. (1993, September). Iowa's Adult Basic Education Programs: A Survey of Learner Demographics and Preliminary Skill Levels. Iowa Department of Education. (1992, March). What Has Happened to Iowa's GED Graduates? Executive Summary. Iowa Department of Education. (1993, May). Performance Indicators of Program Quality for Iowa's Adult Basic Education Programs.

Defining and Measuring Literacy

The plan for developing and conducting the national and state surveys was guided by a panel of experts from business and industry, labor, government, research, and adult education. This Literacy Definition Committee worked with Educational Testing Service staff to prepare a definition of literacy that would guide the development of the assessment objectives as well as the construction and selection of assessment tasks. A second panel, the Technical Review Committee, was formed to help ensure the soundness of the assessment design, the quality of the data collected, the integrity of the analyses conducted, and the appropriateness of the interpretations of the final results. In addition, representatives from each of the states that participated in the State Adult Literacy Survey were invited to attend a series of meetings convened to guide the collection, analysis, and interpretation of data from the state surveys. These representatives were kept informed about the status of the survey through a series of newsletters prepared by Educational Testing Service staff.

The definition of literacy that guided the National Adult Literacy Survey and State Adult Literacy Survey was rooted in two preceding literacy studies funded by the federal government and conducted by Educational Testing Service: a 1985 household survey of the literacy skills of 21- to 25-year-olds, funded by the U.S. Department of Education², and a 1989-90 survey of the literacy proficiencies of job seekers, funded by the U.S. Department of Labor.³ The national panel of experts assembled to construct a definition of literacy for the young adult survey rejected the types of arbitrary standards — such as signing one's name, completing five years of school, or scoring at a particular grade level on a school-based measure of reading achievement — that have long been used to make judgments about adults' literacy skills. Through a consensus process, this panel drafted the following definition of literacy for the young adult survey:

Using printed and written information to function in society, to achieve one's goals, and to develop one's knowledge and potential.

Unlike traditional definitions of literacy, which focused on decoding and comprehension, this definition encompasses a broad range of skills that adults use in accomplishing the many different types of literacy tasks associated with

Introduction 5

²I.S. Kirsch and A. Jungeblut. (1986). Literacy: Profiles of America's Young Adults. Princeton, NJ: Educational Testing Service.

³I.S. Kirsch, A. Jungeblut, and A. Campbell. (1992). Beyond the School Doors: The Literacy Needs of Job Seekers Served by the U.S. Department of Labor. Princeton, NJ: Educational Testing Service.

work, home, and community contexts. This perspective is shaping not only adult literacy assessment, but also policy, as seen in the National Literacy Act of 1991, which defined literacy as "an individual's ability to read, write, and speak in English and compute and solve problems at levels of proficiency necessary to function on the job and in society, to achieve one's goals, and to develop one's knowledge and potential."

The definition of literacy from the young adult survey was adopted by the panel that guided the development of the 1989-90 survey of job seekers, and it also provided the starting point for the discussions of the National Adult Literacy Survey's Literacy Definition Committee. This committee agreed that expressing the literacy proficiencies of adults in school-based terms or gradelevel scores is inappropriate. In addition, while the committee recognized the importance of teamwork skills, interpersonal skills, and communication skills for functioning in various contexts, such as the workplace, it decided that these areas would not be addressed in this survey.

Further, the committee endorsed the notion that literacy is neither a single skill suited to all types of texts, nor an infinite number of skills, each associated with a given type of text or material. Rather, as suggested by the results of the young adult and job-seeker surveys, an ordered set of skills appears to be called into play to accomplish diverse types of tasks. Given this perspective, the Literacy Definition Committee agreed to adopt not only the definition of literacy that was used in the previous surveys, but also the three scales developed as part of those efforts:

Prose literacy — the knowledge and skills needed to understand and use information from texts that include editorials, news stories, poems, and fiction; for example, finding a piece of information in a newspaper article, interpreting instructions from a warranty, inferring a theme from a poem, or contrasting views expressed in editorials.

Document literacy — the knowledge and skills required to locate and use information contained in materials that include job applications, payroll forms, transportation schedules, maps, tables, and graphs; for example, locating a particular intersection on a street map, using a schedule to choose the appropriate bus, or entering information on an application form.

Quantitative literacy — the knowledge and skills required to apply arithmetic operations, either alone or sequentially, using numbers embedded in printed materials; for example, balancing a checkbook, figuring out a tip, completing an order form, or determining the amount of interest from a loan advertisement.

6.... Introduction

The literacy scales, built initially to report on the results of the young adult survey and augmented in the survey of job seekers, provide a useful way to organize a broad array of tasks and to report the assessment results. They represent a substantial improvement over traditional approaches to literacy assessment, which have tended to report on performance in terms of single tasks or to combine the results from diverse tasks into a single, conglomerate score. Such a score fosters the simplistic notion that "literates" and "illiterates" can be neatly distinguished from one another based on a single cutpoint on a single scale. The literacy scales, on the other hand, make it possible to profile the various types and levels of literacy among different subgroups in our society. In so doing, they help us to understand the diverse informationprocessing skills associated with the broad range of printed and written materials that adults read and their many purposes for reading them.

In adopting the three scales for use in this survey, the committee's aim was not to establish a single national standard for literacy. Rather, it was to provide an interpretive scheme that would enable levels of prose, document, and quantitative performance to be identified and allow descriptions of the knowledge and skills associated with each level to be developed.

The Literacy Definition Committee for the National Adult Literacy Survey recommended that a new set of literacy tasks be developed to enhance the literacy scales for this survey, without compromising the ability to compare the results with those of the young adult and job-seeker surveys. The new tasks, like those administered in the earlier studies, were open-ended. They simulated real-life literacy demands, measured a broad range of informationprocessing skills, and covered a wide variety of contexts. There was a greater emphasis on tasks that required brief written and/or oral responses and that asked respondents to describe how they would set up and solve a problem. Finally, some of the new quantitative tasks developed for this survey required respondents to use a simple four-function calculator. In all, approximately 110 new assessment tasks were field tested, and 81 of these were selected for inclusion in the survey, in addition to 85 tasks that were administered in both the young adult and job-seeker assessments. The administration of a common set of simulation tasks in each of the three literacy surveys makes it possible to compare results across time (that is, from the 1985, 1989-90, and 1992 surveys) and across population groups.

A large number of tasks had to be administered in the current survey to ensure the broadest possible coverage of the literacy domains specified. Yet, no individual could be expected to respond to the entire set of 166 simulation tasks. Accordingly, the survey design dictated that each respondent would receive a subset of the total pool of literacy tasks, while at the same time ensuring that each task was administered to a nationally representative sample of adults. The literacy tasks were assigned to sections that could be completed in about 15 minutes, and these sections were then compiled into booklets, each of which could be completed in about 45 minutes. During a personal interview, each survey respondent was asked to complete one booklet of assessment tasks. All tasks were administered in English only, since this was a survey of adults' literacy skills in the English language — not of their proficiencies in other languages.

In addition to performing the literacy tasks, each participant was asked to spend approximately 20 minutes answering a series of questions about his or her background and characteristics. Two versions of this questionnaire were administered, one in English and one in Spanish. Major areas explored included:

- background and demographics country of birth, languages spoken or read, access to reading materials, size of household, educational attainment of parents, age, race/ethnicity, and marital status
- education highest grade completed in school, current aspirations, participation in adult education classes, and education received outside the United States
- labor market experiences employment status, recent labor market experiences, and occupation
- income personal as well as household
- activities voting, hours spent watching television, frequency and content of newspaper reading, and use of literacy skills for work and leisure

This core set of background questions was administered to all adults in the state and national samples. However, each state that participated in the State Adult Literacy Survey was invited to develop up to five additional background questions that would be administered to its respondents, to gather information of particular interest to state decision makers. The supplementary background questions included in the Iowa survey addressed a range of topics:

- number of years lived in the state
- likelihood of moving out of the state in the next five years
- reasons for not taking part in a basic skills program

perceived impact of state literacy rate on out-of-state employers

opinion as to employers' obligation provide literacy education

8.... Introduction

These background data make it possible to investigate the extent to which particular characteristics, experiences, and behaviors are associated with demonstrated performance on each of the three literacy scales.⁴

Conducting the Survey

It was important to administer the State and National Adult Literacy Surveys at the same time to ensure that the results would be comparable. Accordingly, the surveys were carried out during the first eight months of 1992, except in Florida, where the data collection was unavoidably delayed until 1993. The survey instruments were administered through in-person interviews conducted by more than 400 trained staff, some of whom were bilingual in English and Spanish. Survey participants who completed as much of the assessment as their skills allowed were paid \$20 for their time.

In the national survey, data were gathered for a nationally random sample of adults age 16 and older who were living in households. African American and Latino households were oversampled to ensure reliable estimates of literacy proficiencies and to permit analyses of the performance of these subpopulations. In addition to the household population, a random sample of adults in federal and state prisons was surveyed. In the state surveys, a random household sample of adults age 16 to 64 was interviewed in each participating state.

Responses from the national, state, and prison samples were combined to yield the best possible performance estimates. The results of the Florida state survey could not be included in the national estimates, however, due to the

delayed administration.

In all, over 26,000 adults across the nation — randomly selected to represent the approximately 191.3 million adults living in this country — gave more than an hour of their time to complete the literacy tasks and background questionnaires. The national sample included almost 13,600 adults living in households and approximately 1,100 prisoners, as well as state samples of more than 11,300 adults living in households (Table 1). In the Midwest as a whole, 7,494 adults were surveyed, representing some 45.3 million adults in the region, including those living in households and those in prison. A total of 1,246 individuals living in Iowa participated in the study, representing approximately 2.1 million individuals statewide who were living in households.

⁴A more detailed description of the survey design and framework can be found in: A. Campbell, I.S. Kirsch, and A. Kolstad. (1992, October). Assessing Literacy: The Framework for the National Adult Literacy Survey. Washington, DC: National Center for Education Statistics. The Iowa, Midwest, and United States populations are similar in terms of the proportions of men (48 percent) and women (52 percent), but their racial/ ethnic compositions vary. For example, 96 percent of adults in Iowa are White, compared with 76 percent nationwide and 85 percent in the Midwest region. Two percent of the state's residents are African American and 2 percent are



IOWA TABLE 1

Characteristics of the Iowa, Midwest, and National Samples

	Iowa Sample			Midwest Sample			National Sample		
	Survey pop.	Total pop.*	% of pop.	Survey pop	Total pop.*	% of pop.	Survey pop.	Total pop.*	% of pop.
Total	1,246	2,095	100	7,494	45,318	100	26,091	191,289	100
Sex						1.2			
Male	592	1,003	48	3,331	21,621	48	11,770	92,098	48
Female	654	1,093	52	4,152	23,645	52	14,279	98,901	52
Age									
16 to 18	63	137	7	366	2,637	6	1,237	10,424	5
19 to 24	137	220	10	928	5,041	11	3,344	24,515	13
25 to 34	333	423	20	1,895	9,424	21	6,701	41,326	22
35 to 44	295	397	19	1,716	9,230	20	5,930	39,755	21
45 to 54	180	296	14	1,123	6,102	13	3,729	25,992	14
55 to 64	204	296	14	890	4,656	10	2,924	19,503	10
65 and older	33	326	16	574	8,226	18	2,214	29,735	16
Race/Ethnicity									
White	1,180	2,003	96	5,877	38,530	85	17,292	144,968	76
African American	26	34	2	1,161	4,222	9	4,963	21,192	11
Latino (all)	22	34	2	346	1,703	4	3,126	18,481	10
Mexican	13	22	1	213	1,058	2	1,776	10,235	5
Puerto Rican	0	0	0	70	222	0	405	2,190	1
Cuban	0	0	0	4	26	0	147	928	0
C./S. American	3	5	0	34	205	0	424	2,608	1
Other	6	7	0	25	193	0	374	2,520	1
Asian/Pacific Islander	9	12	1	49	282	1	438	4,116	2
Other	9	12	1	61	581	1	272	2,532	1

*Total population figures are in thousands.

Notes: The state sample includes only adults living in households. The regional and national samples include adults living in households and those in prison. The sample sizes for subpopulations may not add up to the total sample sizes due to missing data. Percentages below .5 are rounded to 0. The race/ethnicity categories are mutually exclusive. Some estimates for small subgroups of the national, regional, and state population may be slightly different from 1990 Census estimates due to the sampling procedures used. The state sample of adults age 65 and older may not be representative (see Appendix for more information).

Source: Educational Testing Service, State Adult Literacy Survey, and U.S. Department of Education, National Center for Education Statistics. National Adult Literacy Survey, 1992.

10.... Introduction
Latino, while nationwide, 11 percent of adults are African American and 10 percent are Latino. In the Midwest, 9 percent of the residents are African American and 4 percent are Latino. Furthermore, while the age composition of the Iowa adult population resembles that of the region and nation, Census data show that Iowa has a higher percentage of adults age 65 or older (15.4 percent) than all but two states — Florida and Pennsylvania.⁵ Readers should remember these differences in the composition of the state, regional, and national populations as they interpret the literacy proficiency results discussed in this report.

Reporting the Results

The results of the State Adult Literacy Survey are reported using three scales, each ranging from 0 to 500: a prose scale, a document scale, and a quantitative scale. The scores on each scale represent degrees of proficiency along that particular dimension of literacy. For example, a low score (below 225) on the document scale indicates that an individual demonstrates very limited skills in processing information from tables, charts, graphs, maps, and the like, even those that are brief and uncomplicated. He or she may be able to perform more challenging literacy tasks some of the time — for example, when the material is familiar — but would not be expected to do so with a high degree of consistency. On the other hand, a high score (above 375) indicates that a person displays advanced skills in performing a variety of tasks that involve the use of complex documents. He or she would be expected to process information from challenging materials with a high degree of consistency.

Survey participants obtained proficiency scores according to their

performance on the survey tasks. A relatively small proportion of the respondents answered only a part of the survey, and an imputation procedure was used to make the best possible estimates of their proficiencies. This procedure and related issues are detailed in the forthcoming technical report.

Most respondents tended to obtain similar scores on the three literacy scales, but this does not mean that the underlying skills involved in prose, document, and quantitative literacy are the same. Each scale provides some unique information, especially when comparisons are made across groups defined by variables such as race/ethnicity, education, and age.

The literacy scales allow us not only to summarize results for various subpopulations, but also to determine the relative difficulty of the literacy tasks included in the survey. In other words, just as individuals earned scale scores according to their performance in the assessment, the literacy tasks received

Introduction 11

⁵ U.S. Department of Commerce. (1993). Statistical Abstract of the United States, 113th Edition: The National Data Book. Washington, D.C. p. 33.

scale values according to their difficulty, as determined by the performance of the nationally representative sample of adults who participated in the survey. Previous research has shown that the difficulty of a literacy task, and therefore its placement on the literacy scale, is determined by three factors: the structure of the material — for example, exposition, narrative, table, graph, map, or advertisement; the content of the material and/or the context from which it is drawn — for example, home, work, or community; and the nature of the task — that is, what the individual is asked to do with the material, or his or her purpose for using it.

The literacy tasks administered in the survey varied widely in terms of materials, content, and task requirements, and thus in terms of difficulty. This range is captured in Figure 1, which describes some of the literacy tasks and indicates their scale values. Even a cursory review of this display reveals that tasks at the lower end of each scale differ from ones at the high end. A careful analysis of the range of tasks along each scale reveals an ordered set of information-processing skills and strategies. On the prose scale, for example, tasks with low scale values ask readers to locate or identify information in brief, familiar, or uncomplicated materials, while those at the high end ask them to perform more demanding activities using materials that tend to be lengthy, unfamiliar, or complex. Similarly, on the document and quantitative scales, the tasks at the low end of the scale differ from those at the high end in terms of the structure of the material, the content and context of the material, and the nature of the directive.

In an attempt to capture this progression of information-processing skills and strategies, each scale was divided into five levels:

	Scale range
Level 1	0 to 225
Level 2	226 to 275
Level 3	276 to 325
Level 4	326 to 375
Level 5	376 to 500

The points and score ranges that separate the levels on each scale reflect shifts in the literacy skills and strategies required to perform increasingly complex tasks. Analyses of the types of materials and demands that characterize each

level reveal the progression of literacy demands along each scale (Figure 2). While the literacy levels on each scale can be used to explore the range of literacy demands, these data do not reveal the types of literacy demands that are associated with particular contexts in this pluralistic society. That is, they do not enable us to say what specific level of prose, document, or quantitative skill

12 Introduction

NALS

Difficulty Values of Selected Tasks Along the Prose, Document, and Quantitative Literacy Scales

	1	Prose		Document		Quantitative
,	149	Identify country in short article	69	Sign your name	191	Total a bank deposit entry
\sim	210	Locate one piece of information in sports article	170	Locate expiration date on driver's license		
	224	Underline sentence explaining action	180	Locate time of meeting on a form		
25		stated in short article	214	Using pie graph, locate type of vehicle having specific sales		
	226	Underline meaning of a term given in government brochure on supplemental	230	Locate intersection on a street map	238	Calculate postage and fees for certified mail
		security income	246	Locate eligibility from table of employee benefits	246	Determine difference in price between tickets for two shows
	250	Locate two features of information in sports article	259	Identify and enter background information on application for social	270	Calculate total costs of purchase from
	275	Interpret instructions from an appliance warranty		security card		an order form
75						
	288	Write a brief letter explaining error made on a credit card bill	277	Identify information from bar graph depicting source of energy and year	278	Using calculator, calculate difference between regular and sale price from an advertisement
	304	Read a news article and identify a sentence that provides interpretation of a situation	298	Use sign out sheet to respond to call about resident	308	Using calculator, determine the discount from an oil bill if paid within 10 days
	316	Read lengthy article to identify two behaviors that meet a stated condition	314	Use bus schedule to determine appropriate bus for given set of conditions	321	Calculate miles per gallon using information given on mileage record chart
25			323	Enter information given into an automobile maintenance record form	325	Plan travel arrangements for meeting using flight schedule
25			342	Identify the correct percentage meeting	331	Determine correct change using
	328	State in writing an argument made in lengthy newspaper article	342	specified conditions from a table of such	DOL	information in a menu

- 347 Explain difference between two types of employee benefits
- 359 Contrast views expressed in two editorials on technologies available to make fuel-efficient cars
- 362 Generate unfamiliar theme from short poems
- 374 Compare two metaphors used in poem
- 382 Compare approaches stated in narrative on growing up

375

500

- 410 Summarize two ways lawyers may challenge prospective jurors
- 423 Interpret a brief phrase from a lengthy news article

352 Use bus schedule to determine appropriate bus for given set of conditions

information

352 Use table of information to determine pattern in oil exports across years

Use information in table to complete a

Identify the two categories used and write

Using a table depicting information about

parental involvement in school survey to write a paragraph summarizing extent to

graph including labeling axes

Use table comparing credit cards.

which parents and teachers agree

two differences between them

- 350 Using information stated in news article, calculate amount of money that should go to raising a child
- 368 Using eligibility pamphlet, calculate the yearly amount a couple would receive for basic supplemental security income
- 382 Determine shipping and total costs on an order form for items in a catalog
 - 405 Using information in news article, calculate difference in times for completing a race
 - 421 Using calculator, determine the total cost of carpet to cover a room

Source: U.S. Department of Education, National Center for Education Statistics, National Adult Literacy Survey, 1992.

378

387

395

Introduction 13

Figure 2

Description of the Prose, Document, and Quantitative Literacy Levels

	Prose	Document	Quantitative
Level 1 0-225	Most of the tasks in this level require the reader to read relatively short text to locate a single piece of information which is identical to or synonymous with the information given in the question or directive. If plausible but incorrect information is present in the text, it tends not to be located near the correct information.	Tasks in this level tend to require the reader either to locate a piece of information based on a literal match or to enter information from personal knowledge onto a document. Little, if any, distracting information is present.	Tasks in this level require readers to perform single, relatively simple arithmetic operations, such as addition. The numbers to be used are provided and the arithmetic operation to be performed is specified.
Level 2 226-275	Some tasks in this level require readers to locate a single piece of information in the text; however, several distractors or plausible but incorrect pieces of information may be present, or low- level inferences may be required. Other tasks require the reader to integrate two or more pieces of information or to compare and contrast easily identifiable information based on a criterion provided in the question or directive.	Tasks in this level are more varied than those in Level 1. Some require the readers to match a single piece of information; however, several distractors may be present, or the match may require low-level inferences. Tasks in this level may also ask the reader to cycle through information in a document or to integrate information from various parts of a document.	Tasks in this level typically require readers to perform a single operation using numbers that are either stated in the task or easily located in the material. The operation to be performed may be stated in the question or easily determined from the format of the material (for example, an order form).
Level 3 276-325	Tasks in this level tend to require readers to make literal or synonymous matches between the text and information given in the task, or to make matches that require low-level inferences. Other tasks ask readers to integrate information from dense or lengthy text that contains no organizational aids such as headings. Readers may also be asked to generate a response based on information that can be easily identified in the text. Distracting information is present, but is not located near the correct information.	Some tasks in this level require the reader to integrate multiple pieces of information from one or more documents. Others ask readers to cycle through rather complex tables or graphs which contain information that is irrelevant or inappropriate to the task.	In tasks in this level, two or more numbers are typically needed to solve the problem, and these must be found in the material. The operation(s) needed can be determined from the arithmetic relation terms used in the question or directive.
Level 4 326-375	These tasks require readers to perform multiple-feature matches and to integrate or synthesize information from complex or lengthy passages. More complex inferences are needed to perform successfully. Conditional information is frequently present in tasks at this level and must be taken into consideration by the reader.	Tasks in this level, like those at the previous levels, ask readers to perform multiple-feature matches, cycle through documents, and integrate information; however, they require a greater degree of inferencing. Many of these tasks require readers to provide numerous responses but do not designate how many responses are needed. Conditional information is also present in the document tasks at this level and must be taken into account by the reader.	These tasks tend to require readers to perform two or more sequential operations or a single operation in which the quantities are found in different types of displays, or the operations must be inferred from semantic information given or drawn from prior knowledge.

Some tasks in this level require the

Level 5 376-500

14.... Introduction

reader to search for information in dense text which contains a number of plausible distractors. Others ask readers to make high-level inferences or use specialized background knowledge. Some tasks ask readers to contrast complex information.

Tasks in this level require the reader to search through complex displays that contain multiple distractors, to make high-level text-based inferences, and to use specialized knowledge.

These tasks require readers to perform multiple operations sequentially. They must disembed the features of the problem from text or rely on background knowledge to determine the quantities or operations needed.

Source: U.S. Department of Education, National Center for Education Statistics, National Adult Literacy Survey, 1992.

is required to obtain or hold a job, or advance in a particular occupation, to manage a household, or to obtain legal or community services, for example. Nevertheless, the relationships among performance on the three scales and various social or economic indicators can provide valuable insights.

A Note on Interpretations

The study design and scientific procedures employed in this survey permit a high degree of confidence in the resulting estimates of task difficulty and assure that participants' responses can be generalized to the populations of interest. Readers of this report should bear in mind, however, that the literacy tasks contained in the assessment and the adults invited to participate in the survey are samples drawn from their two respective universes. The results are, accordingly, subject to a measurable degree of uncertainty, which is captured in the standard error enclosed in parentheses after each number presented in the tables.

In situations where there are too few respondents in a group to provide reliable information — specifically, when there are fewer than 45 respondents — no data are provided. Instead, the relevant cells in the table are denoted with asterisks.

Using confidence intervals based on the standard errors provides a way to make inferences about the survey results in a manner that reflects the uncertainty inherent in any sample estimate. An average proficiency score, or a percentage, plus or minus two standard errors represents a 95 percent confidence interval for the corresponding population quantity. For example, full-time employees in Iowa have an average prose score of 299, with a standard error of 4.2. One can conclude with 95 percent certainty that the average prose score of all unemployed adults in Iowa is between 290.6 and 307.4, since $4.2 \times 2 = 8.4$, and $299 \pm 8.4 = 290.6$ to 307.4.

Where this report compares the demonstrated literacy skills of various groups, only those differences that are statistically significant are discussed. Each comparison is based on a statistical test, known as the t statistic, which considers not only the magnitude of the differences between any two groups (for example, the gap in average document proficiency between high school and college graduates), but also the size of the standard errors associated with the numbers being compared and the number of comparisons being made.

The formula used to compute the t statistic is as follows:

 $t = (P_1 - P_2)/\sqrt{(se_1^2 + se_2^2)}$

where P₁ and P₂ are the estimates to be compared and se₁ and se₂ are their corresponding standard errors. Once the t statistic is known, it is necessary to

determine whether this value meets the standard for statistical significance. Generally, when two groups are being compared, determinations of statistical significance are made at the .05 level, indicating that there is only a 5 percent chance that the observed difference is not, in fact, a true difference but is instead due to variability in the population estimates. When multiple comparisons are made using the same data, however, the likelihood of finding a spurious difference increases. To guard against such errors of inference, the Bonferroni procedure is used to correct significance tests for multiple comparisons. This procedure divides the alpha level for a single t test (.05) by the number of comparisons being made.

An example may be helpful. Say that one wanted to compare the average prose proficiencies of Iowa residents whose highest level of education was a high school diploma (283 with a standard error of 3.1) and those whose highest level was a GED or high school equivalency (269 with a standard error of 7.6). The difference in average scores between the two groups $(P_i - P_i)$ is 14, and the standard error associated with the difference $(\sqrt{se_i^2 + se_s^2})$ is 8.2, so the t statistic for this comparison is 1.707.

Since the education variable has nine response categories, the total number of comparisons that could be made using this variable is 36. In actuality, however, we are interested only in comparing one age group with the next higher one. Thus, the number of comparisons being made is eight, rather than 36. Using a published table of critical values that adjusts for multiple comparisons, we find that the statistical significance "threshold" for eight comparisons is 2.735. The t statistic for our comparison (1.707) is below this threshold, so the difference in average prose scores between high school graduates and GED recipients is not considered statistically significant. In comparing various groups, readers are advised to rely on statistical tests of this nature, rather than use the numbers alone. (More detailed information on this topic is provided in the appendices.)

It is important to recognize that even when differences are found among various groups, the nature of the survey makes it impossible to determine the direction of these relationships. In other words, it is impossible to identify the extent to which literacy shapes particular aspects of our lives or is, in turn, shaped by them. For example, there is a strong relationship between educational attainment and literacy proficiencies. On the one hand, it is likely that staying in school longer strengthens an individual's literacy skills. On the other hand, those with more advanced skills tend to remain in school longer. Other variables, as well, are likely to play a role in the relationship between literacy and education.

16.... Introduction

Finally, when comparing the literacy skills of different groups, the range of performance within each group must be kept in mind. While this report describes the literacy proficiencies of subpopulations defined by variables such as age, sex, race, ethnicity, and educational background, clearly the individuals within these groups are not homogeneous with respect to either their characteristics or their proficiencies. Within every group there are some individuals who perform well and some who perform poorly. Accordingly, when one group is said to have lower average scores than another, this does not imply that all adults in the first group performed worse than all those in the second. Such statements are only intended to highlight general patterns of differences among various groups and do not capture the variability within each group.

Above all, the survey results show us that no single factor determines what an individual's literacy proficiencies will be. All of us develop our own unique repertoire of competencies depending on a wide array of conditions and circumstances, including our family backgrounds, educational attainments, interests and aspirations, economic resources, and employment experiences. Any single survey, this one included, can focus on only some of these variables.

About This Report

This report contains five sections. The first, Section I, presents information on the literacy levels and average proficiencies of adults in Iowa, the Midwest, and the nation as a whole. In addition, the performance of different subpopulations is compared — adults in different age groups, the native-born and the foreignborn, and those in different racial/ethnic groups, for example. The remaining sections focus primarily on the Iowa results, although regional and national comparisons are discussed where interesting patterns and differences are evident. Section II provides information on the connection between literacy and education. Section III focuses on the relationships between adults' work and community experiences and their literacy skills. Section IV explores literacy and its association with language use, instruction, and reading and writing practices.

Each of these sections begins with a written summary of the findings, followed by a series of tables that present detailed information. The summaries provide only a general sketch of the data contained in tables, and readers are encouraged to explore the data further to pursue answers to other questions of interest.

SECTION I





he State Adult Literacy Survey, like the National Adult Literacy Survey to which it is linked, collected information on multiple dimensions of adult literacy. This section of the report profiles the prose, document, and quantitative skills of adults in Iowa and compares their performance with that of adults in the Midwest region and the nation as a whole. Performance results are also examined for groups defined by age, country of birth, race/ethnicity, and other characteristics.¹

As described in the Introduction, the results of the National and State Adult Literacy Surveys are reported using three literacy scales — prose, document, and quantitative — each ranging from 0 to 500. In this chapter and throughout the report, these scales are used in two ways to report on adults' literacy skills. Each offers a somewhat different perspective on performance.

Average scores, or "proficiencies," on each scale offer a way to describe literacy skills in general terms. This approach is used, for example, to indicate whether adults in one population group tend to perform better or worse than those in another group. This information is useful, but it reveals little about the distribution of skills within a population or about the types of tasks that can be performed by individuals with varying levels of proficiency. To address these types of questions, it is helpful to examine the percentages of adults in different populations who performed in each of the five levels defined on the prose, document, and quantitative scales: Level 1 (0 to 225), Level 2 (226 to 275), Level 3 (276 to 325), Level 4 (326 to 375), and Level 5 (376 to 500).² Using the literacy levels, it is possible to indicate whether the individuals in one group were more likely than those in another group to demonstrate skills in the lowest, or the highest, levels on each literacy scale.

¹All subpopulations and variables discussed in this report are defined in the appendices.

² An overview of the literacy levels is provided in the Introduction. Section V describes the levels in more detail and includes examples of the types of tasks that were likely to be performed successfully by individuals in each level.

In considering the literacy levels, it is important to remember that each level encompasses a range of performance. As a result, the tasks in any given level are not all of the same difficulty; neither are the individuals who demonstrated skills in that level identical in literacy proficiency. Tasks in the high end of the range for a given level are more challenging than those in the low end; individuals whose proficiencies are in the high end of a level demonstrated success on a more challenging set of literacy tasks than individuals in the low end. The performance of adults in Level 1 is especially heterogeneous, as this level includes individuals who successfully performed only the least demanding literacy tasks in the survey, those who attempted to perform these tasks but seldom succeeded, and those who had such limited skills (or such limited English proficiency) that they did not try to respond to any of the assessment tasks. Thus, while the literacy levels are discussed as distinct units in this section and other parts of the report, the range of performance within each level should be kept in mind.

Results for the Total Population in Iowa, the Midwest, and the Nation

Adults in Iowa had an average proficiency score of 285 on the prose scale, 280 on the document scale, and 287 on the quantitative scale. These average proficiencies are approximately the same as those of adults living in the Midwest region, but are significantly (13 to 16 points) higher than those of adults nationwide (Figure 1.1, Table 1.1). In the state, region, and nation, average scores on each literacy scale were in either the high end of the Level 2 range (226 to 275) or the low end of the Level 3 range (276 to 325). The percentages of adults who demonstrated skills in each level of prose, document, and quantitative proficiency are presented in the pages that follow.

Level 1

Fourteen percent of the adults in Iowa performed in the lowest level defined on the prose scale, while 16 percent were in the lowest level on the document scale and 15 percent were in Level 1 on the quantitative scale. In population terms, approximately 300,000 adults living in the state performed within the range for the lowest literacy level.

In comparison, 16 to 19 percent of adults in the Midwest and 21 to 23 percent of adults nationwide performed in Level 1 on each literacy scale. Thus, the percentages of Iowa adults who demonstrated the most limited skills were equivalent to the percentages of adults in the Midwest who did so and lower than the percentages nationwide who did so.

. Profiles of Adult Literacy in Iowa 22



IOWA FIGURE 1.1

Prose, Document, and Quantitative Literacy Levels and Average Proficiencies: Results for Iowa, the Midwest, and the Nation

PROSE



DOCUMENT



QUANTITATIVE



Level 1 (0 to 225) Level 2 (226 to 275) Level 3 (276 to 325) Level 4 (326 to 375) Level 5 (376 to 500)

Source: Educational Testing Service, State Adult Literacy Survey, and the U.S. Department of Education, National Center for Education Statistics, National Adult Literacy Survey, 1992.



IOWA TABLE 1.1

Prose, Document, and Quantitative Literacy Levels and Average Proficiencies: Results for Iowa, the Midwest, and the Nation

				Percentage of adults in each literacy level								
			Level 1 225 or lower	Level 2 226 to 275	Level 3 276 to 325	Level 4 326 to 375	Level 5 376 or higher	Average Proficiency				
	n	WGT N (/1000)	RPCT (SE)	RPCT (SE)	RPCT (SE)	RPCT (SE)	RPCT (SE)	PROF (SE)				
Prose												
lowa	1,246	2,095	14 (3.3)	24 (2.4)	37 (4.9)	21 (1.6)	3 (0.7)	285 (3.0)				
Midwest	7,494	45,318	16 (0.8)	28 (1.0)	35 (1.2)	18 (0.7)	3 (0.3)	279 (1.1)				
Nation	26,091	191,289	21 (0.4)	27 (0.6)	32 (0.7)	17 (0.4)	3 (0.2)	272 (0.6)				
Document												
lowa	1,246	2,095	16 (2.9)	27 (1.9)	36 (2.9)	19 (1.9)	2 (1.0)	280 (2.8)				
Midwest	7,494	45,318	19 (0.8)	30 (1.1)	33 (1.3)	16 (0.9)	2 (0.3)	274 (1.3)				
Nation	26,091	191,289	23 (0.4)	28 (0.5)	31 (0.5)	15 (0.4)	3 (0.2)	267 (0.7)				
Quantitative	0.000											
lowa	1,246	2,095	15 (2.9)	22 (2.0)	36 (3.4)	23 (2.0)	4 (1.1)	287 (3.4)				
Midwest	7,494	45,318	17 (1.0)	26 (1.5)	34 (1.4)	19 (0.9)	4 (0.3)	280 (1.7)				
Nation	26,091	191,289	22 (0.5)	25 (0.6)	31 (0.6)	17 (0.3)	4 (0.2)	271 (0.7)				

n = sample size; WGT N = population size estimate / 1,000 (the sample sizes for subpopulations may not add up to the total sample sizes, due to missing data); RPCT = row percentage estimate; PROF = average proficiency estimate; (SE) = standard error of the estimate (the reported sample estimate can be said to be within 2 standard errors of the true population value with 95% confidence).

*** Sample size is insufficient to permit a reliable estimate (fewer than 45 respondents).

Source: Educational Testing Service, State Adult Literacy Survey, 1992.

As noted previously, the individuals who performed within the Level 1 range were varied with respect to their skills as well as their characteristics. Some in this literacy level displayed the ability to read relatively short pieces of text to find a single piece of information. Some were able to enter personal information on an application form, or to locate the time of an event on a schedule. Some were able to add numbers provided on a bank deposit slip, or to perform other simple arithmetic operations using numbers presented to them. Others in Level 1, however, were unable to perform even these fairly common and undemanding literacy tasks. Within this group there were individuals who had such limited literacy skills in English that they were able to complete only a portion of the survey, and others who tried to perform the literacy tasks they were given but were largely unsuccessful.

24 Profiles of Adult Literacy in Iowa

Since individuals who performed in the lowest literacy level displayed relatively limited skills, it is important to study their characteristics and compare these with the features of the adult population as a whole (Table 1.2P,D,Q).³ Such an analysis reveals that the educational attainments of adults in Level 1 differ from those of adults in the state population as a whole. On the quantitative scale, for example, Iowa residents with zero to eight years of education were much more prevalent in the Level 1 population (38 percent) than in the statewide population (7 percent). Similarly, about 11 percent of the statewide population reported having nine to 12 years of education, compared with 27 percent of Iowa residents who performed in the lowest level of quantitative literacy. Individuals in Level 1 were much less likely (33 percent) than those in the state population as a whole (77 percent) to have completed high school or a GED or to have attended a postsecondary institution.

Iowa residents who performed in the lowest literacy level were also more likely to be older or disabled than were adults statewide. While 16 percent of the state residents were age 65 or older, about half (53 percent) of the adults in Level 1 were in this age group. Further, about 15 percent of Iowa residents said they had a disability or condition that kept them from participating fully in everyday activities, compared with 35 to 38 percent of the adults who performed in the lowest level on each literacy scale.

Finally, it is interesting to note that although many respondents in Iowa and across the nation demonstrated limited literacy skills, the vast majority described themselves as reading, writing, speaking and understanding English either well or very well. It is possible that their skills, while limited, allow them to meet some or most of their personal and occupational literacy needs. (These

results are explored in Section IV.)

Level 2

Across the three scales, 22 to 27 percent of Iowa adults, or some half a million individuals, performed in the second lowest level of proficiency (Level 2). Twenty-six to 30 percent of adults in the region and 25 to 28 percent of adults nationwide were in this level. Compared with the adults in Level 1, those performing in Level 2 demonstrated skills in performing more diverse and challenging literacy tasks. On the prose scale, respondents whose proficiencies lie within the range for this level demonstrated the ability to make low-level inferences based on what they read and to compare or contrast information that can easily be found in text. Individuals in this level on the document scale were generally able to locate a piece of information in a document in which

³The letters P, D, and Q following the table numbers denote the scale represented in each table: P represents the prose scale; D, the document scale; and Q, the quantitative scale.



IOWA TABLE 1.2P

Characteristics of the Population, by Prose Literacy Level: **Results for Iowa**

CHARACTERISTIC			Percen	tage of adults	in each prose	literacy level v	with each chara	cteristic
	1		Level 1 225 or lower	Level 2 226 to 275	Level 3 276 to 325	Level 4 326 to 375	Level 5 376 or higher	Average Proficiency
	n	WGT N (/1000)	CPCT (SE)	CPCT (SE)	CPCT (SE)	CPCT (SE)	CPCT(SE)	PROF (SE)
Country of birth United States or U.S. territory	1,218	2.057	97 (3.3)	98 (2 3)	99 (4 7)	98 (1 4)	*** / ****)	286 (2.9)
Other country	28	38	3 (13.6)!	2 (13.5)!	1 (8.9)!	2 (11.2)!	()	*** (****)
Race/Ethnicity								
White	1,180	2,003	93 (3.5)	94 (2.1)	96 (5.1)	97 (1.6)	*** (****)	286 (3.1)
African American	26	34	3 (12.1)!	3 (9.2)!	1 (12.2)!	1 (5.3)!	•••• (•••••)	*** (****)
Latino	22	34	2 (13.2)!	2 (14.2)!	1 (18.1)!	1 (10.5)!	*** (****)	*** (****)
Other	9	12	1 (16.1)!	01(22.8)!	1 (16.9)	1(9.0)! $0^{\dagger}(0.4)!$		
Level of education	1.00		1			A 1 2.00		
Still in high school	54	105	3 (3.7)	6 (7.2)	6 (7.0)	4 (4.7)	()	288 (5.1)
0 to 8 years	36	145	39 (11.9)	5 (9.6)	1 (2.5)	ot(0.0)	*** (****)	
9 to 12 years	89	229	27 (14.2)	19 (7.1)	5 (7.0)	2 (3.2)		242 (8.2)
High school	355	708	24 (3.2)	42 (3.4)	39 (5.4)	25 (2.7)	*** (****)	283 (3.1)
GED Some postsecondary	40	500	3 (7.8)	7 (10.5)	4 (8.9)	1 (4.9)		269 (7.6)
Four year degree or more	268	315	0 [†] (0.5)	3 (1.4)	32 (6.9)	31 (3.5) 37 (4.5)	*** (****)	305 (4.7) 338 (2.9)
Age					and the second			
16 to 18	63	137	2 (2.9)	8 (7.6)	8 (7.6)	6 (7,4)	*** (****)	290 (6.4)
19 to 24	137	220	2 (1.8)	11 (3.8)	13 (5.4)	12 (4.7)	*** (****)	302 (4.3)
25 to 39	487	624	13 (2.0)	24 (4.1)	33 (3.1)	40 (4.0)	()	301 (5.4)
40 to 54	321	493	14 (2.9)	21 (4.9)	24 (2.6)	30 (4.8)	•••• (•••••)	298 (8.3)
55 10 64	204	296	16 (4.8)	16 (3.5)!	16 (4.2)	9 (2.8)!	*** (****)	275 (6.3)!
os and older	33	326	53 (4.8)!	19 (32.0)!	7 (24.1)!	3 (9.4)!		()
Physical or mental disability	an			a constant			and and the	
Yes	120	307	38 (7.6)!	19 (10.4)!	9 (11.6)!	4 (4.7)!	()	245 (13.9)!
NO	1,126	1,788	62 (4.6)	81 (4.3)	91 (4.1)	96 (2.1)	()	292 (2.9)

n = sample size; WGT N = population size estimate / 1,000 (the sample sizes for subpopulations may not add up to the total sample sizes, due to missing data); CPCT = column percentage estimate; PROF = average proficiency estimate; (SE) = standard error of the estimate (the reported sample estimate can be said to be within 2 standard errors of the true population value with 95% confidence).

+

Percentages less than 0.5 are rounded to zero. Sample size is insufficient to permit a reliable estimate (fewer than 45 respondents). ***

Interpret with caution - the nature of the sample does not allow accurate determination of the variability of this statistic. 1

Source: Educational Testing Service, State Adult Literacy Survey, 1992.



IOWA TABLE 1.2D

Characteristics of the Population, by Document Literacy Level: **Results for Iowa**

CHARACTERISTIC			Percentag	ge of adults in	each documer	nt literacy level	l with each cha	racteristic
			Level 1 225 or lower	Level 2 226 to 275	Level 3 276 to 325	Level 4 326 to 375	Level 5 376 or higher	Average Proficiency
	n	WGT N (/1000)	CPCT (SE)	CPCT (SE)	CPCT (SE)	CPCT (SE)	CPCT (SE)	PROF (SE)
Country of birth United States or U.S. territory Other country	1,218 28	2,057 38	97 (3.0) 3 (18.1)!	99 (1.9) 1 (4.3)!	98 (2.6) 2 (17.1)!	99 (1.3) 1 (9.9)!	**** (*****) **** (*****)	280 (2.8) *** (****)
Race/Ethnicity White African American Latino Asian/Pacific Islander Other	1,180 26 22 9 9	2,003 34 34 12 12	92 (3.2) 3 (11.6)! 3 (12.1)! 1 (19.1)! 1 (16.9)!	95 (2.1) 2 (9.0)! 1 (14.3)! 0 [†] (0.3)! 1 (36.8)!	96 (2.8) 1 (10.9)! 2 (21.1)! 1 (25.1)! 0 [†] (35.7)!	98 (1.5) 0 [†] (0.4)! 1 (7.0)! 0 [†] (9.0)! 0 [†] (0.5)!	**** (****) *** (****) *** (****) *** (****) *** (****)	281 (2.9) *** (****) *** (****) *** (****) *** (****)
Level of education Still in high school 0 to 8 years 9 to 12 years High school GED Some postsecondary Four year degree or more	54 36 89 355 46 396 268	105 145 229 708 84 509 315	1 (2.6) 35 (10.2) 28 (10.7) 24 (3.5) 3 (4.5) 7 (3.5) 1 (0.6)	6 (5.6) 4 (7.4) 17 (5.6) 43 (3.5) 7 (7.7) 20 (3.2) 4 (2.3)	6 (5.7) 1 (3.0) 4 (6.1) 38 (3.4) 4 (10.5) 32 (3.5) 16 (3.4)	6 (4.8) 0 [†] (0.0) 2 (2.7) 25 (2.7) 1 (4.9) 31 (3.6) 36 (4.2)	*** (****) *** (****)	293 (5.5) *** (****) 235 (12.1) 279 (2.4) 267 (8.7) 300 (5.3) 328 (3.6)
Age 16 to 18 19 to 24 25 to 39 40 to 54 55 to 64 65 and older	63 137 487 321 204 33	137 220 624 493 296 326	2 (2.1) 1 (1.8) 12 (2.0) 15 (3.4) 18 (4.7)! 53 (5.9)!	7 (6.0) 8 (3.6) 22 (2.8) 22 (5.3) 20 (3.8)! 20 (15.1)!	8 (6.2) 15 (4.6) 36 (1.7) 26 (3.6) 12 (4.3)! 4 (20.0)!	8 (7.1) 13 (3.7) 43 (3.4) 27 (4.7) 8 (2.6)! 1 (1.6)!	*** (****) *** (****) *** (****) *** (****) *** (****) *** (****) *** (****)	293 (6.0) 305 (3.2) 300 (5.3) 291 (8.2) 264 (6.3)! **** (*****)
Physical or mental disability Yes No	120 1,126	307 1,788	36 (10.9)! 64 (3.4)	19 (3.6)! 81 (3.3)	8 (8.6)! 92 (1.6)	5 (2.6)! 95 (2.4)	**** (*****) **** (*****)	238 (12.0)! 287 (3.3)

n = sample size; WGT N = population size estimate / 1,000 (the sample sizes for subpopulations may not add up to the total sample sizes, due to missing data); CPCT = column percentage estimate; PROF = average proficiency estimate; (SE) = standard error of the estimate (the reported sample estimate can be said to be within 2 standard errors of the true population value with 95% confidence).

Percentages less than 0.5 are rounded to zero. 1

Sample size is insufficient to permit a reliable estimate (fewer than 45 respondents). Interpret with caution -- the nature of the sample does not allow accurate determination of the variability of this statistic. 1

Source: Educational Testing Service, State Adult Literacy Survey, 1992.



IOWA TABLE 1.2Q

Characteristics of the Population, by Quantitative Literacy Level: **Results for Iowa**

CHARACTERISTIC	A 11-		Percentag	e of adults in	each quantitati	ve literacy lev	el with each ch	aracterístic
	Ľ.,.		Level 1 225 or lower	Level 2 226 to 275	Level 3 276 to 325	Level 4 326 to 375	Level 5 376 or higher	Average Proficiency
	n	WGT N (/1000)	CPCT (SE)	CPCT (SE)	CPCT (SE)	CPCT (SE)	CPCT (SE)	PROF (SE)
Country of birth								
United States or U.S. territory Other country	1,218 28	2,057 38	97 (2.9) 3 (17.1)!	98 (1.5) 2 (13.1)!	99 (3.0) 1 (11.7)!	99 (2.0) 1 (9.5)!	97 (1.8) 3 (3.5)!	288 (3.5)
Race/Ethnicity								
White African American Latino Asian/Pacific Islander	1,180 26 22 9	2,003 34 34 12	91 (3.3) 3 (14.4) 3 (20.7) 1 (19.1)	95 (1.7) 2 (13.6)! 2 (10.2)! 0 [†] (0.0)	96 (3.0) 1 (12.2)! 1 (15.7)! 1 (22.3)!	98 (2.0) 0 [†] (10.2)! 1 (6.9)! 1 (14.9)!	99 (1.6) 0 [†] (0.0) 0 [†] (0.4)! 1 (1.3)!	289 (3.6) () () ()
Other	9	12	1 (16.9)	1 (24.0)!	01(24.8)!	1 (0.3)!	0'(0.0)	()
Level of education Still in high school 0 to 8 years 9 to 12 years High school GED Some postsecondary Four year degree or more	54 36 89 355 46 396 268	105 145 229 708 84 509 315	2 (4.3) 38 (9.1) 27 (11.6) 19 (2.9) 5 (4.4) 9 (5.0) 0 [†] (0.4)	7 (4.6) 4 (6.0) 21 (3.8) 41 (5.4) 6 (6.9) 17 (1.9) 3 (1.7)	6 (5.7) 1 (4.6) 5 (7.9) 41 (3.8) 4 (9.1) 31 (3.9) 12 (2.0)	4 (5.0) 0 [†] (0.2) 1 (2.4) 29 (2.5) 2 (6.0) 30 (3.5) 34 (3.6)	$\begin{array}{c} 2 (2.0) \\ 0^{\dagger} (0.0) \\ 1 (1.3) \\ 15 (4.2) \\ 1 (2.1) \\ 29 (8.1) \\ 52 (7.9) \end{array}$	286 (6.4) *** (****) 240 (9.6) 289 (3.1) 271 (8.6) 307 (7.8) 338 (2.2)
Age 16 to 18 19 to 24 25 to 39 40 to 54 55 to 64 65 and older	63 137 487 321 204 33	137 220 624 493 296 326	2 (3.6) 2 (2.0) 16 (2.7) 13 (3.2) 14 (4.9)! 53 (4.2)!	8 (6.0) 11 (4.1) 22 (3.5) 22 (4.5) 16 (3.6)! 21 (24.8)!	9 (5.7) 13 (5.6) 32 (4.2) 24 (3.4) 16 (4.2)! 6 (18.8)!	5 (5.6) 11 (4.2) 41 (3.5) 29 (4.3) 11 (4.0)! 3 (9.2)!	3 (3.0) 10 (4.2) 42 (5.6) 35 (6.3) 7 (3.5)1 3 (7.8)!	290 (6.0) 302 (3.9) 305 (5.1) 301 (8.7) 281 (8.6)!
Physical or mental disability Yes No	120 1,126	307 1,788	35 (4.9)! 65 (2.3)	23 (9.5)! 77 (2.6)	8 (8.1)! 92 (2.1)	5 (6.3)! 95 (2.4)	5 (2.5)! 95 (2.6)	243 (13.2)! 295 (2.8)

n = sample size; WGT N = population size estimate / 1,000 (the sample sizes for subpopulations may not add up to the total sample sizes, due to missing data); CPCT = column percentage estimate; PROF = average proficiency estimate; (SE) = standard error of the estimate (the reported sample estimate can be said to be within 2 standard errors of the true population value with 95% confidence).

Percentages less than 0.5 are rounded to zero.

•••

Sample size is insufficient to permit a reliable estimate (fewer than 45 respondents). Interpret with caution -- the nature of the sample does not allow accurate determination of the variability of this statistic. L

Source: Educational Testing Service, State Adult Literacy Survey, 1992.

. . Profiles of Adult Literacy in Iowa 28

plausible but incorrect information was also present. Individuals in Level 2 on the quantitative scale were likely to give correct responses to a task involving a single arithmetic operation using numbers that can readily be located in printed material.

Given the differences between the characteristics of the Level 1 population in Iowa and the state population as a whole, it is important to investigate whether certain groups are also over- or underrepresented in the other literacy levels. Iowa residents who performed in Level 2 do resemble the general population in most respects (Table 1.2P,D,Q). For example, the educational attainments and the age and racial/ethnic characteristics of these populations are highly similar.

Level 3

On each scale, 36 to 37 percent of the adults statewide, or about 750,000 individuals, performed in the middle level of literacy (Level 3). Approximately one-third of the adults in the Midwest (33 to 35 percent) and nationwide (31 to 32 percent) performed in this level. Respondents performing in the third level on the prose scale demonstrated skills in matching pieces of information by making low-level inferences and in integrating information from relatively long or dense text. Those in Level 3 on the document scale generally displayed the ability to integrate multiple pieces of information found in documents. Adults in this level on the quantitative scale displayed proficiency in using two or more numbers found in printed material and in interpreting arithmetic terms included in the question.

Iowa residents who performed in the Level 3 range differ in some important respects from the state population as a whole (Table 1.2P,D,Q). For one, Iowa adults who scored in the middle of the proficiency range tend to be better educated than the state's adult population as a whole. For instance, only 5 to 6 percent of the adults in Level 3 reported that they had not attained a high school diploma or GED, compared with 18 percent of the entire state population. Adults who performed in Level 3 were also less likely than adults in the general population to be age 65 or older or to report having an illness or disability.

Level 4

Nineteen to 23 percent of the adults in Iowa, or nearly half a million individuals, demonstrated skills in the fourth literacy level. Similarly, 16 to 19 percent of the adults in the Midwest performed in this level. Nationwide, 17 percent performed in this level on the prose and quantitative scales, and 15 percent were in this level on the document scale. Respondents who demonstrated skills

in the Level 4 range completed many of the more difficult assessment tasks successfully. Looking across the scales, adults in the fourth literacy level displayed an ability to synthesize information from lengthy or complex passages, to make inferences based on text and documents, and to perform sequential arithmetic operations using numbers found in different types of displays. To give correct responses to these types of tasks, readers were often required to make high-level, text-based inferences or to draw on their background knowledge.

When one compares the Level 4 population with the entire adult population in Iowa, interesting contrasts are evident (Table 1.2P,D,Q). As was observed in the previous level, respondents with proficiencies in Level 4 on the prose scale were more likely than those in the state population as a whole to have completed high school or a GED or to have attended some postsecondary education. Respondents who performed in the fourth level on each literacy scale were also far less likely than adults in the total population to be age 65 or older or to report having an illness or disability.

Level 5

On each of the three literacy scales, just 2 to 4 percent of the respondents in Iowa, the Midwest, and the nation as a whole performed in Level 5 — the highest level defined. Some tasks at this level required readers to contrast complex information found in written materials, while others required them to make high-level inferences or to search for information in dense text. On the document scale, adults performing in Level 5 showed the ability to use specialized knowledge and to search through complex displays for particular pieces of information. Respondents in the highest level on the quantitative scale demonstrated the ability to determine the features of arithmetic problems either by examining text or by using background knowledge, and then to perform the multiple arithmetic operations required. Not more than 84,000 individuals statewide, and less than 8 million nationwide, demonstrated success on these types of tasks — the most difficult included in the survey.

Adults who performed in the highest level on the quantitative scale look quite different, on the whole, from adults in the state population at large (Table 1.2P,D,Q). They are less likely to be older or to have a disability, and



Results for Adults in Different Age Groups

The age composition of the Iowa adult population is similar to that of the regional and national populations. Older adults (age 65 and older) comprise 16 percent of the adult population in Iowa and the nation and 18 percent of the adult population in the Midwest (Table 1.3P,D,Q).

Adults age 55 to 64 and particularly those age 65 and older were more likely than younger individuals to perform in the two lowest literacy levels. In Iowa, the number of adults in the oldest age group was too small to provide reliable proficiency estimates; however, 16 percent of the Iowa adults in the 55 to 64 age group performed in Level 1 on the prose scale and another 28 percent were in Level 2. Similarly, among adults in the Midwest and nationwide, individuals in the two oldest age groups were more likely than those in the other age groups to demonstrate skills in the two lowest literacy levels. Nationwide, for example, 55 to 64 percent of the 55- to 64-year-olds and 71 to 85 percent of those age 65 and older were in the two lowest levels.

Among the younger age groups, the differences in the percentages of individuals who performed in each literacy level are relatively small. Individuals in the middle age categories were, however, more likely than those in both the younger and the older age groups to reach the highest levels. This pattern is reflected more clearly in the average proficiency results. On the prose scale, for example, average scores rise from 290 among Iowa's 16- to 18-year-olds to 303 among 35- to 44-year-olds before declining across the older age groups (to 297 among 45- to 54-year-olds and 275 among 55- to 64-year-olds).

What explains these performance declines across the age groups? Given

the association between education and literacy, one hypothesis is that some of the proficiency gap between older and younger adults is associated with differences in years of schooling. The survey results do, in fact, indicate that older adults — in particular, those age 55 to 64 and age 65 and older — tend to have completed fewer years of schooling than younger adults.⁴ (This is explored in Section II; see Table 2.2 and accompanying discussion.) Whereas adults below age 55 had attended an average of 12 to 13 years of schooling, those age 65 and older had completed 11 years, on average.

Even when one controls for level of education, however, significant differences in literacy proficiencies across the age groups remain. Thus, other factors beyond education must contribute to the performance gaps observed. Changing immigration patterns may be a factor, for example, as may factors associated with the aging process.⁵

*The exception to this pattern occurs among 16- to 18-year-olds, many of whom are still in high school.
*I.S. Kirsch, A. Jungeblut, L. Jenkins, and A. Kolstad. (1993). Adult Literacy in America: A First Look at the Results of the National Adult Literacy Survey. Washington, DC: U.S. Government Printing Office.



IOWA TABLE 1.3P

Prose Literacy Levels and Average Proficiencies, by Age: Results for Iowa, the Midwest, and the Nation

AGE					Percent	age of adults in	each prose lite	racy level	
				Level 1 225 or lower	Level 2 226 to 275	Level 3 276 to 325	Level 4 326 to 375	Level 5 376 or higher	Average Proficiency
	n	WGT N (/1000)	PCT	RPCT (SE)	RPCT (SE)	RPCT (SE)	RPCT (SE)	RPCT (SE)	PROF (SE)
lowa									
16 to 18	63	137	7	5 (3.2)	29 (10.9)	46 (12.8)	19 (7.9)	1 (1.7)	290 (6.4)
19 to 24	137	220	10	2 (2.0)	25 (4.7)	46 (6.5)	24 (5.5)	3 (1.5)	302 (4.3)
25 to 34	333	423	20	7 (2.2)	22 (4.3)	40 (4.4)	27 (4.3)	4 (2.2)	299 (6.2)
35 to 44	295	397	19	6 (2.0)	17 (3.3)	43 (2.9)	29 (3.5)	5 (2.6)	303 (4.5)
45 to 54	180	290	14	16 (4 9)	23 (3.7)!	35 (3.9)!	13 (3.5)	2 (0.8)	275 (6.3)
65 and older	33	326	16	*** (****)	*** (****)	*** (****)	*** (****)	*** (****)	*** (****)
Midwest									
16 to 18	366	2,637	6	10 (2.9)	32 (4.2)	44 (4.3)	13 (4.7)	1 (1.2)	282 (3.9)
19 to 24	928	5,041	11	10 (2.1)	29 (3.7)	41 (3.9)	18 (2.3)	2 (0.8)	286 (3.1)
25 to 34	1,895	9,424	21	9 (1.2)	26 (1.7)	38 (1.8)	23 (1.8)	4 (1.1)	294 (2.5)
35 to 44	1,716	9,230	20	10 (1.6)	21 (1.7)	36 (2.9)	27 (1.8)	6 (0.7)	297 (2.8)
45 to 54	1,123	6,102	13	11 (1.8)	25 (2.5)	41 (3.1)	20 (2.8)	4 (1.1)	289 (2.9)
55 to 64	890	4,656	10	18 (1.9)	32 (3.7)	36 (2.5)	13 (1.8)	1 (0.6)	2/1 (2.5)
65 and older	574	8,226	18	42 (3.4)	36 (3.3)	18 (2.1)	4 (1.4)	0'(0.2)	234 (4.1)
Nation						1		and the second	
16 to 18	1,237	10,424	5	16 (1.3)	35 (1.9)	38 (2.4)	11 (1.7)	1 (0.4)	271 (1.8)
19 to 24	3,344	24,515	13	14 (1.1)	29 (1.7)	37 (1.8)	18 (1.3)	2 (0.4)	280 (1.3)
25 to 34	6,701	41,326	22	16 (0.7)	25 (1.0)	34 (0.8)	21 (0.9)	4 (0.4)	282 (1.2)
35 to 44	5,930	39,755	21	14 (0.6)	21 (1.0)	35 (1.2)	24 (0.8)	6 (0.5)	289 (1.3)
45 to 54	3,729	25,992	14	16 (0.9)	25 (1.3)	34 (1.6)	21 (1.0)	5 (0.5)	282 (1.7)
55 to 64	2,924	19,503	10	26 (1.5)	31 (1.3)	30 (1.5)	12 (1.1)	1 (0.3)	200 (1.9)
65 and older	2,214	29,735	10	44 (1.6)	32 (1.6)	19 (1.3)	5 (0.9)	1 (0.3)	200 (2.1)

n = sample size; WGT N = population size estimate / 1,000 (the sample sizes for subpopulations may not add up to the total sample sizes, due to missing data); PCT = percentage in group; RPCT = row percentage estimate; PROF = average proficiency estimate; (SE) = standard error of the estimate (the reported sample estimate can be said to be within 2 standard errors of the true population value with 95% confidence).

† Percentages less than 0.5 are rounded to zero.

*** Sample size is insufficient to permit a reliable estimate (fewer than 45 respondents).

1 Interpret with caution -- the nature of the sample does not allow accurate determination of the variability of this statistic.

Source: Educational Testing Service, State Adult Literacy Survey, 1992.

32 Profiles of Adult Literacy in Iowa

IOWA TABLE 1.3D

Document Literacy Levels and Average Proficiencies, by Age: Results for Iowa, the Midwest, and the Nation

									1
AGE					Percentag	e of adults in ea	ach document li	teracy level	
				Level 1 225 or lowe	r 226 to 275	Level 3 276 to 325	Level 4 326 to 375	Level 5 376 or higher	Average Proficiency
	n	WGT N (/1000)	PCT	RPCT (SE)	RPCT (SE)	RPCT (SE)	RPCT (SE)	RPCT (SE)	PROF (SE)
Lowa 16 to 18 19 to 24 25 to 34 35 to 44 45 to 54 55 to 64 65 and older	63 137 333 295 180 204 33	137 220 423 397 296 296 326	7 10 20 19 14 14 16	4 (2.7) 2 (2.0) 7 (2.4) 7 (2.4) 11 (4.1) 20 (4.7) **** (*****	30 (7.8) 21 (4.6) 22 (3.4) 19 (4.4) 19 (4.4) 19 (6.8)! 38 (4.3)! *** (****)	41 (9.1) 50 (7.4) 42 (5.3) 45 (4.2) 35 (4.6)! 30 (4.5)! **** (****)	23 (7.2) 24 (6.6) 26 (5.3) 25 (4.9) 21 (5.7)! 11 (2.6)! **** (****)	1 (1.5) 3 (2.5) 3 (2.9) 4 (2.4) 3 (2.7)! 1 (0.6)! **** (****)	293 (6.0) 305 (3.2) 299 (6.6) 300 (4.5) 288 (10.8)! 264 (6.3)! **** (****)
Midwest 16 to 18 19 to 24 25 to 34 35 to 44 45 to 54 55 to 64 65 and older	366 928 1,895 1,716 1,123 890 574	2,637 5,041 9,424 9,230 6,102 4,656 8,226	6 11 21 20 13 10 18	8 (2.4 9 (1.7 10 (1.6 12 (1.6 13 (1.5 23 (2.0 49 (3.5) 32 (3.5) 29 (3.1) 25 (1.4) 24 (2.3) 31 (2.8) 39 (3.2) 36 (4.1)	43 (4.4) 42 (3.5) 38 (2.1) 36 (2.7) 36 (3.4) 30 (2.5) 13 (2.5)	16 (2.6) 18 (2.1) 23 (2.0) 24 (2.2) 18 (1.9) 8 (1.3) 2 (1.1)	1 (0.8) 2 (0.8) 3 (0.9) 5 (1.0) 2 (1.0) 1 (0.4) 0 [†] (0.1)	286 (3.8) 287 (2.9) 292 (2.2) 292 (3.0) 280 (2.4) 259 (2.1) 222 (3.8)
Nation 16 to 18 19 to 24 25 to 34 35 to 44 45 to 54 55 to 64 65 and older	1,237 3,344 6,701 5,930 3,729 2,924 2,214	10,424 24,515 41,326 39,755 25,992 19,503 29,735	5 13 22 21 14 10 16	15 (1.4 14 (1.0 16 (0.7 15 (0.9 18 (1.1 30 (1.4 53 (1.5	$\begin{array}{c ccccccccccccccccccccccccccccccccccc$	38 (2.6) 37 (1.6) 35 (0.8) 35 (1.1) 33 (1.4) 26 (1.3) 13 (1.0)	12 (1.9) 18 (1.1) 21 (0.9) 22 (1.1) 17 (0.8) 8 (0.8) 2 (0.5)	1 (0.5) 2 (0.4) 4 (0.3) 5 (0.5) 3 (0.6) 1 (0.3) 0 [†] (0.1)	274 (1.8) 280 (1.3) 281 (1.2) 283 (1.4) 273 (1.4) 249 (1.9) 217 (2.1)

- n = sample size; WGT N = population size estimate / 1,000 (the sample sizes for subpopulations may not add up to the total sample sizes, due to missing data); PCT = percentage in group; RPCT = row percentage estimate; PROF = average proficiency estimate; (SE) = standard error of the estimate (the reported sample estimate can be said to be within 2 standard errors of the true population value with 95%) confidence).
- Percentages less than 0.5 are rounded to zero. t
- ***
- Sample size is insufficient to permit a reliable estimate (fewer than 45 respondents). Interpret with caution -- the nature of the sample does not allow accurate determination of the variability of this statistic. 1

Source: Educational Testing Service, State Adult Literacy Survey, 1992.



IOWA TABLE 1.3Q

Quantitative Literacy Levels and Average Proficiencies, by Age: Results for Iowa, the Midwest, and the Nation

AGE				Percentage of adults in each quantitative literacy level							
_				Level 1 225 or lower	Level 2 226 to 275	Level 3 276 to 325	Level 4 326 to 375	Level 5 376 or higher	Average Proficiency		
	n	WGT N (/1000)	PCT	RPCT (SE)	RPCT (SE)	RPCT (SE)	RPCT (SE)	RPCT (SE)	PROF (SE)		
lowa	-					Max man			-		
16 to 18	63	137	7	5 (3.9)	29 (7.9)	49 (7.3)	16 (6.4)	2 (1.9)	290 (6.0)		
19 to 24	137	220	10	3 (2.4)	23 (6.0)	46 (8.0)	23 (4.9)	4 (2.4)	302 (3.9)		
25 to 34	333	423	20	9 (3.0)	16 (2.9)	41 (4.4)	28 (3.3)	6 (2.3)	302 (5.8)		
35 to 44	295	397	19	6 (2.2)	18 (4.1)	36 (2.4)	34 (4.1)	0(2.1)	307 (4.0)		
45 to 54	180	296	14	8(4.0)	23 (5.4)	35 (4.6)!	19 (1.9)!	2 (1 0)	281 (8 6)		
55 to 64	204	326	14	14 (5.0)	*** (****)	*** (****)	10 (4.0):	*** (****)	*** (****)		
Midwest											
16 to 18	366	2 637	6	12(27)	33 (5 5)	38 (4.9)	16 (3.2)	1 (0.5)	281 (4.0)		
19 to 24	928	5.041	11	12 (2.1)	27 (2.8)	41 (2.9)	17 (2.3)	3 (0.7)	285 (3.2)		
25 to 34	1.895	9,424	21	11 (1.6)	24 (2.2)	37 (2.0)	23 (1.6)	5 (1.2)	293 (2.5)		
35 to 44	1,716	9,230	20	11 (1.5)	21 (3.0)	33 (2.7)	27 (2.7)	7 (1.3)	297 (3.7)		
45 to 54	1,123	6,102	13	12 (2.2)	25 (2.3)	35 (2.7)	23 (2.6)	4 (1.3)	290 (2.5)		
55 to 64	890	4,656	10	18 (1.9)	29 (4.1)	36 (3.4)	14 (2.1)	2 (0.7)	272 (2.8)		
65 and older	574	8,226	18	39 (3.6)	30 (2.7)	22 (3.0)	7 (1.3)	1 (0.6)	237 (5.0)		
Nation	1							20.20			
16 to 18	1,237	10,424	5	20 (1.7)	35 (2.6)	33 (1.9)	12 (1.5)	1 (0.5)	268 (1.8)		
19 to 24	3,344	24,515	13	16 (1.1)	28 (1.4)	37 (1.4)	16 (1.0)	2 (0.5)	277 (1.6)		
25 to 34	6,701	41,326	22	17 (0.7)	24 (0.7)	34 (0.8)	20 (0.8)	5 (0.5)	281 (1.1)		
35 to 44	5,930	39,755	21	15 (0.8)	21 (1.1)	33 (1.0)	25 (0.7)	6 (0.5)	288 (1.4)		
45 to 54	3,729	25,992	14	17 (1.1)	24 (1.2)	33 (1.2)	21 (1.4)	5 (0.5)	262 (1.0)		
55 to 64	2,924	19,503	10	25 (1.5)	30 (1.9)	30 (1.6)	13 (1.2)	2(0.0)	207 (2.0)		
65 and older	2,214	29,735	16	45 (1.6)	26 (1.2)	20 (1.2)	7 (0.7)	2 (0.4)	227 (2.0)		

n = sample size; WGT N = population size estimate / 1,000 (the sample sizes for subpopulations may not add up to the total sample sizes, due to missing data); PCT = percentage in group; RPCT = row percentage estimate; PROF = average proficiency estimate; (SE) = standard error of the estimate (the reported sample estimate can be said to be within 2 standard errors of the true population value with 95% confidence).

† Percentages less than 0.5 are rounded to zero.

*** Sample size is insufficient to permit a reliable estimate (fewer than 45 respondents).

I Interpret with caution - the nature of the sample does not allow accurate determination of the variability of this statistic.

Source: Educational Testing Service, State Adult Literacy Survey, 1992.

34 Profiles of Adult Literacy in Iowa

Results for Adults Born in the United States and Those Born in Other Countries

The vast majority of Iowa residents (98 percent) were born in this country or in one of its territories (Table 1.4P,D,Q). The proportion of the state's residents who were born in another country (2 percent) is approximately the same as the proportion in the Midwest (3 percent). In contrast, nationwide, about 10 percent of the adults were born outside the United States.

As expected, adults born in the United States tended to be more proficient in English than individuals born abroad, many of whom have learned English as a second language. The number of foreign-born adults in Iowa is too small to provide reliable proficiency estimates. In the Midwest and nation, however, about half the foreign-born adults (43 to 52 percent) performed in Level 1 on each scale, compared with 15 to 20 percent of the native-born residents.



IOWA TABLE 1.4P

Prose Literacy Levels and Average Proficiencies, by Country of Birth: Results for Iowa, the Midwest, and the Nation

COUNTRY OF BIRTH		Percenta	ge of adults in	each prose lite	eracy level	
	Level 1 225 or lower	Level 2 226 to 275	Level 3 276 to 325	Level 4 326 to 375	Level 5 376 or higher	Average Proficience

	n	WGT N (/1000)	PCT	RPCT (SE)	RPCT (SE)	RPCT (SE)	RPCT (SE)	RPCT (SE)	PROF (SE)
Lowa U.S. or U.S. territory Other country	1,218	2,057 38	98 2	14 (3.3) *** (****)	24 (2.4) *** (****)	38 (4.8) *** (****)	21 (1.6) *** (****)	3 (0.8) *** (****)	286 (2.9) *** (****)
Midwest U.S. or U.S. territory Other country	7,179 315	43,733 1,585	97 3	15 (0.9) 46 (5.4)	28 (1.1) 26 (4.3)	35 (1.3) 19 (5.1)	18 (0.8) 7 (1.3)	3 (0.3) 3 (1.7)	281 (1.1) 223 (7.9)
Nation U.S. or U.S. territory Other country	23,376 2,715	172,162 19,127	90 10	17 (0.4) 52 (1.4)	27 (0.6) 22 (1.1)	34 (0.8) 17 (1.3)	18 (0.5) 7 (0.7)	3 (0.2) 1 (0.4)	279 (0.7) 212 (2.4)

n = sample size; WGT N = population size estimate / 1,000 (the sample sizes for subpopulations may not add up to the total sample sizes, due to missing data); PCT = percentage in group; RPCT = row percentage estimate; PROF = average proficiency estimate; (SE) = standard error of the estimate (the reported sample estimate can be said to be within 2 standard errors of the true population value with 95% confidence).

Percentages less than 0.5 are rounded to zero.

*** Sample size is insufficient to permit a reliable estimate (fewer than 45 respondents).

I Interpret with caution -- the nature of the sample does not allow accurate determination of the variability of this statistic.

Source: Educational Testing Service, State Adult Literacy Survey, 1992.



IOWA TABLE 1.4D

Document Literacy Levels and Average Proficiencies, by Country of Birth: Results for Iowa, the Midwest, and the Nation

COUNTRY OF BIRTH					Percentag	e of adults in e	ach document	literacy level	
				Level 1 225 or lower	Level 2 226 to 275	Level 3 276 to 325	Level 4 326 to 375	Level 5 376 or higher	Average Proficiency
	n	WGT N (/1000)	PCT	RPCT (SE)	RPCT (SE)	RPCT (SE)	RPCT (SE)	RPCT (SE)	PROF (SE)
Lowa U.S. or U.S. territory	1,218	2,057	98	16 (3.0)	27 (1.9)	36 (3.0)	19 (2.0)	2 (1.0)	280 (2.8)
Midwest	28	30	2		()				
U.S. or U.S. territory Other country	7,179 315	43,733 1,585	97 3	18 (0.8) 43 (5.0)	30 (1.1) 26 (4.7)	33 (1.4) 21 (3.8)	16 (0.9) 8 (3.5)	2 (0.4) 2 (1.8)	275 (1.3) 227 (8.5)
Nation		170 100	-				101.0.0		070 (0.7)
Other country	23,376 2,715	172,162	10	51 (1.4)	29 (0.5) 23 (1.2)	32 (0.6) 18 (1.0)	7 (0.8)	1 (0.2)	212 (2.3)

n = sample size; WGT N = population size estimate / 1,000 (the sample sizes for subpopulations may not add up to the total sample sizes, due to missing data); PCT = percentage in group; RPCT = row percentage estimate; PROF = average proficiency estimate; (SE) = standard error of the estimate (the reported sample estimate can be said to be within 2 standard errors of the true population value with 95% confidence).

[†] Percentages less than 0.5 are rounded to zero.

*** Sample size is insufficient to permit a reliable estimate (fewer than 45 respondents).

Interpret with caution -- the nature of the sample does not allow accurate determination of the variability of this statistic. Source: Educational Testing Service, State Adult Literacy Survey, 1992.

> Similar percentages of native-born and foreign-born adults performed in the second lowest level on each literacy scale (22 to 30 percent), but adults born in the United States or a territory were much more likely than those born abroad to reach the third and fourth proficiency levels. Across the literacy scales, approximately one-third of the native-born individuals in the region and nation demonstrated skills in the Level 3 range, and 16 to 19 percent performed in the Level 4 range. In contrast, only 17 to 21 percent of the foreign-born adults reached the third level and just 7 to 9 percent attained the fourth. Across the scales, 2 to 4 percent of adults born in the United States

and 1 to 3 percent of those born abroad performed in the highest literacy level (Level 5).

36 Profiles of Adult Literacy in Iowa



IOWA TABLE 1.4Q

Quantitative Literacy Levels and Average Proficiencies, by Country of Birth: Results for Iowa, the Midwest, and the Nation

COUNTRY OF BIRTH					Percentage	of adults in ea	ch quantitative	literacy level	
				Level 1 225 or lower	Level 2 226 to 275	Level 3 276 to 325	Level 4 326 to 375	Level 5 376 or higher	Average Proficiency
	n	WGT N (/1000)	PCT	RPCT (SE)	RPCT (SE)	RPCT (SE)	RPCT (SE)	RPCT (SE)	PROF (SE)
Iowa U.S. or U.S. territory Other country	1,218 28	2,057 38	98 2	15 (3.0) *** (****)	22 (2.1) **** (*****)	36 (3.4) *** (****)	23 (2.0) *** (****)	4 (1.0) **** (*****)	288(3.5) ***(****)
Midwest U.S. or U.S. territory Other country	7,179 315	43,733 1,585	97 3	16 (0.9) 43 (5.4)	26 (1.5) 25 (4.8)	34 (1.4) 20 (4.3)	19 (0.9) 9 (3.5)	4 (0.3) 2 (1.6)	281 (1.7) 229 (9.3)
Nation U.S. or U.S. territory Other country	23,376 2,715	172,162 19,127	90 10	19 (0.5) 49 (1.6)	26 (0.5) 22 (1.6)	33 (0.6) 19 (1.1)	19 (0.3) 8 (0.7)	4 (0.2) 2 (0.4)	278 (0.8) 214 (2.8)

n = sample size; WGT N = population size estimate / 1,000 (the sample sizes for subpopulations may not add up to the total sample sizes, due to missing data); PCT = percentage in group; RPCT = row percentage estimate; PROF = average proficiency estimate; (SE) = standard error of the estimate (the reported sample estimate can be said to be within 2 standard errors of the true population value with 95% confidence).

- † Percentages less than 0.5 are rounded to zero.
- *** Sample size is insufficient to permit a reliable estimate (fewer than 45 respondents).
- Interpret with caution -- the nature of the sample does not allow accurate determination of the variability of this statistic.

Source: Educational Testing Service, State Adult Literacy Survey, 1992.

The performance gap between native-born and foreign-born adults is also reflected in the average proficiency results. Among Midwest adults, the gap in average prose scores between these two groups is approximately 60 points. Foreign-born residents had average scores in the high end of the Level 1 range, while native-born individuals had scores in the Level 3 range. On the document and quantitative scales, the difference in average scores between the two groups is about 50 points.

Results for foreign-born adults by the number of years lived in the United States

In addition to contrasting the literacy skills of adults born in this country with the skills of those born elsewhere, it is useful to compare the performance of foreign-born individuals who have lived in this country for varying lengths of time (Table 1.5P,D,Q). One might expect individuals who have lived in this country for many years to demonstrate higher proficiencies in English than those who immigrated more recently.

The numbers of foreign-born Iowa residents are too small to provide reliable proficiency estimates for those who have lived in this country for various lengths of time. In addition, while the samples of such adults in the



IOWA TABLE 1.5P

Prose Literacy Levels and Average Proficiencies of Foreign-born Adults, by Years Lived in the United States: Results for Iowa, the Midwest, and the Nation

YEARS LIVED IN THE UNITED STATES				-	Percente	age of adults in	each prose lite	racy level	
				Level 1 225 or lower	Level 2 226 to 275	Level 3 276 to 325	Level 4 326 to 375	Level 5 376 or higher	Average Proficiency
	n	WGT N (/1000)	PCT	RPCT (SE)	RPCT (SE)	RPCT (SE)	RPCT (SE)	RPCT (SE)	PROF (SE)
lowa				and a		- marine	and a	Landa	
1 to 5 years	8	12	32	**** (*****)	*** (****)	*** (****)	()	()	()
6 to 10 years	4	5	13	*** (****)	()	()	()	()	()
More than 10 years	15	21	55	()	()	()	()	()	()
Midwest								and the second second	
1 to 5 years	62	278	18	54 (12.1)	19 (7.2)	22 (12.0)	5 (4.1)	0 [†] (0.2)	210 (22.0)
6 to 10 years	56	294	19	39 (13.9)	40 (17.3)1	18 (16.6)!	2 (1.4)1	01(0.3)!	225 (14.6)!
More than 10 years	181	940	62	46 (6.5)	23 (5.2)	19 (4.7)	8 (2.3)	4 (2.9)	229 (9.1)
Nation									
1 to 5 years	568	3,998	22	61 (2.6)	18 (2.6)	15 (2.2)	6 (1.6)	0*(0.7)	197 (4.3)
6 to 10 years	482	3,184	17	61 (3.4)	22 (4.5)	12 (3.8)	3 (1.8)	1 (0.7)	200 (4.9)
More than 10 years	1,565	11,266	61	48 (2.0)	24 (1.4)	19 (1.7)	8 (1.0)	1 (0.6)	220 (3.2)

n = sample size; WGT N = population size estimate / 1,000 (the sample sizes for subpopulations may not add up to the total sample sizes, due

- to missing data); PCT = percentage in group; RPCT = row percentage estimate; PROF = average proficiency estimate; (SE) = standard error of the estimate (the reported sample estimate can be said to be within 2 standard errors of the true population value with 95% confidence).
- Percentages less than 0.5 are rounded to zero.
- Sample size is insufficient to permit a reliable estimate (fewer than 45 respondents).
 Interpret with caution the nature of the sample does not allow accurate determination of the variability of this statistic.
 Source: Educational Testing Service, State Adult Literacy Survey, 1992.

38 Profiles of Adult Literacy in Iowa

IOWA TABLE 1.5D

Document Literacy Levels and Average Proficiencies of Foreign-born Adults, by Years Lived in the United States: Results for Iowa, the Midwest, and the Nation

VEARS LIVED IN THE UNITED STATES				-	Percentage	e of adults in ea	ich document li	teracy level	
				Level 1 225 or lower	Level 2 226 to 275	Level 3 276 to 325	Level 4 326 to 375	Level 5 376 or higher	Average Proficiency
	n	WGT N (/1000)	PCT	RPCT (SE)	RPCT (SE)	RPCT (SE)	RPCT (SE)	RPCT (SE)	PROF (SE)
lowa									
1 to 5 years	8	12	32	*** (****)	**** (*****)			*** (****)	*** (****)
6 to 10 years	4	5	13	()		*** (****)	*** (****)	*** (****)	*** (****)
More than 10 years	15	21	55		()	()	()		()
Midwest	1						in the second	and and	
1 to 5 years	62	278	18	47 (11.5)	19 (5.5)	21 (10.2)	12 (12.5)	1 (0.9)	223 (27.0)
6 to 10 years	56	294	19	34 (13.5)!	38 (17.0)!	23 (11.8)!	4 (7.3)!	1 (1.6)!	235 (13.6)
More than 10 years	181	940	62	45 (5.6)	25 (4.2)	20 (3.8)	7 (2.0)	3 (3.2)	227 (9.9)
Nation									
1 to 5 years	568	3,998	22	58 (3.1)	21 (3.2)	15 (3.1)	6 (1.4)	1 (0.4)	198 (4.9)
6 to 10 years	482	3,184	17	58 (3.4)	21 (3.5)	16 (2.5)	4 (2.4)	1 (0.9)	202 (5.1)
More than 10 years	1,565	11,266	61	48 (1.9)	25 (1.8)	19 (1.5)	7 (0.9)	1 (0.3)	218 (2.9)

n = sample size; WGT N = population size estimate / 1,000 (the sample sizes for subpopulations may not add up to the total sample sizes, due to missing data); PCT = percentage in group; RPCT = row percentage estimate; PROF = average proficiency estimate; (SE) = standard error of the estimate (the reported sample estimate can be said to be within 2 standard errors of the true population value with 95% confidence).

Percentages less than 0.5 are rounded to zero.

*** Sample size is insufficient to permit a reliable estimate (fewer than 45 respondents).

Interpret with caution -- the nature of the sample does not allow accurate determination of the variability of this statistic.

Source: Educational Testing Service, State Adult Literacy Survey, 1992.

Midwest are larger, the estimates they yield are still unstable (note the large standard errors). The national results therefore provide firmer ground for comparisons.

As seen in the preceding tables, approximately 10 percent of the adults living in the United States — or about 19 million individuals — were born in other countries. About 22 percent of these foreign-born adults have lived in this country for one to five years, 17 percent have lived here for six to 10 years, and 61 percent have lived here for more than 10 years.



IOWA TABLE 1.5Q

Quantitative Literacy Levels and Average Proficiencies of Foreign-born Adults, by Years Lived in the United States: Results for Iowa, the Midwest, and the Nation

VEARS LIVED IN THE UNITED STATES					Percentage	of adults in ea	ch quantitative	literacy level	
				Level 1 225 or lower	Level 2 226 to 275	Level 3 276 to 325	Level 4 326 to 375	Level 5 376 or higher	Average Proficiency
	n	WGT N (/1000)	PCT	RPCT (SE)	RPCT (SE)	RPCT (SE)	RPCT (SE)	RPCT (SE)	PROF (SE)
lowa									
1 to 5 years	в	12	32	*** (****)	*** (****)	*** (****)	*** (****)	*** (****)	*** (****)
6 to 10 years	4	5	13	*** (****)	*** (****)	*** (****)	*** (****)	*** (****)	*** (****)
More than 10 years	15	21	55	*** (****)	*** (****)	*** (****)	•••• (•••••)	*** (****)	•••• (•••••)
Midwest									
1 to 5 years	62	278	18	45 (11.3)	23 (8.2)	23 (9.8)	8 (12.4)	1 (0.8)	222 (22.9)
6 to 10 years	56	294	19	39 (15.2)1	41 (15.2)!	14 (6.6)!	6 (6.1)!	0†(0.4)!	231 (15.7)1
More than 10 years	181	940	62	44 (6.0)	22 (6.3)	20 (5.3)	10 (2.2)	4 (2.7)	232 (11.7)
Nation									
1 to 5 years	568	3,998	22	56 (3.2)	20 (3.1)	16 (2.3)	7 (1.5)	2 (1.2)	201 (5.6)
6 to 10 years	482	3,184	17	57 (3.0)	22 (2.9)	15 (2.6)	5 (1.0)	1 (0.9)	204 (5.2)
More than 10 years	1,565	11,266	61	46 (2.0)	23 (2.4)	21 (1.7)	9 (0.9)	2 (0.6)	221 (3.5)

n = sample size; WGT N = population size estimate / 1,000 (the sample sizes for subpopulations may not add up to the total sample sizes, due to missing data); PCT = percentage in group; RPCT = row percentage estimate; PROF = average proficiency estimate; (SE) = standard error of the estimate (the reported sample estimate can be said to be within 2 standard errors of the true population value with 95% confidence).

Percentages less than 0.5 are rounded to zero.

*** Sample size is insufficient to permit a reliable estimate (fewer than 45 respondents).

I Interpret with caution -- the nature of the sample does not allow accurate determination of the variability of this statistic.

Source: Educational Testing Service, State Adult Literacy Survey, 1992.

The literacy skills of immigrants who have lived in the United States for six to 10 years tend to be similar to those of immigrants who have lived here for one to five years. Foreign-born adults who have lived in this country for more than 10 years, however, outperformed individuals who have lived in this country for fewer years. Across the scales, foreign-born adults who had been in this country the longest had average proficiency scores that were approximately 20 points higher than those of immigrants who had lived in the United States for fewer years. Furthermore, they were less likely to perform in the lowest level on each literacy scale. For example, about half of the foreign-born adults who had lived in this country for more than 10 years performed in the lowest level of prose literacy, compared with 61 percent of the more recent immigrants.

40.... Profiles of Adult Literacy in Iowa

Results for Adults in Different Racial/Ethnic Groups

Ninety-six percent of the adults in Iowa are White, 2 percent are African American, 2 percent are Latino, and about 1 percent are Asian/Pacific Islander (Table 1.6P,D,Q). In racial and ethnic terms, the Iowa population is less diverse than the national and regional populations. Nationwide, 76 percent of the adults are White, 11 percent are African American, 10 percent are Latino, 2 percent are Asian/Pacific Islander, and 1 percent are in other racial or ethnic groups (such as American Indian). In contrast, 85 percent of the adults in the Midwest are White, 9 percent are African American, 4 percent are Latino, 1 percent are Asian/Pacific Islander, and 1 percent are in other racial/ethnic groups.

Because most of the adult population in Iowa is White, the performance results for White adults are nearly identical to those for the Iowa adult population as a whole. In both the region and nation, White adults were less likely than African American or Latino adults to demonstrate limited English literacy skills (that is, to perform in Levels 1 and 2) and more likely to demonstrate advanced skills (that is, to attain Levels 4 and 5). Across the three scales, 13 to 16 percent of the White respondents performed in the lowest level of literacy, compared with 33 to 46 percent of the African American respondents and 40 to 50 percent of the Latino respondents.

In the Midwest and nation, 34 to 41 percent of African American adults and 25 to 31 percent of Latino adults performed in the second lowest proficiency level. At the other end of the performance spectrum, only 3 to 4 percent of the African American adults and 6 to 7 percent of the Latino adults reached Levels 4 and 5.

These racial/ethnic differences in literacy are repeated in the average prose proficiency results, where White individuals generally performed better than African American individuals, who generally performed better than Latino individuals. The average prose score of White adults nationwide was 286, which lies at the low end of the Level 3 range. For African American adults, it was 237, which lies in the low end of the Level 2 range. For Latino adults, it was 215 — within the range for Level 1.

Performance gaps among the racial/ethnic groups are also found on the other literacy scales. The average score difference between White and African American adults nationwide is larger on the quantitative scale (63 points) than on the prose (49 points) or document scale (50 points). Similarly, the score difference between White and Latino adults is larger on the quantitative (75 points) and prose (71 points) scales than on the document scale (67 points).



IOWA TABLE 1.6P

Prose Literacy Levels and Average Proficiencies, by Race/Ethnicity: Results for Iowa, the Midwest, and the Nation

RACE/ETHNICITY				Percent	age of adults in	each prose lite	racy level	
			Level 1 225 or lower	Level 2 226 to 275	Level 3 276 to 325	Level 4 326 to 375	Level 5 376 or higher	Average Proficiency
	n	WGT N (/1000)	RPCT (SE)	RPCT (SE)	RPCT (SE)	RPCT (SE)	RPCT (SE)	PROF (SE)
lowa								
White	1 180	2.003	14 (3.5)	24 (25)	38 (5 3)	21/17)	3/08	286 / 2 1)
African American	26	34	*** (****)	*** (****)	*** (****)	*** (****)	*** (****)	*** (****)
Latino	22	34	+++ (++++)	*** (****)	*** (****)	*** / ****	*** / ****	*** . ****
Mexican	13	22	*** (****)	*** (****)		*** (****)	*** / ****	*** / ****
Puerto Rican	i o	0	*** (****)	*** (****)	*** (****)	*** (****)	*** (****)	*** / ****
Cuban	ŏ	ŏ	*** (****)	*** (****)	*** (****)	*** (*****	*** (****)	
C./S. American	3	5	*** (****)	*** (****)	*** / ****	*** / ****		*** (****)
Other	6	7	*** (****)	*** (****)	*** (****)	*** (****	*** (****)	*** (****)
Asian/Pacific Islander	9	12	*** (****)	*** (****)	*** (****)			*** (****
Other	9	12	*** (****)	*** } ****	*** (****)	*** (****)		(
Midwoot				1 1		. ,		1 /
Midwest			10000		1000 0000		- and the	
White	5,877	38,530	13 (0.9)	26 (1.1)	37 (1.1)	20 (0.9)	3 (0.4)	286 (1.2)
African American	1,161	4,222	33 (1.8)	41 (2.8)	22 (3.2)	4 (1.0)	0 [†] (0.3)	245 (2.1)
Latino	346	1,703	42 (4.0)	26 (3.3)	25 (3.8)	6 (2.5)	01(0.3)	232 (5.2)
Mexican Duarte Diago	213	1,058	48 (6.0)	25 (5.3)	21 (6.0)	5 (3.2)	01(0.3)	221 (8.7)
Puerto Rican	10	222	42 (10.1)	38 (10.5)	20 (6.7)1	01(0.4)!	01(0.9)1	226 (7.9)
Cuban C/C American	4	20	*** (****)		()			
C./S. American	34	205	*** (****		()	()		()
Asian/Pasifia Islandar	20	193	22 (10 2)		()			()
Other	49	202	33 (10.3)	28 (10.4)	25 (10.2)	11 (5.1)	3 (2.1)!	236 (13.8)!
Other	01	501	10 (12.3)1	44 (10.4)1	32 (15.2)1	6 (1.7)1	1 (1.7)	258 (20.1)1
Nation			Sec. 200					
White	17,292	144,968	14 (0.4)	25 (0.6)	36 (0.8)	21 (0.5)	4 (0.3)	286 (0.7)
African American	4,963	21,192	38 (1.1)	37 (1.3)	21 (1.0)	4 (0.5)	0†(0.1)	237 (1.4)
Latino	3,126	18,481	49 (1.4)	26 (1.4)	19 (1.4)	6 (0.8)	1 (0.3)	215 (2.2)
Mexican	1,776	10,235	54 (1.9)	25 (1.6)	16 (1.3)	5 (0.8)	0†(0.3)	206 (3.2)
Puerto Rican	405	2,190	47 (5.0)	32 (5.5)	17 (3.6)	3 (1.7)	0†(0.3)	218 (6.1)
Cuban	147	928	53 (6.7)	24 (7.0)	17 (4.2)	6 (4.7)	1 (2.1)	211 (8.7)
C./S. American	424	2,608	56 (3.8)	22 (3.4)	17 (3.9)	4 (1.5)	0†(0.3)	207 (5.8)
Other	374	2,520	25 (3.2)	27 (5.9)	33 (5.2)	13 (3.4)	2 (1.6)	260 (5.3)
Asian/Pacific Islander	438	4,116	36 (4.4)	25 (3.8)	25 (3.1)	12 (1.9)	2 (0.7)	242 (6.7)
Other	272	2,532	33 (5.7)!	35 (5.5)!	24 (7.5)!	7 (2.4)1	1 (1.0)!	242 (7.0)!

n = sample size; WGT N = population size estimate / 1,000 (the sample sizes for subpopulations may not add up to the total sample sizes, due to missing data); RPCT = row percentage estimate; PROF = average proficiency estimate; (SE) = standard error of the estimate (the

reported sample estimate can be said to be within 2 standard errors of the true population value with 95% confidence).

Percentages less than 0.5 are rounded to zero.

1

ŧ,

*** 1

Sample size is insufficient to permit a reliable estimate (fewer than 45 respondents). Interpret with caution -- the nature of the sample does not allow accurate determination of the variability of this statistic.

Source: Educational Testing Service, State Adult Literacy Survey, 1992.





IOWA TABLE 1.6D

Document Literacy Levels and Average Proficiencies, by Race/Ethnicity: Results for Iowa, the Midwest, and the Nation

RACE/ETHNICITY				Percentage	e of adults in ea	ch document lit	eracy level	
			Level 1 225 or lower	Level 2 226 to 275	Level 3 276 to 325	Level 4 326 to 375	Level 5 376 or higher	Average Proficiency
	n	WGT N (/1000)	RPCT (SE)	RPCT (SE)	RPCT (SE)	RPCT (SE)	RPCT (SE)	PROF (SE)
Lowa White African American Latino Mexican Puerto Rican Cuban C./S. American Other Asian/Pacific Islander Other Mhite African American Latino Mexican Puerto Rican Cuban C./S. American Cuban C./S. American Other Asian/Pacific Islander	1,180 26 22 13 0 0 3 6 9 9 9 9 5,877 1,161 346 213 70 4 34 25 49 61	2,003 34 34 22 0 0 5 7 12 12 12 38,530 4,222 1,703 1,058 222 26 205 193 282 581	$ \begin{array}{c} 15 (3.2) \\ \cdots (\cdots) \\ 16 (0.8) \\ 38 (2.3) \\ 40 (4.0) \\ 43 (5.9) \\ 41 (9.3)! \\ \cdots (\cdots) \\ 41 (9.3)! \\ \cdots (\cdots) \\ \cdots (\cdots) \\ 28 (12.6)! \\ 25 (5.2)! \\ \end{array} $	27 (2.0) *** (****) *** (****)	36 (3.3) *** (****) *** (****) 35 (1.7) 17 (1.5) 23 (5.8) 19 (6.7) 25 (8.1)! *** (****) *** (****) *** (****) 25 (8.1)! *** (****) *** (****)	19 (1.9) *** (****) *** (****) 6 (2.3) 2 (2.9)! *** (****) *** (****)	$\begin{array}{c} 2 (1.0) \\ & \cdots \\ & (\cdots) \\ & 0^{\dagger} (0.4) \\ & 0^{\dagger} (0.5) \\ & 0^{\dagger} (0.0) \\ & \cdots \\ & (\cdots) \\ & 1 (2.6) 1 \end{array}$	281 (2.9) *** (****) *** (****) 280 (1.3) 237 (2.3) 232 (5.9) 223 (9.4) 233 (8.8)! *** (****) *** (****) *** (****) 246 (13.3)! 253 (12.1)!
NationWhiteAfrican AmericanLatinoMexicanPuerto RicanCubanC./S. AmericanOtherAsian/Pacific IslanderOther	17,292 4,963 3,126 1,776 405 147 424 374 438 272	144,968 21,192 18,481 10,235 2,190 928 2,608 2,520 4,116 2,532	16 (0.5) 43 (1.0) 50 (1.7) 54 (2.1) 49 (3.8) 48 (8.1) 53 (3.9) 28 (3.0) 34 (3.5) 34 (5.7)!	27 (0.6) 36 (1.2) 26 (1.6) 25 (1.9) 29 (5.1) 30 (6.2) 25 (3.8) 26 (3.6) 25 (3.6) 33 (4.4)!	34 (0.7) 18 (0.9) 18 (1.4) 16 (1.6) 18 (2.6) 16 (4.3) 16 (3.6) 32 (4.4) 28 (3.7) 25 (4.8)!	19 (0.5) 3 (0.4) 5 (0.8) 4 (0.8) 3 (1.1) 4 (3.9) 4 (1.5) 12 (4.4) 12 (2.3) 7 (2.8)!	$\begin{array}{c} 3 (0.2) \\ 0^{\dagger} (0.1) \\ 1 (0.3) \\ 0^{\dagger} (0.2) \\ 0^{\dagger} (0.2) \\ 0^{\dagger} (0.3) \\ 2 (1.2) \\ 0^{\dagger} (0.5) \\ 2 (1.8) \\ 2 (0.9) \\ 1 (0.7)! \end{array}$	280 (0.8) 230 (1.2) 213 (2.5) 205 (3.5) 215 (6.6) 212 (11.3) 206 (5.5) 254 (5.3) 245 (5.6) 243 (7.6)!

n = sample size; WGT N = population size estimate / 1,000 (the sample sizes for subpopulations may not add up to the total sample sizes, due to missing data); RPCT = row percentage estimate; PROF = average proficiency estimate; (SE) = standard error of the estimate (the reported sample estimate can be said to be within 2 standard errors of the true population value with 95% confidence).

1

Percentages less than 0.5 are rounded to zero. Sample size is insufficient to permit a reliable estimate (fewer than 45 respondents). Interpret with caution -- the nature of the sample does not allow accurate determination of the variability of this statistic. 1

Source: Educational Testing Service, State Adult Literacy Survey, 1992.



IOWA TABLE 1.6Q

Quantitative Literacy Levels and Average Proficiencies, by Race/Ethnicity: Results for Iowa, the Midwest, and the Nation

RACE/ETHNICITY				Percentage	e of adults in ea	ch quantitative	literacy level	
	4-		Level 1 225 or lower	Level 2 226 to 275	Level 3 276 to 325	Level 4 326 to 375	Level 5 376 or higher	Average Proficiency
	n	WGT N (/1000)	RPCT (SE)	RPCT (SE)	RPCT (SE)	RPCT (SE)	RPCT (SE)	PROF (SE)
lowa								
White	1,180	2,003	14 (3.2)	22 (2.1)	36 (3.4)	23 (2.0)	4 (1.1)	289 (3.6)
African American	26	34	*** (****)	*** (****)	*** (****)	*** (****)	*** (****)	*** (****)
Latino	22	34	*** (****)	*** (****)	*** (****)	*** (****)	()	*** (****)
Mexican	13	22	*** (****)	*** (****)	*** (****)	*** (****)	*** (****)	*** (****)
Puerto Rican	0	0	**** (*****)	*** (****)	•••• (•••••)	**** (*****)	*** (****)	**** (*****)
Cuban	0	0		•••• (•••••)	*** (****)	()	*** (****)	()
C./S. American	3	5	(*** (****)	()	*** (****)	*** (****)	()
Other	6	7	()		()	()	()	
Asian/Pacific Islander	9	12	*** (****)	*** (****)				
Other	9	14	1 1 1	()		()	(-)	(-)
Midwest	Sec.		Nu states 1					
White	5,877	38,530	13 (0.8)	25 (1.5)	36 (1.4)	22 (1.0)	4 (0.4)	288 (1.5)
African American	1,161	4,222	43 (2.8)	36 (3.4)	17 (2.4)	3 (1.2)	0 [†] (0.1)	231 (2.8)
Latino	346	1,703	40 (5.4)	30 (7.3)	24 (5.9)	6 (3.0)	1 (0.6)	231 (7.3)
Mexican	213	1,058	42 (7.7)	29 (9.7)	24 (7.6)	4 (3.5)	1 (0.8)	225 (10.5)
Puerto Rican	70	222	42 (6.4)!	31 (10.5)	24 (8.5)!	3 (1.7)!	0 [†] (0.0)	229 (8.1)
Cuban	4	26		()	()			
C./S. American	34	205	*** (****	()	()	()		
Asian/Pacific Islandor	20	193	22 (11 0)	20 (0.5)	00 (7 0)	17 . 0.01	21001	DEA HE EN
Other	61	581	30 (15.8)	32 (16.3)	31 (26 5)	6 (4 7)	01(18)	253 (30.6)
Mallan					01 (20.0)1	0 (4.7)	0 (1.0)	200 (00.0).
Nation	17.000		241 8 8			122 1 2 2 2		
White	17,292	144,968	14 (0.5)	24 (0.6)	35 (0.7)	21 (0.4)	5 (0.2)	287 (0.8)
Amcan American	4,903	21,192	46 (1.0)	34 (1.1)	17 (1.0)	3 (0.4)	0'(0.1)	224 (1.4)
Latino	1 776	10,401	50 (1,3)	25 (1.3)	19 (1.3)	5 (1.1)	1 (0.2)	212 (2.5)
Puorto Ricon	405	2 100	51 (9.2)	25 (2.0)	17 (2.0)	4 (0.8)	0'(0.2)	205 (3.0)
Cuban	147	028	46 (6 4)	20 (4.0)	25 (5 2)	3 (1.3)	2 (2.5)	223 (12.9)
C/S American	424	2 608	53 (3.7)	25 (4.1)	18 (2.9)	4 (1 5)	ot(0.4)	203 (57)
Other	374	2,520	31 (3.0)	25 (4.6)	31 (3 1)	11 (47)	1(07)	246 (6.9)
Asian/Pacific Islander	438	4,116	30 (3.9)	23 (3.4)	27 (3.0)	16 (2 4)	4(17)	256 (6.7)
Other	272	2,532	38 (4.9)	29 (5.5)	26 (4.5)	6(29)	1 (0.8)	241 (5.5)
				a surf.	and we have	S.I. Siete		

n = sample size; WGT N = population size estimate / 1,000 (the sample sizes for subpopulations may not add up to the total sample sizes, due to missing data); RPCT = row percentage estimate; PROF = average proficiency estimate; (SE) = standard error of the estimate (the reported sample estimate can be said to be within 2 standard errors of the true population value with 95% confidence).

Percentages less than 0.5 are rounded to zero.

1

*** 1

Sample size is insufficient to permit a reliable estimate (fewer than 45 respondents). Interpret with caution -- the nature of the sample does not allow accurate determination of the variability of this statistic.

Source: Educational Testing Service, State Adult Literacy Survey, 1992.



Results for adults in different racial/ethnic groups, by country of birth

To better understand the differences in performance among various racial/ ethnic groups, it is helpful to examine the percentages of adults in each group who were born inside and outside this country. In Iowa, as in the Midwest and nation, nearly all White adults were born in the United States (Table 1.7). The numbers of Iowa adults in various racial/ethnic groups are too small to support analysis. Nationwide, however, a vast majority of the African American adults (95 percent) were born in the United States or a U.S. territory, but nearly half of the Latino adults and more than three-quarters of the Asian/Pacific Islander adults were born abroad.

Regardless of the racial/ethnic group to which they belong, adults born in the United States tended to display higher literacy proficiencies in English than did foreign-born adults (Table 1.8P,D,Q). Again, the numbers of Iowa adults in different racial/ethnic groups are too small to provide stable estimates. Nationwide, however, the average prose score of native-born Latino adults was 252, while for those born abroad it was 175 — 77 points lower. Similar patterns are found on the document and quantitative scales.

Indeed, when the differences in literacy proficiency among various racial/ ethnic groups are viewed through the lens of country of birth, the pattern of results seen in Table 1.6P,D,Q changes substantially. In the Midwest and across the nation, Latino adults born in the United States had higher average literacy scores than African American adults, virtually all of whom were born here. Further, when one takes country of birth into consideration, the proficiency differences between White and Latino adults diminish sharply on all three literacy scales. While in the national population the average scores of these two groups differ by 67 to 75 points on each scale, the difference is reduced to between 32 and 41 points among native-born individuals.

Results for Adults by the Number of Years Lived in Iowa

Decision makers in Iowa were interested in gathering information on the percentages of state residents who had lived in Iowa for various lengths of time. Accordingly, one of the survey questions administered to Iowa residents asked how many years they had lived in the state: less than one year, one to five years, six to 10 years, 11 to 15 years, and 15 to 20 years. Due to the small number of respondents in some of these groups, they were collapsed into three categories for reporting purposes.



IOWA TABLE 1.7

Country of Birth, by Race/Ethnicity: Results for Iowa, the Midwest, and the Nation

			Percentage of adults with ea	ach country of birth
			United States or U.S. territory	Other country
	n	WGT N (/1000)	RPCT (SE)	RPCT (SE)
		1		
owa	0			
White	1,180	2,003	99 (0,4)	1 (0.4)
African American	26	34	*** (****)	*** (****)
Latino (all)	22	34	*** (****)	*** (****)
Mexican	13	22	*** (****)	*** (****)
Puerto Rican	0	0	*** (****)	*** (****)
Cuban	0	0	*** (****)	*** (****)
C./S. American	3	5	*** (****)	*** (****)
Other	6	7	*** (****)	*** (****)
Asian/Pacific Islander	9	12	*** (****)	*** (****)
Other	9	12	()	*** (****)
idwest				
White	5,877	38,530	98 (0.3)	2 (0.3)
African American	1,161	4,222	98 (0.6)	2 (0.6)
Latino (all)	346	1,703	66 (4.6)	34 (4.6)
Mexican	213	1,058	58 (8.5)	42 (8.5)
Puerto Rican	70	222	95 (2.8)	5 (2.8)
Cuban	4	26	•••• (•••••)	*** (****)
C./S. American	34	205	*** (****)	*** (****)
Other	25	193	*** (****)	*** (****)
Asian/Pacific Islander	49	282	15 (7.1)	85 (7.1)
Other	61	581	97 (6.0)	3 (6.0)
ation				
White	17,292	144,968	96 (0.2)	4 (0.2)
African American	4,963	21,192	95 (0.5)	6 (0.5)
atino (all)	3,126	18,481	52 (1.8)	48 (1.8)
Mexican	1,776	10,235	54 (2.2)	46 (2.2)
Puerto Rican	405	2,190	80 (2.9)	20 (2.9)
Cuban	147	928	11 (2.8)	89 (2.8)
C./S. American	424	2,608	21 (3.1)	79 (3.1)
Other	374	2,520	68 (5.5)	32 (5.5)
Asian/Pacific Islander	438	4,116	22 (2.5)	78 (2.5)

Other	272 2,5	532 78 (6.6)	22 (6.6)

- n = sample size; WGT N = population size estimate / 1,000 (the sample sizes for subpopulations may not add up to the total sample sizes, due to missing data); RPCT = row percentage estimate; (SE) = standard error of the estimate (the reported sample estimate can be said to be within 2 standard errors of the true population value with 95% confidence).

÷

- Percentages less than 0.5 are rounded to zero.
 Sample size is insufficient to permit a reliable estimate (fewer than 45 respondents).
- Source: Educational Testing Service, State Adult Literacy Survey, 1992.

46 Profiles of Adult Literacy in Iowa

IOWA TABLE 1.8P

Average Prose Literacy Proficiencies, by Country of Birth and Race/Ethnicity: Results for Iowa, the Midwest, and the Nation

			Average prose proficiency of adults with each country of birth					
		[United States or U.S. territory	Other country				
	n	WGT N (/1000)	PROF (SE)	PROF (SE)				
owa								
White	1 180	2.003	286 (3.1)	*** (****)				
African American	26	34	*** (****)	*** (****)				
Latino (all)	22	34	*** (****)	*** (****)				
Movican	13	22	*** (****)	*** (****)				
Puorte Picen	0	0	*** (****)	*** (****)				
Cuban	0	0	*** (****)	*** (****)				
C/S Amorican	2	5	*** (****)	*** (****)				
Othor	6	7	*** (****)	*** (****)				
Acien/Decific Islander	0	12	*** (****)	*** (****)				
Other	9	12	*** (****)	*** (****)				
Midwest				054 (11 0)				
White	5,877	38,530	286 (1.2)	254 (11.0)!				
African American	1,161	4,222	244 (1.9)					
Latino (all)	346	1,703	260 (5.5)!	177 (12.0)!				
Mexican	213	1,058	262 (7.7)!	164 (9.0)!				
Puerto Rican	70	222	227 (9.5)!	*** (****)				
Cuban	4	26	*** (****)	*** (****)				
C./S. American	34	205	*** (****)	*** (****)				
Other	25	193	*** (****)					
Asian/Pacific Islander	49	282	*** (****)					
Other	61	581	257 (26.7)!	•••• (•••••)				
Nation	17 202	144.969	287 (0.8)	258 (4.3)				
African American	1,252	21 102	237 (1.4)	230 (6.4)				
American American	4,500	18 /81	252 (2 4)	175 (2.7)				
Latino (all)	1 776	10,401	246 (3.2)	158 (3.7)				
Ruedo Bison	1,770	2 100	226 (6.9)	186 (10.3)!				
Cuber	403	028	*** (****)	202 (10.9)				
Cuban C/S Amoricon	147	2 608	281 (6.3)	187 (6.0)				
Other	274	2,000	283 (7 7)	210 (10.5)!				
Asian/Pacific Islandor	129	4 116	274 (11.2)	233 (7.2)				
Asian/Pacific Islander	430	2 532	254 (4 6)	198 (16,2)!				

n = sample size; WGT N = population size estimate / 1,000 (the sample sizes for subpopulations may not add up to the total sample sizes, due to missing data); PROF = average proficiency estimate; (SE) = standard error of the estimate (the reported sample estimate can be said to be within 2 standard errors of the true population value with 95% confidence).

- t
- ***
- Percentages less than 0.5 are rounded to zero. Sample size is insufficient to permit a reliable estimate (fewer than 45 respondents). Interpret with caution -- the nature of the sample does not allow accurate determination of the variability of this statistic. 1

Source: Educational Testing Service, State Adult Literacy Survey, 1992.



IOWA TABLE 1.8D

Average Document Literacy Proficiencies, by Country of Birth and Race/Ethnicity: Results for Iowa, the Midwest, and the Nation

			Average document proficiency of adu	Its with each country of birth
			United States or U.S. territory	Other country
	n	WGT N (/1000)	PROF (SE)	PROF (SE)
lowa				
White	1,180	2,003	281 (3.0)	*** (****)
African American	26	34	*** (****)	*** (****)
Latino (all)	22	34	*** (****)	*** (****)
Mexican	13	22	*** (****)	*** (****)
Puerto Rican	0	0	*** (****)	*** (****)
Cuban	0	0	*** (****)	*** (****)
C./S. American	3	5	*** (****)	*** (****)
Other	6	7	*** (****)	*** (****)
Asian/Pacific Islander	9	12	*** (****)	*** (****)
Other	9	12	*** (****)	*** (****)
Midwest				
White	5.877	38,530	281 (1.3)	252 (12 7)
African American	1,161	4,222	236 (1.9)	*** (****)
Latino (all)	346	1,703	255 (6.2)	186 (10.6)
Mexican	213	1,058	261 (9.2)	172 (10.5)
Puerto Rican	70	222	233 (10,1)	*** (****)
Cuban	4	26	*** (****)	*** (****)
C./S. American	34	205	*** (****)	*** (****)
Other	25	193	*** (****)	*** (****)
Asian/Pacific Islander	49	282	*** (****)	*** (****)
Other	61	581	253 (12.8)!	*** (****)
Nation	Con-			
White	17,292	144,968	281 (0.9)	255 (3 3)
African American	4,963	21,192	230 (1.2)	225 (8.7)
Latino (all)	3,126	18,481	249 (2.4)	174 (3.2)
Mexican	1,776	10,235	245 (3.0)	158 (4 3)
Puerto Rican	405	2,190	225 (6.7)	171 (12 4)
Cuban	147	928	*** (****)	204 (13.0)
C./S. American	424	2,608	277 (5.0)	188 (5.9)
Other	374	2,520	277 (7.5)	204 (11,1)
Asian/Pacific Islander	438	4,116	266 (12,4)!	240 (5.4)



- n = sample size; WGT N = population size estimate / 1,000 (the sample sizes for subpopulations may not add up to the total sample sizes, due to missing data); PROF = average proficiency estimate; (SE) = standard error of the estimate (the reported sample estimate can be said to be within 2 standard errors of the true population value with 95% confidence).
- † Percentages less than 0.5 are rounded to zero.

•••

- *** Sample size is insufficient to permit a reliable estimate (fewer than 45 respondents).
- 1 Interpret with caution -- the nature of the sample does not allow accurate determination of the variability of this statistic.

Source: Educational Testing Service, State Adult Literacy Survey, 1992.

48.... Profiles of Adult Literacy in Iowa

IOWA TABLE 1.8Q

Average Quantitative Literacy Proficiencies, by Country of Birth and Race/Ethnicity: Results for Iowa, the Midwest, and the Nation

			Average quantitative proficiency of adults with each country of birth	
			United States or U.S. territory	Other country
	n	WGT N (/1000)	PROF (SE)	PROF (SE)
lowa			the second se	
White	1 180	2 003	289 (3.7)	*** (****)
African American	26	34	*** (****)	*** (****)
Latino (all)	22	34	*** (****)	*** (****)
Movican	13	22	*** (****)	*** (****)
Puerto Bican	0	0	*** (****)	*** (****)
Cuban	0	Ő	*** (****)	*** (****)
CUDan C/S Amorican	0 0	5	*** (****)	*** (****)
Other	6	7	*** (****)	*** (****)
Asian/Dasifia Islandor	0	12	*** (****)	*** (****)
Other	9	12	*** (****)	*** (****)
Midwest	-			
White	5,877	38,530	288 (1.5)	255 (15.9)!
African American	1,161	4,222	230 (2.6)	
Latino (all)	346	1,703	255 (8.0)!	185 (10.7)!
Mexican	213	1,058	262 (9.5)!	174 (8.9)!
Puerto Rican	70	222	229 (10.0)!	
Cuban	4	26	*** (****)	
C./S. American	34	205		
Other	25	193	(
Asian/Pacific Islander	49	282		
Other	61	581	252 (31.6)!	
Nation				000 (4 0)
White	17,292	144,968	288 (0.8)	260 (4.2)
African American	4,963	21,192	224 (1.4)	227 (7.1)
Latino (all)	3,126	18,481	247 (2.7)	173 (3.0)
Mexican	1,776	10,235	244 (3.1)	158 (4.5)
Puerto Rican	405	2,190	223 (6.6)	166 (16.0)!
Cuban	147	928	*** (****)	217 (14.6)
C./S. American	424	2,608	275 (5.1)!	185 (6.4)
Other	374	2,520	271 (8.2)	191 (13.1)!
Asian/Pacific Islander	438	4,116	279 (10.0)!	249 (7.9)
Other	272	2,532	252 (5.4)!	203 (12.2)!

n = sample size; WGT N = population size estimate / 1,000 (the sample sizes for subpopulations may not add up to the total sample sizes, due to missing data); PROF = average proficiency estimate; (SE) = standard error of the estimate (the reported sample estimate can be said to be within 2 standard errors of the true population value with 95% confidence).

Section I 49

t

Percentages less than 0.5 are rounded to zero. Sample size is insufficient to permit a reliable estimate (fewer than 45 respondents). ***

Interpret with caution -- the nature of the sample does not allow accurate determination of the variability of this statistic. 1

Source: Educational Testing Service, State Adult Literacy Survey, 1992.
Approximately three-quarters of the Iowa survey respondents reported having lived in the state for more than 20 years (Table 1.9). Fourteen percent said they had lived in Iowa for 11 to 20 years, and 11 percent said they had lived there for 10 years or less.

In comparing the performance of adults who had lived in the state for various lengths of time, we find no statistically significant differences in the prose, document, or quantitative literacy scores among the various groups.



IOWA TABLE 1.9

Prose, Document, and Quantitative Literacy Levels and Average Proficiencies, by Years Lived in Iowa: Results for Iowa

-					_		Perc	entage o	f adult.	s in each	literad	cy level			
Long Street				Lev 225 or	el 1 Iower	Lev 226 t	el 2 o 275	Lev 276 to	el 3 o 325	Lev 326 to	el 4 o 375	Lev 376 or	el 5 higher	Aver Profic	age lency
	n	WGT N (/1000)	PCT	RPCT (SE)	RPCT (SE)	RPCT (SE)	RPCT (SE)	RPCT (SE)	PROF (SE)
Prose															
10 years or less	160	204	11	8 (4.9)	24 (5.6)	36 (7.4)	25 (7.4)	6(24)	297 (10.0)
11 to 20 years	153	257	14	6(3.1)	22 (6.2)	47 (7.9)	23 (6.5)	21	1.6)	297 (6.7)
More than 20 years	905	1,313	74	8 (2.0)	24 (3.7)	41 (2.6)	24 (3.2)	3(1.1)	295 (6.4)
Document												1		1	
10 years or less	160	204	11	9(6.5)	25 (4.6)	40 (5.4)	21 (4.7)	4(3.6)	293 (*	10.8)
11 to 20 years	153	257	14	6 (3.4)	21 (4.3)	45 (4.8)	26 (5.9)	21	1.3)	299 (6.4)
More than 20 years	905	1,313	74	9 (2.8)	27 (3.3)	40 (2.9)	21 (3.1)	3(1.3)	290 (6.8)
Quantitative															
10 years or less	160	204	11	9(6.1)	23 (4.0)	37 (6.3)	26 (6.3)	6(2.3)	297 (9.8)
11 to 20 years	153	257	14	6 (3.5)	21 (4.9)	47 (5.7)	23 (4.6)	41	1.9)	297 (5.9)
More than 20 years	905	1,313	74	8 (2.7)	211	3.4)	39 (3.3)	271	3.5)	51	1.8)	299 (7.2)

n = sample size; WGT N = population size estimate / 1,000 (the sample sizes for subpopulations may not add up to the total sample sizes, due to missing data); PCT = percentage in group; RPCT = row percentage estimate; PROF = average proficiency estimate; (SE) = standard error of the estimate (the reported sample estimate can be said to be within 2 standard errors of the true population value with 95% confidence).

*** Sample size is insufficient to permit a reliable estimate (fewer than 45 respondents).

Source: Educational Testing Service, State Adult Literacy Survey, 1992.



Results for Adults by Their Likelihood of Moving Out of Iowa in the Next Five Years

Decision makers in Iowa also wished to collect information on the proportions of state residents who believed they might move out of Iowa in the next five years. Accordingly, as part of the survey, Iowa residents were asked what the likelihood was that they would move out of the state in the next five years.

Nearly three-quarters of the Iowa respondents said it was unlikely they would move out of the state in the next five years (Table 1.10). Eighteen percent reported that it was somewhat likely, and 10 percent said it was very likely.



IOWA TABLE 1.10

Prose, Document, and Quantitative Literacy Levels and Average Proficiencies, by Likelihood of Moving Out of Iowa in the Next Five Years: Results for Iowa

					Perce	entage of adults	in each literac	y level	
				Level 1 225 or lower	Level 2 226 to 275	Level 3 276 to 325	Level 4 326 to 375	Level 5 376 or higher	Average Proficiency
	n	WGT N (/1000)	PGT	RPCT (SE)	RPCT (SE)	RPCT (SE)	RPCT (SE)	RPCT (SE)	PROF (SE)
Prose Not likely Somewhat likely Very likely	881 219 118	1,285 313 175	72 18 10	9 (1.9) 5 (2.5) 7 (5.3)	24 (3.7) 20 (5.2) 23 (6.1)	41 (2.3) 40 (5.9) 41 (7.2)	22 (3.2) 30 (5.3) 25 (8.5)	3 (1.1) 5 (2.4) 4 (1.9)	293 (6.1) 305 (5.6) 298 (8.1)
Document Not likely Somewhat likely Very likely	881 219 118	1,285 313 175	72 18 10	10 (2.6) 5 (2.3) 10 (7.5)	28 (3.4) 22 (4.1) 18 (5.2)	39 (2.5) 42 (5.3) 45 (8.8)	20 (2.7) 27 (6.1) 24 (5.2)	2 (1.1) 5 (2.6) 3 (3.1)	288 (6.4) 303 (5.3) 297 (9.6)
Quantitative Not likely Somewhat likely Very likely	881 219 118	1,285 313 175	72 18 10	9 (2.2) 5 (3.0) 9 (7.0)	22 (3.4) 19 (3.2) 18 (5.5)	39 (3.1) 42 (4.9) 46 (8.4)	26 (3.2) 28 (5.5) 22 (5.4)	5 (1.8) 6 (3.0) 4 (2.1)	297 (6.7) 306 (5.6) 296 (10.3)

n = sample size; WGT N = population size estimate / 1,000 (the sample sizes for subpopulations may not add up to the total sample sizes, due to missing data); PCT = percentage in group; RPCT = row percentage estimate; PROF = average proficiency estimate; (SE) = standard error of the estimate (the reported sample estimate can be said to be within 2 standard errors of the true population value with 95% confidence).

Section I 51

† Percentages less than 0.5 are rounded to zero.

*** Sample size is insufficient to permit a reliable estimate (fewer than 45 respondents).

I Interpret with caution -- the nature of the sample does not allow accurate determination of the variability of this statistic.

Source: Educational Testing Service, State Adult Literacy Survey, 1992.

Although there appear to be differences in the average prose, document, and quantitative proficiencies of Iowa adults who said that it was unlikely they would move out of the state in the next five years and those who said it was somewhat or very likely, these differences are not statistically significant. Thus, in general, the literacy skills of adults who plan to remain in the state are comparable to the skills of those who plan to leave.

Results for Adults with Physical or Mental Conditions

One of the background questions included in the survey asked respondents whether they had physical or mental conditions that keep them from participating fully in work, school, housework, or other activities. Fifteen percent of the adults in Iowa and 12 percent in the region and the nation reported having such conditions (Table 1.11P,D,Q).



IOWA TABLE 1.11P

Prose Literacy Levels and Average Proficiencies, by Disability Status: Results for Iowa, the Midwest, and the Nation

PHYSICAL OR MENTAL				Percentage of adults in each prose literacy level										
DISABILITY				Level 1 225 or lower	Level 2 226 to 275	Level 3 276 to 325	Level 4 326 to 375	Level 5 376 or higher	Average Proficiency					
	n	WGT N (/1000)	PCT	RPCT (SE)	RPCT (SE)	RPCT (SE)	RPCT (SE)	RPCT (SE)	PROF (SE)					
<u>Iowa</u> Yes No	120 1,126	307 1,788	15 85	37(7.0) 10(2.1)	31 (12.1)! 23 (2.4)	24 (13.6)! 40 (3.2)	6 (4.8)! 23 (2.1)	1 (0.6)! 3 (0.9)	245 (13.9)! 292 (2.9)					
Yes No	739 6,747	5,378 39,906	12 88	44 (2.9) 13 (0.7)	32 (3.7) 27 (1.0)	19 (3.3) 37 (1.5)	6 (1.5) 20 (0.9)	1 (0.5) 3 (0.4)	232 (4.1) 286 (0.9)					
Nation Yes No	2,806 23,256	22,205 168,879	12 88	46 (1.1) 17 (0.4)	30 (1.6) 26 (0.6)	18 (1.5) 34 (0.8)	5 (0.9) 19 (0.5)	1 (0.2) 4 (0.2)	227 (1.6) 278 (0.6)					

- n = sample size; WGT N = population size estimate / 1,000 (the sample sizes for subpopulations may not add up to the total sample sizes, due to missing data); PCT = percentage in group; RPCT = row percentage estimate; PROF = average proficiency estimate; (SE) = standard error of the estimate (the reported sample estimate can be said to be within 2 standard errors of the true population value with 95% confidence).
- † Percentages less than 0.5 are rounded to zero.
- "" Sample size is insufficient to permit a reliable estimate (fewer than 45 respondents).
- Interpret with caution the nature of the sample does not allow accurate determination of the variability of this statistic. Source: Educational Testing Service, State Adult Literacy Survey, 1992.

52 Profiles of Adult Literacy in Iowa

IOWA TABLE 1.11D

Document Literacy Levels and Average Proficiencies, by Disability Status: Results for Iowa, the Midwest, and the Nation

PHYSICAL OR				Percentage of adults in each document literacy level											
DISABILITY				Level 1 225 or lower	Level 2 226 to 275	Level 3 276 to 325	Level 4 326 to 375	Level 5 376 or higher	Average Proficiency						
	n	WGT N (/1000)	PCT	RPCT (SE)	RPCT (SE)	RPCT (SE)	RPCT (SE)	RPCT (SE)	PROF (SE)						
<u>lowa</u> Yes No	120 1,126	307 1,788	15 85	39 (11.3)! 12 (1.1)	35 (5.7)! 26 (2.2)	19 (9.0)! 39 (2.1)	6 (3.1)! 21 (2.4)	1 (0.9)! 3 (1.1)	238 (12.0)! 287 (3.3)						
<u>Midwest</u> Yes No	739 6,747	5,378 39,906	12 88	48 (3.3) 15 (0.7)	32 (3.8) 30 (1.1)	15 (2.9) 35 (1.5)	4 (1.8) 18 (1.0)	1 (0.5) 3 (0.4)	223 (4.8) 280 (1.0)						
Nation Yes No	2,806 23,256	22,205 168,879	12 88	51 (1.3) 19 (0.4)	30 (1.2) 28 (0.5)	15 (0.9) 33 (0.6)	4 (0.6) 17 (0.4)	1 (0.2) 3 (0.2)	219 (1.9) 273 (0.6)						

n = sample size; WGT N = population size estimate / 1,000 (the sample sizes for subpopulations may not add up to the total sample sizes, due to missing data); PCT = percentage in group; RPCT = row percentage estimate; PROF = average proficiency estimate; (SE) = standard error of the estimate (the reported sample estimate can be said to be within 2 standard errors of the true population value with 95% confidence).

Percentages less than 0.5 are rounded to zero.

Sample size is insufficient to permit a reliable estimate (fewer than 45 respondents). ***

Interpret with caution -- the nature of the sample does not allow accurate determination of the variability of this statistic.

When the literacy levels and proficiencies of individuals who reported having conditions are compared with those of other adults, sharp contrasts are evident. On each scale, Iowa residents who said they have limiting physical or mental conditions were far more likely than others to perform in the lowest literacy level and far less likely to reach the highest levels. On the document scale, for example, respondents with limiting conditions were approximately three times more likely to perform in Level 1 than were those without such conditions. At the other end of the spectrum, only 7 percent of those who reported having physical or mental conditions performed in the two highest levels of document literacy, compared with 24 percent of the adults without conditions.

Section I 53



IOWA TABLE 1.11Q

Quantitative Literacy Levels and Average Proficiencies, by Disability Status: Results for Iowa, the Midwest, and the Nation

PHYSICAL OR MENTAL				Percentage of adults in each quantitative literacy level											
DISABILITY				Level 1 225 or lower	Level 2 226 to 275	Level 3 276 to 325	Level 4 326 to 375	Level 5 376 or higher	Average Proficiency						
	п	WGT N (/1000)	PCT	RPCT (SE)	RPCT (SE)	RPCT (SE)	RPCT (SE)	RPCT (SE)	PROF (SE)						
lowa															
Yes	120	307	15	35 (5.4)!	35 (9.9)!	21 (8.8)	8 (6 6)	1/111	242 (12 2)						
No	1,126	1,788	85	11 (1.6)	20 (2.4)	38 (2.5)	25 (1.9)	5(1.2)	295 (2.8)						
Midwest									200 (2.0)						
Yes	739	5,378	12	46 (3.0)	26 (2.2)	21 (34)	6(11)	1(08)	000 (0.0)						
No	6,747	39,906	88	13 (0.9)	26 (1.6)	35 (1.4)	21 (1.1)	4(0.4)	220 (0.0)						
Nation									201 (1.0)						
Yes	2,806	22,205	12	49 (1.2)	25 (1.1)	19(12)	6(07)	1(04)	200 (0 4)						
No	23,256	168,879	88	19 (0.5)	25 (0.6)	33 (0.6)	19 (0.4)	4 (0.2)	278 (0.6)						

n = sample size; WGT N = population size estimate / 1,000 (the sample sizes for subpopulations may not add up to the total sample sizes, due to missing data); PCT = percentage in group; RPCT = row percentage estimate; PROF = average proficiency estimate; (SE) = standard error of the estimate (the reported sample estimate can be said to be within 2 standard errors of the true population value with 95% confidence).

Percentages less than 0.5 are rounded to zero. Ŧ.

Sample size is insufficient to permit a reliable estimate (fewer than 45 respondents).

Interpret with caution -- the nature of the sample does not allow accurate determination of the variability of this statistic. L Source: Educational Testing Service, State Adult Literacy Survey, 1992.

As a result of the differences in the distributions of performance for these two groups, the average proficiencies of Iowa respondents who reported having limiting physical or mental conditions were considerably lower than those of individuals who reported no such conditions. Their average document score (238) is in the low end of the Level 2 range, for example, while the average score of other adults is 287 - in the Level 3 range. Similar patterns are found on the prose and quantitative scales.

54 Profiles of Adult Literacy in Iowa

Results for Males and Females

The performance results for men and women differ across the three literacy scales (Table 1.12P,D,Q). Among adults in Iowa, the average prose scores of men and women are nearly the same (286 and 285, respectively). The four-point gap between men (282) and women (278) on the document scale is not statistically significant, nor is the large (18-point) gap on the quantitative scale, where the average score for men was 297, while for women it was 279. The proficiency gaps between men and women differed somewhat among adults in the Midwest region and the nation as a whole. Here, men had higher average quantitative scores than women, but the prose and document scores of the two groups were comparable.



IOWA TABLE 1.12P

Prose Literacy Levels and Average Proficiencies, by Sex: Results for Iowa, the Midwest, and the Nation

SEX				Percentage of adults in each prose literacy level												
				Level 1 225 or lower	Level 2 226 to 275	Level 3 276 to 325	Level 4 326 to 375	Level 5 376 or higher	Average Proficiency							
	ñ	WGT N (/1000)	PCT	RPCT (SE)	RPCT (SE)	RPCT (SE)	RPCT (SE)	RPCT (SE)	PROF (SE)							
<u>lowa</u> Male Female	592 654	1,003 1,093	48 52	14 (2.9) 15 (7.9)	24 (3.3) 25 (2.6)	39 (2.5) 36 (7.8)	20 (2.3) 21 (2.2)	3 (0.9) 3 (0.9)	286 (5.0) 285 (7.8)							
<u>Midwest</u> Male Female	3,331 4,152	21,621 23,645	48 52	17 (1.2) 16 (1.1)	28 (1.3) 28 (1.3)	34 (2.1) 36 (1.0)	18 (1.0) 18 (1.0)	3 (0.6) 3 (0.4)	278 (1.6) 280 (1.6)							
<u>Nation</u> Male Female	11,770 14,279	92,098 98,901	48 52	22 (0.6) 20 (0.5)	26 (0.9) 28 (0.7)	31 (1.2) 33 (0.7)	18 (0.5) 17 (0.5)	4 (0.3) 3 (0.2)	272 (0.9) 273 (0.8)							

n = sample size; WGT N = population size estimate / 1,000 (the sample sizes for subpopulations may not add up to the total sample sizes, due to missing data); PCT = percentage in group; RPCT = row percentage estimate; PROF = average proficiency estimate; (SE) = standard error of the estimate (the reported sample estimate can be said to be within 2 standard errors of the true population value with 95% confidence).

† Percentages less than 0.5 are rounded to zero.

*** Sample size is insufficient to permit a reliable estimate (fewer than 45 respondents).

! Interpret with caution -- the nature of the sample does not allow accurate determination of the variability of this statistic.

Source: Educational Testing Service, State Adult Literacy Survey, 1992.

Section I 55



IOWA TABLE 1.12D

Document Literacy Levels and Average Proficiencies, by Sex: Results for Iowa, the Midwest, and the Nation

SEX					Percenta	ge of adults in e	ach document li	teracy level	
				Level 1 225 or lower	Level 2 226 to 275	Level 3 276 to 325	Level 4 326 to 375	Level 5 376 or higher	Average Proficiency
	n	WGT N (/1000)	PCT	RPCT (SE)	RPCT (SE)	RPCT (SE)	RPCT (SE)	RPCT (SE)	PROF (SE)
<u>lowa</u> Male Female	592 654	1,003 1,093	48 52	14 (2.0) 18 (6.1)	26 (3.8) 28 (2.8)	36 (3.8) 35 (5.9)	21 (2.6) 17 (2.5)	3 (1.2) 2 (0.9)	282 (5.9) 278 (8.3)
<u>Midwest</u> Male Female	3,331 4,152	21,621 23,645	48 52	18 (1.2) 19 (1.2)	29 (1.4) 31 (1.4)	33 (2.0) 33 (1.6)	17 (1.8) 15 (1.1)	3 (0.5) 2 (0.4)	275 (1.9) 272 (1.7)
<u>Nation</u> Male Female	11,770 14,279	92,098 98,901	48 52	23 (0.6) 23 (0.6)	27 (0.5) 30 (0.7)	31 (0.8) 31 (0.6)	17 (0.5) 14 (0.5)	3 (0.2) 2 (0.2)	269 (0.9) 265 (0.9)

n = sample size; WGT N = population size estimate / 1,000 (the sample sizes for subpopulations may not add up to the total sample sizes, due to missing data); PCT = percentage in group; RPCT = row percentage estimate; PROF = average proficiency estimate; (SE) = standard error of the estimate (the reported sample estimate can be said to be within 2 standard errors of the true population value with 95% confidence).

† Percentages less than 0.5 are rounded to zero.

*** Sample size is insufficient to permit a reliable estimate (fewer than 45 respondents).

I Interpret with caution -- the nature of the sample does not allow accurate determination of the variability of this statistic.

Source: Educational Testing Service, State Adult Literacy Survey, 1992.

These performance differences between men and women may be the result of many variables. One factor may be that women tend to live longer than men and that older adults tend to have lower literacy proficiencies than younger adults, as seen earlier in this section. Further, among older individuals, women tend to have fewer years of schooling than men, and lower levels of education are also associated with lower proficiencies.





IOWA TABLE 1.12Q

Quantitative Literacy Levels and Average Proficiencies, by Sex: Results for Iowa, the Midwest, and the Nation

SEX					Percentage	e of adults in eac	ch quantitative li	iteracy level	
				Level 1 225 or lower	Level 2 226 to 275	Level 3 276 to 325	Level 4 326 to 375	Level 5 376 or higher	Average Proficiency
	n	WGT N (/1000)	PCT	RPCT (SE)	RPCT (SE)	RPCT (SE)	RPCT (SE)	RPCT (SE)	PROF (SE)
<u>lowa</u> Male Female	592 654	1,003 1,093	48 52	11 (3.5) 18 (7.6)	20 (3.3) 25 (2.0)	36 (3.3) 35 (6.6)	27 (2.4) 19 (2.2)	6 (1.6) 3 (0.8)	297 (5.3) 279 (8.9)
<u>Midwest</u> Male Female	3,331 4,152	21,621 23,645	48 52	15 (1.4) 19 (1.2)	24 (2.2) 28 (1.5)	34 (1.9) 33 (2.1)	22 (1.2) 17 (1.4)	5 (0.4) 3 (0.5)	285 (1.9) 275 (2.2)
<u>Nation</u> Male Female	11,770 14,279	92,098 98,901	48 52	21 (0.7) 23 (0.5)	23 (0.5) 28 (0.9)	31 (0.6) 31 (1.0)	20 (0.4) 15 (0.6)	5 (0.3) 3 (0.3)	277 (0.9) 266 (0.9)

n = sample size; WGT N = population size estimate / 1,000 (the sample sizes for subpopulations may not add up to the total sample sizes, due to missing data); PCT = percentage in group; RPCT = row percentage estimate; PROF = average proficiency estimate; (SE) = standard error of the estimate (the reported sample estimate can be said to be within 2 standard errors of the true population value with 95% confidence).

† Percentages less than 0.5 are rounded to zero.

*** Sample size is insufficient to permit a reliable estimate (fewer than 45 respondents).

I Interpret with caution -- the nature of the sample does not allow accurate determination of the variability of this statistic.

Source: Educational Testing Service, State Adult Literacy Survey, 1992.

The question, then, is whether young men and women have comparable literacy skills, and the answer is yes. There were no differences in average prose, document, or quantitative proficiency between young men and women (age 21 to 25) who participated in this survey. The performance gap between men and women in the adult population as a whole therefore appears to be associated with age and is not found among younger adults.

Summary

The average prose, document, and quantitative proficiency scores of adults in Iowa were approximately the same as those of adults in the Midwest region, but were higher than those of adults nationwide. In all three of these populations, the average scores on each literacy scale were in either the high end of the Level 2 range (226 to 275) or the low end of the Level 3 range (276 to 325).

Fourteen percent of the adults in Iowa had scores in the lowest level defined on the prose scale, 16 percent were in the lowest level on the document scale, and 15 percent were in the lowest level on the quantitative scale. Those who performed in the Level 1 range were varied with respect to their characteristics as well as their skills. Iowa residents who performed in Level 1, for example, were more likely than adults statewide to have less than a high school education; to be age 65 or older; and to have a limiting physical condition.

Across the three scales, 22 to 27 percent of Iowa adults had scores in the second lowest proficiency level (Level 2). Thirty-six to 37 percent performed in the third level on each scale, and 19 to 23 percent demonstrated skills in the fourth level. Just 2 to 4 percent of the respondents in Iowa, the Midwest, and the nation performed in Level 5 on each literacy scale — the highest proficiency level defined in the survey.

Older adults (those age 55 to 64 and age 65 and older) were more likely than younger adults to perform in the two lowest levels on each scale. Among the younger age groups, the differences in the percentages of individuals who performed in each level are relatively small, but individuals in the middle age categories were more likely than those in both the younger and the older age groups to reach the highest proficiency levels.

As expected, adults born in the United States tended to be more proficient in English than individuals born abroad, many of whom have learned English as a second language. The number of foreign-born adults in Iowa is too small to provide reliable performance estimates. Nationwide, however, the literacy skills of immigrants who have lived in the United States for six to 10 years were similar to those of immigrants who have lived here for one to five years. Foreign-born adults who have lived in this country for more than a decade, however, outperformed individuals who have lived in this country for

fewer years.

58 Profiles of Adult Literacy in Iowa

Because 96 percent of the Iowa population is White, the numbers of Iowa adults in other racial/ethnic groups were too small to provide reliable proficiency estimates. In the region and nation, however, White adults were less likely than African American or Latino adults to demonstrate limited English literacy skills and more likely to demonstrate advanced skills. Native-born Latino adults had higher average literacy scores than African American adults.

Approximately three-quarters of the Iowa survey respondents reported that they had lived in the state for more than 20 years. Fourteen percent had lived there for 11 to 20 years, and the remainder had lived there for fewer years. On average, there are no statistically significant differences in the prose, document, or quantitative literacy scores of adults who had lived in the state for various lengths of time.

Nearly three-quarters of the Iowa respondents reported that it was unlikely they would move out of the state in the next five years, while 18 percent said it was somewhat likely and 10 percent said it was very likely. Again, there were no significant differences between the literacy proficiencies of Iowa adults who planned to leave the state in the next five years and those who expected to remain.

Iowa residents who said they had a limiting physical or mental condition were far more likely than individuals without such a condition to perform in the lowest literacy level on each scale and far less likely to reach the highest levels.

Finally, the average prose, document, and quantitative scores of men and women in Iowa are comparable.

SECTION II





and a subscription of the second second

SECTION II



Education and Training

In the past few decades, the American educational system has been the subject of unprecedented scrutiny. Spurred by numerous studies decrying the quality of primary and secondary education in this country, many business leaders, policymakers, and others have become alarmed about the capacity of American schools to prepare individuals to lead productive, rewarding lives, as well as to promote social well-being and ensure our nation's economic competitiveness.

Given these concerns, and given the close ties between education and literacy, the committees that guided the State and National Adult Literacy Surveys determined that respondents should be asked an extensive series of questions about their educational attainments in the formal school system, as well as about their participation in adult education and training. These areas are addressed in this section of the report, and the relationship between education and literacy is probed.

Educational Background

The level of education attained is strongly associated with literacy skills.¹ The following pages present survey data on the educational attainments of adults in Iowa and nationwide, as well as on the attainments of respondents who belong to various racial/ethnic, age, and other groups. In addition, data are presented on respondents' educational goals and their participation in high school equivalency programs.

¹ In this section, "level of education" refers to the highest level of education that respondents reported having completed at the time of the survey.

Section II 63

Highest level of education attained in the United States

The educational attainments of adults in Iowa were nearly identical to those of adults nationwide (Table 2.1P,D,Q). Five percent of the state's residents were still in high school at the time of the survey. Seven percent had completed less than nine years of schooling, and another 11 percent had completed some high school without receiving a diploma. A high school diploma was the highest level of education attained by approximately one-third (34 percent) of the state's population, and a GED or high school equivalency was the highest level reached by another 4 percent.

Forty-one percent of the adults living in Iowa had continued their education beyond high school or the GED. Twenty-one percent of the state's residents had completed some postsecondary education without receiving a degree, while 4 percent had earned a two-year degree and 8 percent had earned a degree from a four-year institution. Another 8 percent of the state's residents had continued their education beyond a four-year college degree.

As expected, adults who had completed higher levels of schooling tended to outperform those with more limited education. They were much less likely to perform in the lowest literacy levels on each scale and much more likely to attain the highest levels.

In fact, average literacy proficiencies rise steadily across the entire range of education levels. The number of Iowa residents who did not go beyond eighth grade was too small to permit reliable performance estimates. The average prose proficiency of those who had completed between nine and 12 years of schooling was 242, however, compared with 283 for those who earned a high school diploma but went no further. Individuals with some postsecondary education but no degree had an average prose score of 302, compared with 316 for those with a two-year degree, 333 for individuals whose highest level of education was a four-year degree, and 342 for those who had completed some postgraduate studies beyond the four-year degree. Similar patterns are found on the document and quantitative scales, where average literacy proficiencies also rise with each successive level of educational attainment.

Stated differently, the difference in average prose proficiency between adults who had nine to 12 years of education and those who had finished at least some graduate work is approximately 100 points. This translates to a gap of about two proficiency levels — a very large difference in the difficulty and complexity of literacy skills and strategies. This might mean the difference, for example, between being able to identify a piece of information in a short news article and being able to extract information from more lengthy or complex prose materials. While the average scores of adults with less than a high school

64 Education and Training



Prose Literacy Levels and Average Proficiencies, by Level of Education in the United States: Results for Iowa, the Midwest, and the Nation

LEVEL OF EDUCATION ATTAINED IN THE UNITED					Percentag	ge of adults in	each prose lite	eracy level	
STATES				Level 1 225 or lower	Level 2 226 to 275	Level 3 276 to 325	Level 4 326 to 375	Level 5 376 or higher	Average Proficiency
	n	WGT N (/1000)	PCT	RPCT (SE)	RPCT (SE)	RPCT (SE)	RPCT (SE)	RPCT (SE)	PROF (SE)
Lowa Still in high school 0 to 8 years 9 to 12 years High school GED Some postsecondary Two year degree Four year degree	54 36 89 355 46 331 65 141	105 145 229 708 84 432 77 157	5 7 11 34 4 21 4 8 8	7 (4.4) **** (*****) 36 (14.1) 10 (3.0) 12 (8.6) 2 (1.0) $0^{\dagger}(1.2)$ $0^{\dagger}(1.0)$ $0^{\dagger}(0.0)$	28 (13.0) **** (*****) 43 (9.0) 30 (3.0) 42 (15.2) 20 (11.9) 10 (6.4) 5 (1.6) 4 (1.8)	46 (13.3) **** (*****) 18 (9.3) 43 (5.5) 38 (13.5) 49 (9.2) 54 (8.9) 34 (5.2) 28 (6.1)	18 (5.1) **** (*****) 3 (3.3) 15 (2.7) 8 (5.4) 25 (3.8) 30 (7.2) 53 (6.5) 51 (7.1)	$ \begin{array}{c} 1 (2.3) \\ **** (****) \\ 0^{\dagger} (0.4) \\ 1 (0.6) \\ 0^{\dagger} (0.6) \\ 2 (1.2) \\ 5 (3.4) \\ 8 (3.7) \\ 18 (4.3) \end{array} $	288 (5.1) **** (*****). 242 (8.2) 283 (3.1) 269 (7.6) 302 (5.7) 316 (5.5) 333 (3.3)! 342 (4.7)
Graduate studies/degree Midwest Still in high school 0 to 8 years 9 to 12 years High school GED Some postsecondary Two year degree Four year degree Graduate studies/degree	301 412 865 1,992 314 1,983 300 704 618	2,343 3,558 5,820 13,306 1,594 10,149 1,395 3,816 3,333	5 8 13 29 4 22 3 8 7	11 (2.7) 64 (5.2) 39 (4.1) 13 (1.5) 12 (3.8) 6 (1.2) 3 (2.1) 2 (1.0) 1 (1.3)	32 (4.6) 31 (4.5) 40 (3.3) 36 (1.9) 37 (6.4) 23 (1.9) 17 (4.0) 9 (2.0) 4 (1.4)	43 (4.8) 5 (2.0) 19 (2.2) 40 (2.9) 44 (5.8) 46 (2.5) 42 (5.3) 33 (3.8) 28 (4.2)	$\begin{array}{c} 13 (5.2) \\ 0^{\dagger} (0.2) \\ 3 (1.1) \\ 10 (1.5) \\ 7 (2.7) \\ 22 (1.7) \\ 33 (5.7) \\ 44 (3.7) \\ 52 (3.5) \end{array}$	$\begin{array}{c} 1 (1.1) \\ 0^{\dagger} (0.0) \\ 0^{\dagger} (0.1) \\ 1 (0.2) \\ 0^{\dagger} (0.0) \\ 2 (0.6) \\ 5 (3.7) \\ 11 (2.6) \\ 15 (2.3) \end{array}$	282 (3.9) 199 (5.8) 235 (3.3) 274 (1.3) 271 (4.6) 297 (1.8) 310 (5.0) 328 (3.1) 340 (3.3)
Nation Still in high school 0 to 8 years 9 to 12 years High school GED Some postsecondary Two year degree Four year degree Graduate studies/degree	973 2,167 3,311 6,107 1,062 6,587 1,033 2,534 2,253	8,268 18,356 24,982 51,290 7,224 39,634 6,831 17,804 16,306	4 10 13 27 4 21 4 9 9	16 (1.8) 75 (1.7) 42 (1.4) 16 (0.8) 14 (1.6) 8 (0.5) 4 (1.1) 4 (0.7) 2 (0.4)	36 (2.2) 20 (1.4) 38 (1.1) 36 (1.3) 39 (2.5) 23 (0.8) 19 (2.3) 11 (1.2) 7 (1.0)	37 (2.6) 4 (0.9) 17 (1.0) 37 (1.7) 39 (2.8) 45 (0.9) 41 (2.9) 35 (2.0) 28 (1.4)	$\begin{array}{c} 11 (1.9) \\ 0^{\dagger} (0.3) \\ 2 (0.4) \\ 10 (0.9) \\ 7 (1.2) \\ 22 (0.8) \\ 32 (2.5) \\ 40 (1.5) \\ 47 (1.8) \end{array}$	$\begin{array}{c} 0^{\dagger}(\ 0.5) \\ 0^{\dagger}(\ 0.0) \\ 0^{\dagger}(\ 0.1) \\ 1 \ (\ 0.2) \\ 0^{\dagger}(\ 0.6) \\ 3 \ (\ 0.3) \\ 4 \ (\ 0.9) \\ 10 \ (\ 1.3) \\ 16 \ (\ 1.1) \end{array}$	271 (2.0) 177 (2.6) 231 (1.5) 270 (1.1) 268 (1.8) 294 (1.0) 308 (2.4) 322 (1.6) 336 (1.4)

n = sample size; WGT N = population size estimate / 1,000 (the sample sizes for subpopulations may not add up to the total sample sizes, due to missing data); PCT = percentage in group; RPCT = row percentage estimate; PROF = average proficiency estimate; (SE) = standard error of the estimate (the reported sample estimate can be said to be within 2 standard errors of the true population value with 95% confidence).

Percentages less than 0.5 are rounded to zero. +

Sample size is insufficient to permit a reliable estimate (fewer than 45 respondents). Interpret with caution -- the nature of the sample does not allow accurate determination of the variability of this statistic. 1

Source: Educational Testing Service, State Adult Literacy Survey, 1992.

Section II 65



Document Literacy Levels and Average Proficiencies, by Level of Education in the United States: Results for Iowa, the Midwest, and the Nation

LEVEL OF EDUCATION ATTAINED IN THE UNITED					Percentage	of adults in e	ach document	literacy level	
STATES				Level 1 225 or lower	Level 2 226 to 275	Level 3 276 to 325	Level 4 326 to 375	Level 5 376 or higher	Average Proficiency
	n	WGT N (/1000)	PCT	RPCT (SE)	RPCT (SE)	RPCT (SE)	RPCT (SE)	RPCT (SE)	PROF (SE)
lowa									
Still in high school	54	105	5	5 (2.7)	30 (8.2)	40 (9.6)	23 (5.9)	2 (2.0)	293 (5.5)
0 to 8 years	36	145	7	*** (****)	*** (****)	*** (****)	*** (****)	*** (****)	*** (****)
9 to 12 years	89	229	11	41 (11.0)	43 (6.2)	13 (6.6)	3 (2.7)	0 [†] (0.0)	235 (12.1)
High school	355	708	34	12 (2.6)	34 (3.2)	40 (3.3)	14 (2.6)	1 (0.6)	279 (2.4)
GED	46	84	4	13 (4.7)	45 (10.7)	37 (12.8)	6 (5.0)	0 [†] (0.0)	267 (8.7)
Some postsecondary	331	432	21	5 (3.4)	23 (4.1)	46 (5.4)	23 (4.6)	3 (1.5)	298 (6.8)
Two year degree	65	77	4	1 (2.9)	17 (6.3)	52 (7.8)	27 (6.9)	3 (2.8)	309 (6.9)
Four year degree	141	157	8	0 [†] (0.5)	8 (2.6)	37 (7.0)	47 (7.1)	7 (4.0)	327 (3.6)1
Graduate studies/degree	127	157	8	1 (1.2)	7 (6.6)	38 (7.8)	42 (4.9)	12 (6.4)	330 (6.0)
Midwest									
Still in high school	301	2,343	5	8 (2.9)	31 (4.0)	44 (4.8)	16 (2.7)	1 (0.9)	286 (3.8)
0 to 8 years	412	3,558	8	70 (6.4)	28 (6.0)	2 (1.2)	0 [†] (0.0)	0†(0.0)	191 (6.1)
9 to 12 years	865	5,820	13	43 (3.3)	39 (2.8)	15 (1.9)	3 (1.1)	0 [†] (0.5)	230 (3.2)
High school	1,992	13,306	29	16 (1.1)	39 (2.3)	35 (2.2)	9 (0.7)	0 [†] (0.1)	269 (1.5)
GED	314	1,594	4	16 (4.0)	40 (6.3)	37 (5.8)	7 (3.9)	0 [†] (0.1)	267 (5.0)
Some postsecondary	1,983	10,149	22	7 (1.2)	26 (2.0)	44 (2.3)	21 (2.0)	2 (0.8)	292 (2.4)
Two year degree	300	1,395	3	5 (3.0)	20 (4.7)	43 (4.8)	27 (4.1)	5 (2.3)	303 (4.7)
Four year degree	704	3,816	8	2 (0.9)	12 (1.9)	37 (3.6)	39 (3.0)	9 (1.7)	321 (2.5)
Graduate studies/degree	618	3,333	7	3 (1.1)	8 (1.9)	33 (3.3)	45 (3.8)	11 (2.7)	329 (2.7)
Nation	1.0								
Still in high school	973	8,268	4	15 (1.5)	35 (2.3)	38 (2.6)	12 (1.5)	1 (0.6)	274 (1.9)
0 to 8 years	2,167	18,356	10	79 (1.7)	18 (1.6)	3 (0.8)	01(0.1)	0 [†] (0.0)	170 (2.4)
9 to 12 years	3,311	24,982	13	46 (1.7)	37 (1.6)	15 (1.3)	2 (0.4)	01(0.1)	227 (1.6)
High school	6,107	51,290	27	20 (0.8)	38 (1.0)	33 (1.1)	9 (0.6)	1 (0.2)	264 (1.1)
GED	1,062	7,224	4	17 (2.0)	42 (2.7)	34 (2.3)	7 (1.1)	0 [†] (0.5)	264 (2.2)
Some postsecondary	6,587	39,634	21	9 (0.4)	27 (0.8)	42 (1.0)	20 (0.8)	2 (0.4)	290 (0.9)
Two year degree	1,033	6,831	4	6 (1.4)	23 (2.0)	43 (2.6)	25 (2.7)	3 (0.9)	299 (2.6)
Four year degree	2,534	17,804	9	4 (0.5)	15 (1.3)	37 (1.5)	36 (1.2)	8 (1.2)	314 (1.4)
Graduate studies/degree	2,253	16,306	9	3 (0.6)	10 (0.9)	34 (1.8)	41 (1.9)	12 (1.1)	326 (1.8)

n = sample size; WGT N = population size estimate / 1,000 (the sample sizes for subpopulations may not add up to the total sample sizes, due to missing data); PCT = percentage in group; RPCT = row percentage estimate; PROF = average proficiency estimate; (SE) = standard error of the estimate (the reported sample estimate can be said to be within 2 standard errors of the true population value with 95% confidence).

† Percentages less than 0.5 are rounded to zero.

*** Sample size is insufficient to permit a reliable estimate (fewer than 45 respondents).

Interpret with caution -- the nature of the sample does not allow accurate determination of the variability of this statistic. Source: Educational Testing Service, State Adult Literacy Survey, 1992.

66 Education and Training

Quantitative Literacy Levels and Average Proficiencies, by Level of Education in the United States: Results for Iowa, the Midwest, and the Nation

LEVEL OF EDUCATION ATTAINED IN THE UNITED					Percentage	of adults in ea	ch quantitative	literacy level	
STATES				Level 1 225 or lower	Level 2 226 to 275	Level 3 276 to 325	Level 4 326 to 375	Level 5 376 or higher	Average Proficiency
	n	WGT N (/1000)	PCT	RPCT (SE)	RPCT (SE)	RPCT (SE)	RPCT (SE)	RPCT (SE)	PROF (SE)
lowa				1					-
Still in high school	54	105	5	6 (4.7)	34 (5.6)	42 (7.9)	17 (6.9)	1 (1.7)	286 (6.4)
0 to 8 years	36	145	7	*** (****)	*** (****)	*** (****)	*** (****)	*** (****)	*** (****)
9 to 12 years	89	229	11	37 (11.8)	43 (5.5)	17 (8.2)	3 (2.7)	0 [†] (0.5)	240 (9.6)
High school	355	708	34	8 (1.7)	27 (5.4)	43 (4.5)	19 (2.6)	2 (1.2)	289 (3.1)
GED	46	84	4	17 (4.9)	34 (9.1)	35 (11.5)	13 (6.6)	1 (2.4)	271 (8.6)
Some postsecondary	331	432	21	6 (5.7)	16 (2.0)	45 (4.5)	28 (3.9)	4 (2.1)	305 (9.4)
Two year degree	65	77	4	0 [†] (0.0)	13 (6.8)	51 (8.5)	28 (8.5)	9 (3.9)	317 (6.9)
Four year degree	141	157	8	0 [†] (0.6)	7 (3.2)	34 (4.8)	48 (8.1)	11 (4.8)	332 (3.8)1
Graduate studies/degree	127	157	8	0 ^T (0.6)	3 (2.4)	22 (4.7)	56 (6.1)	18 (6.1)	345 (4.2)
Midwest							1 minute		
Still in high school	301	2,343	5	11 (3.4)	34 (5.9)	39 (4.9)	17 (3.7)	1 (0.5)	282 (4.5)
0 to 8 years	412	3,558	8	64 (6.0)	29 (4.8)	7 (2.9)	0 [†] (0.5)	01(0.0)	194 (7.2)
9 to 12 years	865	5,820	13	41 (3.8)	36 (4.8)	20 (3.4)	3 (1.0)	0'(0.5)	232 (3.9)
High school	1,992	13,306	29	14 (1.4)	32 (2.2)	40 (2.2)	13 (1.4)	1 (0.3)	277 (1.5)
GED	314	1,594	4	14 (4.0)	36 (6.8)	40 (7.7)	10 (3.3)	0'(0.2)	272 (5.5)
Some postsecondary	1,983	10,149	22	7 (1.2)	22 (2.4)	41 (2.1)	26 (2.4)	4 (0.8)	300 (2.7)
Two year degree	300	1,395	3	3 (1.6)	19 (3.8)	41 (4.8)	30 (4.7)	10 (2.7)	310 (4.9)
Four year degree	704	3,816	8	2(1.1)	10 (2.1)	33 (2.7)	43 (3.8)	13 (2.0)	329 (2.3)
Graduate studies/degree	618	3,333	1	2 (1.2)	7 (2.1)	28 (3.5)	40 (4.0)	17 (2.0)	330 (2.0)
Nation					N. Constant	1 and a second	1 march	1 and inter	
Still in high school	973	8,268	4	19 (1.7)	35 (3.0)	32 (2.3)	12 (2.0)	1 (0.9)	269 (2.2)
0 to 8 years	2,167	18,356	10	76 (2.0)	18 (1.8)	5 (1.1)	1 (0.3)	0 [†] (0.2)	169 (3.1)
9 to 12 years	3,311	24,982	13	45 (1.6)	34 (1.6)	17 (1.3)	3 (0.6)	0 [†] (0.1)	227 (1.7)
High school	6,107	51,290	27	18 (0.8)	33 (1.1)	37 (1.1)	12 (0.5)	1 (0.2)	270 (1.1)
GED	1,062	7,224	4	16 (2.0)	38 (2.5)	35 (2.5)	10 (1.4)	1 (0.5)	268 (2.7)
Some postsecondary	6,587	39,634	21	8 (0.6)	23 (1.2)	42 (1.4)	23 (1.3)	4 (0.4)	295 (1.4)
Two year degree	1,033	6,831	4	4 (0.8)	19 (2.0)	43 (2.0)	29 (2.7)	5 (1.3)	307 (2.8)
Four year degree	2,534	17,804	9	4 (0.5)	12 (1.0)	35 (1.4)	38 (1.4)	12 (1.1)	322 (1.2)
Graduate studies/degree	2,253	16,306	9	2 (0.5)	9 (0.8)	30 (1.4)	42 (1.7)	17 (1.4)	334 (1.3)

n = sample size; WGT N = population size estimate / 1,000 (the sample sizes for subpopulations may not add up to the total sample sizes, due to missing data); PCT = percentage in group; RPCT = row percentage estimate; PROF = average proficiency estimate; (SE) = standard error of the estimate (the reported sample estimate can be said to be within 2 standard errors of the true population value with 95% confidence).

† Percentages less than 0.5 are rounded to zero.

*** Sample size is insufficient to permit a reliable estimate (fewer than 45 respondents).

Interpret with caution -- the nature of the sample does not allow accurate determination of the variability of this statistic.

Source: Educational Testing Service, State Adult Literacy Survey, 1992.

Section II 67

education are in the Level 2 range, the average scores of those who received a diploma are in the Level 3 range, and of those who pursued postsecondary studies, in the range for Level 3 or 4.

Survey respondents in Iowa and in the Midwest who reported they were still in high school at the time of the survey demonstrated higher average prose and document proficiencies than their counterparts nationwide. For example, Iowa high school students had an average document score of 293, and students in the Midwest had an average score of 286, compared with only 274 for students nationwide.

Some high school graduates in the state, region, and nation did poorly in the assessment. On each literacy scale, 8 to 12 percent of the high school graduates in Iowa performed in the Level 1 range, and another 27 to 34 percent performed in Level 2, while 15 to 21 percent reached the two highest levels on each scale. *High school graduates and GED recipients performed comparably in the assessment. Although there appear to be differences between these two groups in the Iowa population, the differences are not statistically significant.*

Adults who had completed a two-year college degree performed significantly better than those whose highest level of education was a high school diploma. Two-year college graduates had an average prose score of 316, a document score of 309, and a quantitative score of 317 — all in the Level 3 range. Four-year college graduates had still higher scores, achieving an average prose score of 333, an average document score of 327, and an average quantitative score of 332 — all in the low end of the Level 4 range.

These results make it clear that education and literacy skills are interconnected. One can infer that education strengthens an individual's ability to read and use various types of materials. It is also true, however, that those with higher proficiencies are more likely to extend their schooling.

Average years of schooling completed by various population groups

A question that arises from these data is whether the differences in literacy proficiency among certain groups in the population (as seen in Section I) can be explained, at least in part, by differences in educational attainment. In other words, do the groups that demonstrate lower proficiencies also report having had fewer years of schooling? To address such questions, it was necessary to calculate the average years of schooling completed by survey respondents, based on the highest level of education they reported having achieved in this country. This new variable offers a way to compare the educational attainments of adults in groups defined by sex, age, race/ethnicity, and other characteristics of interest.

68.... Education and Training

Adults in Iowa, like those in the Midwest and the nation as a whole, completed an average of approximately 12.5 years of schooling, or slightly more than a high school diploma (Table 2.2). Males had completed slightly more years of schooling than females (12.8 years, compared with 12.3 years). As seen in Section I, however, there were no significant differences in performance between these two groups in the Iowa population.

The differences in years of schooling among the various age groups are noteworthy. Though not all the differences are statistically significant, average years of schooling tend to increase from the youngest age group to the middle age groups and then to decline across the older groups. The most striking difference is seen between the oldest age group (age 65 and older) and the other groups. In the national sample, adults age 65 and older had completed about 11 years of schooling, on average, or less than a high school diploma. In contrast, all the other age groups had completed at least 12 years of schooling, on average. As seen in Section I, the prose, document, and quantitative proficiencies of older adults were also substantially lower than those of adults in the other age groups.

Given the strong connection between adults' level of education and their literacy skills, another question of interest concerns the intergenerational nature of education. The survey data show that adults' academic attainments are, in fact, related to those of their parents. Iowa respondents whose parents had a four-year degree had completed an average of almost 15 years of schooling, compared with only 13 years for respondents whose parents had ended their education upon receiving a high school diploma. Iowa respondents with parents who had not finished high school reported an average of 11 years of schooling — less than a high school diploma. The regional and national results are very similar.

Thus, the more education adults' parents had completed, the more education they themselves were likely to have completed — and the higher their literacy proficiencies were likely to be. Still, respondents' own levels of education are better predictors of their literacy skills than their parents' levels of education.²

Level of education attained before coming to the United States

Because there are so many immigrants in this country, and because many of these individuals were educated in their native countries, foreign-born survey respondents were asked to indicate the highest level of education they had completed before coming to the United States.

Section II 69

²I.S. Kirsch, A. Jungeblut, L. Jenkins, and A. Kolstad. (1993). Adult Literacy in America: A First Look at the Results of the National Adult Literacy Survey. Washington, D.C.: Government Printing Office. pp. 28-9.



Average Years of Schooling Completed in the United States by Various Population Groups: Results for Iowa, the Midwest, and the Nation

		Average yea	rs of schoolin	g completed l	by adults in	
	lov	va	Mid	west	Na	tion
		(SE)		(SE)		(SE)
Total	12.5	(0.1)	12.5	(0.1)	12.4	(0.0)
Sex Male Female	12.8 12.3	(0.4) (0.3)	12.7 12.4	(0.1) (0.1)	12.5 12.3	(0.0) (0.0)
Age 16 to 18 19 to 24 25 to 39 40 to 54 55 to 64 65 and older	12.8 13.1 13.3 13.2 12.3 10.3	(0.1) (0.4) (0.4) (0.6) (0.4) (0.5)	12.5 13.1 13.2 13.0 12.2 11.0	(0.1) (0.1) (0.1) (0.1) (0.2) (0.1)	12.3 12.9 13.1 13.0 11.8 10.7	(0.0) (0.0) (0.1) (0.1) (0.1) (0.1)
Race/Ethnicity White African American Latino (all) Asian/Pacific Islander Other	12.5 *** ***	(0.1) (****) (****) (****) (****)	12.7 11.8 10.1 13.2 11.7	(0.1) (0.1) (0.3) (0.8) (0.7)	12.8 11.6 10.2 13.0 11.3	(0.0) (0.1) (0.1) (0.3) (0.3)
Country of Birth United States or U.S. territory Other country	12.5	(0.1) (****)	12.6 8.2	(0.0) (0.6)	12.6 8.7	(0.0) (0.2)
Parents' Highest Level of Education 0 to 12 years High school GED Some postsecondary Four year deg. or more	11.2 12.9 12.4 13.6 14.6	(0.2) (0.3) (0.3) (0.3) (0.3) (0.3)	11.4 12.8 12.9 13.6 14.5	(0.1) (0.1) (0.1) (0.1) (0.1)	11.1 12.9 12.7 13.6 14.6	(0.0) (0.0) (0.1) (0.1) (0.0)

- (SE) = standard error of the estimate (the reported sample estimate can be said to be within 2 standard errors of the true population value with 95% confidence).
- *** Sample size is insufficient to permit a reliable estimate (fewer than 45 respondents).
- Source: Educational Testing Service, State Adult Literacy Survey, 1992.

70 Education and Training

÷

The numbers of Iowa residents born outside this country were too small to support such analyses. Nationwide, however, 8 percent of the foreign-born adults reported that their highest level of education before coming to this country was primary school; 26 percent said it was elementary school; 31 percent said it was secondary school; 3 percent said it was vocational school;³ and 14 percent said it was college or university. Fourteen percent of the foreign-born residents of this country said they did not complete any schooling before coming to the United States (Table 2.3).

Foreign-born adults who had completed a college or university education abroad tended to demonstrate higher proficiencies in English than individuals who had completed lower levels of education before coming to this country.

It is interesting to compare the levels of education attained abroad by foreign-born adults who have lived in this country for varying lengths of time. The numbers of foreign-born Iowa residents are again too small to support such analyses. Nationwide, however, about 80 percent of the foreign-born adults who have lived in this country for more than a decade reported having completed some schooling before coming (Table 2.4). One-quarter had finished secondary school abroad, and 10 percent had completed a college or university education before moving to the United States.

Foreign-born adults who have lived in this country for a decade or less that is, from six to 10 years, or one to five years — were more likely than longer-term residents to have completed some education before coming. Nationwide, virtually all (95 percent) of the foreign-born adults who have lived in this country for between six and 10 years said they had completed some schooling before coming. Forty percent had attended secondary school, and 15 percent had attended a college or university. Among foreign-born adults who have lived in the United States for five years or less, 96 percent had attended school before coming. Thirty-nine percent had completed secondary school, and one-quarter had attained a college or university education before coming.

Section II 71

³ In this report, the term "vocational" refers to vocational, technical, or business programs at the postsecondary level.



Average Literacy Proficiencies of Foreign-born Adults, by Highest Level of Education Attained Before Coming to the U.S.

LEVEL OF EDUCATION				Average profic	ciency of adults on eac	ch literacy scale
OMING TO THE U.S.				Prose	Document	Quantitative
	n	WGT N (/1000)	PCT	PROF (SE)	PROF (SE)	PROF (SE)
lowa						
Name	0	÷.	10	*** / ****	*** / ****	*** / ****
Drimonu	3	5	12	*** (****)	*** (****)	*** (****)
Flomontany	2	5	10	*** (****)	*** (****)	*** (****)
Elementary	0	10	19	*** (****)	())	*** (****)
Vecetional	2	10	20	*** (****)	*** (****)	*** (****)
Collogo/university	0	Ö	24	*** (****)	*** / ****)	*** (****)
Other	2	2	5	*** (****)	*** (****)	*** (****)
Midwest					• •	
None	43	250	17	*** (****)	*** / ****	*** (****)
Primary	25	98	6	*** (****)	*** (****)	*** (****)
Elementary	87	414	27	175 (8 6)	177 (9 3)	179 (11 9)
Secondary	83	491	33	230 (14 1)	238 (17.4)	241 (14 2)
Vocational	10	43	3	*** (****)	*** (****)	*** (****)
College/university	44	187	12	*** (****)	*** (****)	*** (****)
Other	6	26	2	*** (****)	*** (****)	*** (****)
Nation						
None	344	2,660	14	253 (5.4)	245 (5.1)	244 (6.3)
Primary	254	1,563	8	182 (8.3)	179 (8.0)	174 (8.6)
Elementary	712	4,836	26	169 (4.7)	169 (5.0)	168 (6.2)
Secondary	771	5,713	31	209 (4.1)	210 (3.8)	216 (3.8)
Vocational	93	613	3	225 (8.9)	226 (9.2)	232 (9.8)
College/university	387	2,680	14	257 (4.9)	259 (5.2)	270 (4.8)
Other	54	421	2	267 (13.1)	267 (11.3)	280 (13.8)

n = sample size; WGT N = population size estimate / 1,000 (the sample sizes for subpopulations may not add up to the total sample sizes, due to missing data); PCT = percentage in group; PROF = average proficiency estimate; (SE) = standard error of the estimate (the reported sample estimate can be said to be within 2 standard errors of the true population value with 95% confidence).

Sample size is insufficient to permit a reliable estimate (fewer than 45 respondents). Interpret with caution -- the nature of the sample does not allow accurate determination of the variability of this statistic. *** 1 Source: Educational Testing Service, State Adult Literacy Survey, 1992.

72 Education and Training

Highest Level of Education Attained Before Coming to the U.S., by Years Lived in the U.S.: Results for Iowa, the Midwest, and the Nation

YEARS LIVED IN THE U. S.					Percentage of	adults who at	ained each lev	el of education	n
				None	Primary & Elementary	Secondary	Vocational	College/ university	Other
	n	WGT N (/1000)	PCT	RPCT (SE)	RPCT (SE)	RPCT (SE)	RPCT (SE)	RPCT (SE)	RPCT (SE)
Iowa									
1 to 5 years	8	12	32	*** (****)	*** (****)	*** (****)	*** (****)	*** (****)	*** (****)
6 to 10 years	4	5	13	*** (****)	*** (****)	*** (****)	*** (****)	*** (****)	*** (****)
More than 10 years	15	21	55	*** (****)	*** (****)	*** (****)	*** (****)	*** (****)	*** (****)
Midwest									
1 to 5 years	62	278	18	1 (0.5)	26 (5.5)	52 (7.2)	7 (4.0)	14 (4.4)	0 [†] (0.1)
6 to 10 years	56	294	19	17 (14.7)	33 (9.7)	30 (9.9)	0 [†] (0.0)	13 (6.0)	6 (5.2)
More than 10 years	180	938	62	21 (4.4)	36 (8.5)	27 (3.7)	3 (1.6)	12 (5.7)	1 (0.5)
Nation									
1 to 5 years	568	3,998	22	4 (1.1)	25 (2.2)	39 (3.0)	3 (0.7)	25 (2.6)	3 (1.1)
6 to 10 years	481	3,181	17	5 (1.5)	35 (2.6)	40 (3.0)	4 (1.0)	15 (2.0)	3 (1.1)
More than 10 years	1,556	11,207	61	21 (1.5)	38 (1.9)	25 (1.6)	3 (0.6)	10 (1.0)	2 (0.3)

n = sample size; WGT N = population size estimate / 1,000 (the sample sizes for subpopulations may not add up to the total sample sizes, due to missing data); PCT = percentage in group; RPCT = row percentage estimate; (SE) = standard error of the estimate (the reported sample estimate can be said to be within 2 standard errors of the true population value with 95% confidence).

Percentages less than 0.5 are rounded to zero.

- Sample size is insufficient to permit a reliable estimate (fewer than 45 respondents). ***
- Interpret with caution -- the nature of the sample does not allow accurate determination of the variability of this statistic.

Source: Educational Testing Service, State Adult Literacy Survey, 1992.

Participation in a GED or high school equivalency program

Of the nearly half-million adults in Iowa who had not completed high school that is, residents of all ages who had not earned a diploma - 32 percent said they had participated in a GED or high school equivalency program. Slightly more than half (57 percent) of these program participants reported they had actually received their GED or equivalency diploma (Table 2.5).

Section II 73



Average Literacy Proficiencies of Dropouts, by Participation in a GED Program: Results for Iowa, the Midwest, and the Nation

STUDIED FOR, RECEIVED A GED OR HIGH SCHOOL			[Average proficiency of adults on each literacy scale									
EGOIVALENCI			[Prose	Document	Quantitative							
	n	WGT N (/1000)	PCT	PROF (SE)	PROF (SE)	PROF (SE)							
Ever studied for a GED?													
Yes	83	148	32	262 (6.2)	260 (6.8)	264 (6 7)]							
No	87	308	68	217 (9.1)	205 (9.4)	212 (10.9)							
Midwest													
Yes	632	3,124	29	260 (3.5)	256 (4.5)	257 (5.1)							
No	945	7,761	71	216 (3.4)	210 (3.4)	213 (4.2)							
Yes	2 3 3 5	14 777	30	254 (1 2)	251 (1 7)	252 (1 9)							
No	4,118	35,062	70	201 (1.9)	195 (1.9)	196 (2.1)							
If yes did you receive it?													
lowa						1							
Yes	46	84	57	269 (7.6)	267 (8.7)	271 (8.6)							
No	37	64	43	*** (****)	*** (****)	*** (****)							
Midwest	1.11												
Yes	314	1,594	51	271 (4.6)	267 (5.0)	272 (5.5)							
No	318	1,530	49	249 (4.3)	245 (5.6)	242 (7.2)							
Yes	1.062	7 994	10	268 (1 8)	264 (2 2)	268 (27)							
No	1,273	7,552	51	241 (2 1)	239 (2.4)	236 (2.6)							

n = sample size; WGT N = population size estimate / 1,000 (the sample sizes for subpopulations may not add up to the total sample sizes, due to missing data); PCT = percentage in group; PROF = average proficiency estimate; (SE) = standard error of the estimate (the reported sample estimate can be said to be within 2 standard errors of the true population value with 95% confidence).

*** Sample size is insufficient to permit a reliable estimate (fewer than 45 respondents).

I Interpret with caution -- the nature of the sample does not allow accurate determination of the variability of this statistic.

Source: Educational Testing Service, State Adult Literacy Survey, 1992.

74 Education and Training

Nationwide, 30 percent of the roughly 50 million school dropouts reported having studied for a GED or high school equivalency, and half (49 percent) of these said they had received it.

In the national population, as in the regional and state populations, school dropouts who had not participated in a GED or high school equivalency program demonstrated average prose, document, and quantitative proficiencies in the Level 1 range (201, 195, and 196, respectively). Program participants' scores were at least 50 points higher, lying at the middle of the Level 2 range. The average scores of adults who had received a GED or high school equivalency were significantly (25 to 32 points) higher than the scores of those who had participated in the program but had not completed it.

A vast majority of the GED program participants in Iowa (82 percent) were age 25 or older. Nearly half (44 percent) were between the ages of 25 and 39, another 22 percent were in the 40 to 54 age group, and 16 percent were 55 or older. Eighteen percent were below age 25 (Table 2.6).



IOWA TABLE 2.6

Among School Dropouts, Participation in a GED Program, by Age: Results for Iowa

STUDIED FOR, RECEIVED A GED OR Percentage of adults in each age group

HIGH SCHOOL EQUIVALENCY				16 to 18	19 to 24	25 to 39	40 to 54	55 to 64	65 and older
	n	WGT N (/1000)	PCT	RPCT (SE)	RPCT (SE)				
Ever studied for a GED? Yes No	83 87	148 308	32 68	2 (0.7) 2 (0.8)	16 (6.2) 1 (0.8)	44 (7.2) 11 (7.5)	22 (8.1) 12 (12.5)	14 (3.1) 17 (14.4)	2 (1.6) 58 (33.4)
If yes, did you receive it? Yes No	46 37	84 64	57 43	3 (1.3) *** (****)	14 (6.9) *** (****)	50 (9.9) *** (****)	16 (8.4) *** (****)	16 (4.2) *** (****)	0 [†] (0.0) *** (****)

n = sample size; WGT N = population size estimate / 1,000 (the sample sizes for subpopulations may not add up to the total sample sizes, due to missing data); PCT = percentage in group; RPCT = row percentage estimate; (SE) = standard error of the estimate (the reported sample estimate can be said to be within 2 standard errors of the true population value with 95% confidence).

Percentages less than 0.5 are rounded to zero.

*** Sample size is insufficient to permit a reliable estimate (fewer than 45 respondents).

Interpret with caution -- the nature of the sample does not allow accurate determination of the variability of this statistic.

Source: Educational Testing Service, State Adult Literacy Survey, 1992.

Section II 75

The survey results do not provide insight into adults' reasons or motivations for participating in programs such as the GED. Still, one plausible interpretation is that after a few years in the labor force, young school dropouts discover the importance of a high school diploma or an equivalent credential in obtaining a job and advancing in the workplace.

Current educational enrollment

Survey respondents were asked to indicate whether or not they were enrolled in school or college, either full or part time. Eleven percent of the adults in Iowa and an equivalent percentage nationwide responded that they were currently enrolled, compared with 9 percent of the adults in the region (Table 2.7).

These individuals demonstrated significantly higher average prose, document, and quantitative proficiencies than respondents who were not enrolled in school or college. On the prose scale, for example, students in Iowa had an average score of 311, 29 points higher than that of non-students (282). On the document scale, the gap was 32 points (308, compared with 276), and on the quantitative scale it was 23 points (308, compared with 285). Similar patterns are seen in the regional and national results.

When respondents who were enrolled in school or college were asked what diploma, certificate, or other credential they expected to earn, their answers varied considerably. The Iowa sample was too small to support reliable estimates. In the national sample, however, 10 percent of the adults who were enrolled in an educational program said they expected to earn a high school diploma or equivalency, and an equivalent percentage said they were pursuing a vocational, trade, or business credential. About 13 percent expected to receive an associate's degree, 38 percent were pursuing a four-year college degree, and 19 percent were working toward a master's, Ph.D., M.D., or other advanced degree. Seven percent were pursuing some other goal, and about 4 percent said they had no expectation as to what credential they would earn.

It is not surprising to find that respondents who said they were working toward an advanced degree had the highest average proficiencies on each literacy scale (326 to 332), followed by those who were pursuing a four-year degree (312 to 316). Students who reported expecting to earn a high school

diploma or equivalency demonstrated the lowest skills, on average (233 to 242).

Participation in Adult Education and Training

76 Education and Training

The National and State Adult Literacy Surveys asked respondents to provide information on their involvement in various types of adult education and

Average Literacy Proficiencies, by Current Educational Enrollment and Goals: Results for Iowa, the Midwest, and the Nation

CURRENT ENROLLMENT IN SCHOOL OR COLLEGE, AND EDUCATIONAL GOAL				Average pro	oficiency of ad literacy scale	ults on each
				Prose	Document	Quantitative
	n	WGT N (/1000)	PCT	PROF (SE)	PROF (SE)	PROF (SE)
Currently enrolled in school or college? lowa				011 (5 0)	202 (5 0)	202 (7 0)
Yes No	151 1,036	218 1,765	11 89	282 (3.5)	276 (3.4)	285 (4.3)
Yes No	794 6,097	3,845 38,797	9 91	308 (2.9) 276 (1.0)	304 (3.2) 270 (1.2)	305 (3.0) 277 (1.7)
Nation Yes No	2,850 21,009	19,316 162,012	11 89	303 (1.7) 269 (0.6)	299 (1.4) 263 (0.7)	299 (1.5) 268 (0.8)
If yes, what is your goal? Nation High School Diploma/GED Vocational/trade Two year degree Four year degree Graduate degree Other None	258 277 392 1,074 511 187 109	1,849 1,891 2,435 7,226 3,649 1,285 669	10 10 13 38 19 7 4	242 (4.7) 276 (6.2) 299 (3.9) 316 (2.5) 332 (2.7) 293 (7.1) 290 (10.5)	241 (5.0) 276 (6.2) 296 (3.3) 313 (2.2) 326 (3.1) 288 (6.0) 284 (10.3)	233 (4.1) 267 (5.0) 295 (3.7) 312 (2.3) 331 (2.6) 293 (6.0) 290 (9.6)

n = sample size; WGT N = population size estimate / 1,000 (the sample sizes for subpopulations may not add up to the total sample sizes, due to missing data); PCT = percentage in group; PROF = average proficiency estimate; (SE) = standard error of the estimate (the reported sample estimate can be said to be within 2 standard errors of the true population value with 95% confidence).

Percentages less than 0.5 are rounded to zero. t

Sample size is insufficient to permit a reliable estimate (fewer than 45 respondents). ***

Interpret with caution -- the nature of the sample does not allow accurate determination of the variability of this statistic.

Source: Educational Testing Service, State Adult Literacy Survey, 1992.

Section II 77

training. A primary goal was to investigate respondents' participation in various types of basic skills programs and their reasons for not enrolling in such programs. Respondents were also asked to express their opinions on particular literacy issues. These areas, and their relationship with literacy, are examined in the remaining pages of this section.

Enrollment in a basic skills program

Survey respondents were asked whether they were currently or previously enrolled in a program to improve their basic skills — that is, their basic reading, writing, and arithmetic skills. Six percent of the adults in Iowa reported that they had participated in such a program (Table 2.8).



IOWA TABLE 2.8

Literacy Levels and Average Proficiencies, by Enrollment in a Basic Skills Program: Results for Iowa

EVER ENROLLED IN				Percentage of adults in each literacy level											
A BASIC SKILLS PROGRAM?			Level 1 225 or lower		Lev 226 to	Level 2 226 to 275		el 3 o 325	Level 4 326 to 375	Level 5 376 or higher	Average Proficiency				
	n	WGT N (/1000)	PCT	RPCT (SE)	RPCT (SE)	RPCT (SE)	RPCT (SE)	RPCT (SE)	PROF (SE)			
Prose															
Yes	89	130	6	14 (6.9)	25 (8.2)	45 (9.0)	16 (7.7)	1(1.4)	281 (11.2)			
No	1,151	1,954	94	14 (3.6)	24 (2.3)	37 (5.4)	21 (1.6)	3 (0.8)	286 (3.2)			
Document										1					
Yes	89	130	6	14 (4.5)	31 (7.5)	421	8.0)	13 (4.3)	0 [†] (0.5)	276 (9.3)			
No	1,151	1,954	94	16 (3.1)	27 (1.8)	35 (3.2)	19 (2.1)	3 (1.1)	280 (3.1)			
Quantitative															
Yes	89	130	6	15 (5.4)	26 (6.0)	42 (8.1)	15 (6.8)	2 (2.0)	280 (9.5)			
No	1,151	1,954	94	15 (3.2)	22 (2.1)	35 (3.7)	23 (2.2)	4 (1.1)	288 (3.9)			

n = sample size; WGT N = population size estimate / 1,000 (the sample sizes for subpopulations may not add up to the total sample sizes, due to missing data); PCT = percentage in group; RPCT = row percentage estimate; PROF = average proficiency estimate; (SE) = standard error of the estimate (the reported sample estimate can be said to be within 2 standard errors of the true population value with 95% confidence).

Percentages less than 0.5 are rounded to zero.

Sample size is insufficient to permit a reliable estimate (fewer than 45 respondents).
 Interpret with caution -- the nature of the sample does not allow accurate determination of the variability of this statistic.

Source: Educational Testing Service, State Adult Literacy Survey, 1992.



On average, there were no significant differences in prose, document, or quantitative proficiency between individuals who said they had enrolled in a basic skills program and those who had not. Though the average scores of basic skills program participants appear to be lower on all three scales, the relatively large variability in the samples (reflected in the standard errors) prevents the differences from reaching statistical significance.

The data cannot tell us whether individuals who had participated in a basic skills program had lower proficiencies before they enrolled, and whether their skills improved as a result of their involvement. It may be the case that those who need the most help are not receiving it. Further analyses are needed to investigate the characteristics of basic skills program participants and of the target populations for these programs.

Main reason for not enrolling in a basic skills program

Iowa decision makers sought to collect information on the reasons why state residents might not participate in basic skills programs. Accordingly, survey participants in Iowa were given a list of potential reasons and asked to indicate which was the most important reason that would keep them from taking part in such a program.

Thirty-eight percent of the adults in Iowa indicated that they did not think they needed to improve their basic skills, and one-quarter said they did not have time to take part in a skills program (Table 2.9). Twelve percent said they had too many conflicts, and another 12 percent said they did not have any information about available basic skills programs. Seven percent of the Iowa respondents said they were too old to go back to school, and 5 percent said they did not like school. One percent said that school was too hard, and an equivalent proportion said it would take too long to finish a basic skills program. Iowa residents who said they did not believe they needed to improve their basic skills did, in fact, demonstrate higher average prose, document, and quantitative proficiencies than those who cited other reasons for not participating in a skills program. Across the three literacy scales, their average scores ranged from 312 to 320. There were no significant differences in performance, on average, among adults who identified other reasons for not taking part in such a program.

Effect of state literacy rate on employers' relocation decisions

Iowa decision makers wished to know whether Iowa residents believed that a state's literacy rate affects an out-of-state employer's decision to establish a new location there. When asked for their opinion on this issue, 75 percent of the

Section II 79



Literacy Levels and Average Proficiencies, by Reasons for Not Participating in a Basic Skills Program: Results for Iowa

REASONS FOR NOT PARTICIPATING IN A BASIC					Percer	ntage of adults	s in each litera	acy level	
SKILLS PROGRAM				Level 1 225 or lower	Level 2 226 to 275	Level 3 276 to 325	Level 4 326 to 375	Level 5 376 or higher	Average Proficiency
	n	WGT N (/1000)	PCT	RPCT (SE)	RPCT (SE)	RPCT (SE)	RPCT (SE)	RPCT (SE)	PROF (SE)
Prose									
Don't need to improve skills	499	672	38	3 (0.9)	13 (2.4)	39 (3,1)	38 (2.9)	7 (1.9)	318 (2.5)
Too old to go to school	78	121	7	27 (6,4)	29 (8.5)	35 (7.9)	9 (4.0)	01 (0.5)	253 (12.1)
School is too hard	14	25	1	*** (****)	()	*** (****)	*** (****)	*** (****)	*** (****)
Don't have time	271	415	24	5 (1.2)	27 (4.4)	46 (4.0)	20 (4.3)	1 (0.9)	293 (3.6)
Don't like school	45	79	5	15 (7.7)	44 (11.6)	29 (10.9)	10 (7.3)	1 (1.2)	267 (11.7)!
Too many conflicts	144	202	12	7 (4.1)	22 (5.4)	50 (5.5)	20 (4.5)	2 (1.6)	295 (5.8)
Would take too long	14	23	1	*** (****)	*** (****)	*** (****)	*** (****)	*** (****)	*** (****)
Don't have information	135	213	12	11 (3.4)	36 (3.8)	39 (5.2)	12 (4.7)	1 (1.4)	277 (7.4)!
Document									
Don't need to improve skills	499	672	38	3 (0.6)	16 (3.0)	40 (2.7)	34 (3.5)	6 (2.9)	312 (3.3)
Too old to go to school	78	121	7	26 (8.4)	36 (7.4)	31 (7.9)	7 (3.1)	0 [†] (0.0)	249 (12.3)!
School is too hard	14	25	1	*** (****)	*** (****)	*** (****)	*** (****)	*** (****)	*** (****)
Don't have time	271	415	24	4 (1.8)	30 (4.1)	45 (3.9)	19 (3.6)	2 (1.1)	292 (3.6)
Don't like school	45	79	5	17 (10.5)	45 (10.7)	26 (8.2)	11 (5.9)	1 (0.7)	265 (10.6)!
Too many conflicts	144	202	12	7 (3.7)	23 (4.9)	52 (6.5)	17 (4.0)	1 (1.2)	292 (4.4)
Would take too long	14	23	1	*** (****)	*** (****)	*** (****)	•••• (•••••)	*** (****)	*** (****)
Don't have information	135	213	12	15 (4.6)	35 (6.7)	37 (7.3)	13 (5.3)	0 ^T (0.8)	273 (9.3)!
Quantitative									
Don't need to improve skills	499	672	38	2(09)	12 (2 1)	39 (27)	38 (3 3)	9(24)	320 (2 6)
Too old to go to school	78	121	7	25 (8.3)	26 (73)	38 (7 7)	10 (4 2)	1(0.5)	258 (13 1)
School is too hard	14	25	1	*** (****)	*** (****)	*** (****)	*** (****)	*** (****)	*** (****)
Don't have time	271	415	24	5 (1.9)	24 (5.0)	44 (5.8)	24 (3.7)	3 (1.6)	299 (4.1)
Don't like school	45	79	5	19 (11.3)	34 (10.7)	32 (15.2)	14 (6.5)	1 (1.9)	271 (13.0)!
Too many conflicts	144	202	12	7 (3.5)	21 (4.7)	44 (7.2)	25 (6.2)	3 (2.0)	298 (6.0)
Would take too long	14	23	1	*** (****)	*** (****)	*** (****)	*** (****)	*** (****)	*** (****)
Don't have information	135	213	12	14 (4.7)	33 (7.8)	36 (9.8)	16 (5.8)	2 (2.2)	278 (10.0)!

n = sample size; WGT N = population size estimate / 1,000 (the sample sizes for subpopulations may not add up to the total sample sizes, due to missing data); PCT = percentage in group; RPCT = row percentage estimate; PROF = average proficiency estimate; (SE) = standard error of the estimate (the reported sample estimate can be said to be within 2 standard errors of the true population value with 95% confidence).

[†] Percentages less than 0.5 are rounded to zero.

*** Sample size is insufficient to permit a reliable estimate (fewer than 45 respondents).

Interpret with caution -- the nature of the sample does not allow accurate determination of the variability of this statistic.

Source: Educational Testing Service, State Adult Literacy Survey, 1992.

80 Education and Training

Iowa respondents said they believed that a state's literacy rate does influence employers' decision making. Nineteen percent did not believe this has an impact, and 6 percent of the respondents had no opinion about the matter (Table 2.10).

Interestingly, those who expressed the opinion that a state's literacy rate affects employers' location decisions demonstrated higher average prose, document, and quantitative proficiencies than did those who either did not believe this was the case or who had no opinion on the issue.



IOWA TABLE 2.10

Literacy Levels and Average Proficiencies, by Opinion as to the Effect of a State's Literacy Rate on Employers' Business Decisions: Results for Iowa

DOES LITERACY				Percentage of adults in each literacy level													
EMPLOYER'S LOCATION DECISION?						Leve 225 or	el 1 Iower	Leve 226 to	el 2 275	Lev 276 to	el 3 5 325	Leve 326 to	el 4 9 375	Leve 376 or h	el 5 higher	Avera Profici	age ency
	n	WGT N (/1000)	PCT	RPCT (SE)	RPCT (SE)	RPCT (SE)	RPCT (SE)	RPCT (SE)	PROF (SE)		
Prose					0.01	01/	2.01	40 /	2.0)	29 (3.4)	1	1.4)	303 (5.3)		
Yes	928	1,324	75	5(0.9)	21 (5.8)	42 (6.0)	15 (4.5)	2(0.9)	278 (6.9)		
No opinion	69	104	6	26 (7.7)	34 (9.8)	33 (8.4)	7 (4.2)	0†(0.3)	256 (8.9)!		
Document								1			~	1					
Yes	928	1,324	75	6 (1.9)	24 (3.2)	42 (2.3)	25 (3.1)	3(1.6)	299 (5.9)		
No	221	346	19	15 (4.0)	34 (5.0)	35 (6.5)	15 (4.9)	1(1.0)	2/5 (7.3)		
No opinion	69	104	6	31 (7.1)	28 (6.6)	35 (8.1)	6 (5.2)	01(0.3)	253 (7.3)!		
Quantitative										1							
Yes	928	1,324	75	5 (1.9)	18 (2.9)	41	(2.3)	30 (3.2)	6 (2.1)	306 (6.0)		
No	221	346	19	13 (3.6)	29 (5.4)	38	(6.1)	18 (3.7)	2(1.5)	282 (7.3)		
No opinion	69	104	6	29 (7.3)	29 (5.8)	34	(8.2)	8 (6.0)	01(0.9)	257 (8.2)!		

n = sample size; WGT N = population size estimate / 1,000 (the sample sizes for subpopulations may not add up to the total sample sizes, due to missing data); PCT = percentage in group; RPCT = row percentage estimate; PROF = average proficiency estimate; (SE) = standard error of the estimate (the reported sample estimate can be said to be within 2 standard errors of the true population value with 95% confidence).

† Percentages less than 0.5 are rounded to zero.

*** Sample size is insufficient to permit a reliable estimate (fewer than 45 respondents).

I Interpret with caution -- the nature of the sample does not allow accurate determination of the variability of this statistic.

Source: Educational Testing Service, State Adult Literacy Survey, 1992.

Opinion as to employers' obligation to provide literacy education to employees

Decision makers in Iowa also were interested in knowing what percentage of the state's residents believed that employers had an obligation to provide literacy education to employees who need assistance. When asked to express their opinion on this matter, slightly more than half (58 percent) of the Iowa respondents supported the view that employers are obligated to offer literacy education to employees (Table 2.11). Thirty-six percent did not support this view, and 7 percent had no opinion about this matter.



IOWA TABLE 2.11

Literacy Levels and Average Proficiencies, by Opinion as to Employers' Obligation to Provide Literacy Education: Results for Iowa

SHOULD	-			Percentage of adults in each literacy level														
PROVIDE LITERACY EDUCATION?	PROVIDE ITERACY UCATION?	1?						el 1 Iower	Lev 226 to	el 2 o 275	Lev 276 to	el 3 o 325	Lev 326 to	el 4 o 375	Lev 376 or	el 5 higher	Aver Profic	rage liency
	n	WGT N (/1000)	PCT	RPCT (SE)	RPCT (SE)	RPCT (SE)	RPCT (SE)	RPCT (SE)	PROF (SE)			
Prose	14					0												
Yes	708	1.020	58	8 (1.8)	26 (4.4)	41 (3.0)	221	4.0)	31	12)	203 (6 7)			
No	436	636	36	6(1.8)	19 (2.5)	421	3.0)	291	2.8)	41	13)	301 (5.2)			
No opinion	73	116	7	14 (5.3)	25 (9.8)	37 (9.4)	19 (3.9)	5(2.8)	285 (8.4)!			
Document												1						
Yes	708	1,020	58	9(3.2)	28 (3.4)	40 (3.8)	201	2 91	31	1.8)	200 (7 1)			
No	436	636	36	7(2.1)	23 (3.3)	40 (27)	26 (37)	31	11)	296 (5.6)			
No opinion	73	116	7	18 (5.7)	25 (8.3)	41 (7.0)	13 (6.0)	4(2.4)	278 (8.7)!			
Quantitative																		
Yes	708	1,020	58	9(2.9)	24 (3.6)	40 (37)	231	2 01	A1	1.8)	205 (74)			
No	436	636	36	6 (2.4)	171	2.6)	39 (2.9)	32 (2.9)	61	23)	306 (5.7)			
No opinion	73	116	7	14 (5.4)	211	79)	401	9.9)	21 (71)	51	3.5)	288 (10 21			

n = sample size; WGT N = population size estimate / 1,000 (the sample sizes for subpopulations may not add up to the total sample sizes, due to missing data); PCT = percentage in group; RPCT = row percentage estimate; PROF = average proficiency estimate; (SE) = standard error of the estimate (the reported sample estimate can be said to be within 2 standard errors of the true population value with 95% confidence).

Percentages less than 0.5 are rounded to zero.

82 Education and Training

- *** Sample size is insufficient to permit a reliable estimate (fewer than 45 respondents).
 - Interpret with caution -- the nature of the sample does not allow accurate determination of the variability of this statistic.

Source: Educational Testing Service, State Adult Literacy Survey, 1992.

There were no significant differences in the average literacy proficiencies of Iowa adults who believed that employers should provide literacy education, those who did not, and those who did not have an opinion. Though some of the differences among these groups appear to be large, the variability in the samples (indicated by the large standard errors) prevents them from reaching statistical significance.

Summary

In general, the educational attainments of adults in Iowa were similar to those of adults nationwide. As expected, adults who had completed higher levels of schooling outperformed those with more limited education. The average proficiencies of adults who had completed nine to 12 years of education were about 100 points lower than the average scores of those who had finished at least some graduate work, representing a great difference in the difficulty and complexity of literacy skills and strategies.

Some high school graduates in the state, region, and nation did poorly in the assessment. On each literacy scale, 8 to 12 percent of the high school graduates in Iowa performed in the Level 1 range, and another 27 to 34 percent performed in Level 2. As expected, adults who had completed a two-year college degree outperformed those whose highest level of education was a high school diploma, and four-year college graduates performed better still.

The performance differences among various subpopulations can be at least partly explained by differences in years of schooling. Older adults tended to have completed fewer years of schooling than younger adults, for example.

Further, the more years of schooling respondents' parents had completed, the more education they themselves were likely to have had.

One-third of the school dropouts in Iowa had participated in a GED or high school equivalency program. Fifty-seven percent of the program participants had earned a diploma, and their average scores were significantly higher than those of participants who did not earn one. Most GED program participants were age 25 or older.

Eleven percent of the adults in Iowa and an equivalent percentage nationwide were currently enrolled in school or college, and their average prose, document, and quantitative scores were significantly higher than those of respondents who were not enrolled. Nationwide, 10 percent of those enrolled said they expected to earn a high school diploma or equivalency, 13 percent were pursuing an associate's degree, 38 percent were working on a four-year college degree, and 19 percent were working toward an advanced degree. Six percent of the Iowa residents were currently or previously enrolled in a program to improve their basic skills. The average proficiencies of individuals who said they had participated in such a program were no different, on average, from the scores of those who had not.

Thirty-eight percent of the Iowa respondents said they did not think they needed to improve their basic skills. Their average proficiencies were significantly higher than those of respondents who indicated various other reasons for not participating in a basic skills program. One-quarter of the adults in the state said they would not take part in a basic skills program because they did not have the time, 12 percent said they had too many conflicts, and another 12 percent said they did not have any information about available programs.

Three-quarters of the Iowa respondents believed that a state's literacy rate affects an out-of-state employer's decision about establishing a new location there. Their average literacy scores were higher than those of adults who did not share this opinion. Fifty-eight percent of the adults in Iowa believed that employers should provide literacy education to employers who need assistance, but their literacy skills were no different, on average, from those who disagreed.









SECTION III

SECTION III

Employment, Economic Status, and Civic Responsibility

he first two sections of this report offered a portrait of the literacy skills of adults in Iowa and illuminated some of the relationships that exist between education and literacy. In this section, the focus shifts to the connections between literacy and other aspects of adults' lives — including their employment, earnings, economic status, and civic participation.

The State and National Adult Literacy Surveys gathered information from household survey respondents on an array of social and economic variables, making it possible to examine the extent to which adults' literacy proficiencies vary according to their employment and economic characteristics. Do adults who are employed, who hold certain types of jobs, or who earn high wages tend to demonstrate advanced literacy skills? Are individuals who are poor or near poor, or who rely on public assistance or food stamps, more likely than their more affluent peers to perform in the lowest literacy levels? Do the literacy proficiencies of voters tend to differ from those of nonvoters? These types of questions are addressed in the pages that follow.

Employment

While our nation's concerns over adult literacy appropriately encompass all areas of life, in recent years much attention has been focused on the role that literacy plays in the workplace. Accordingly, one of the primary aims of this survey was to explore the connections between adults' work lives and their literacy skills. Respondents were asked a series of questions about their employment status and their current or most recent jobs. This section examines the relationships between adults' responses to these questions and their performance in the literacy assessment.

Section III 87

Labor force status

Household survey participants were asked to indicate what their employment situation had been during the week before the survey. Approximately half (52 percent) of the adults in Iowa reported that they were employed full time, and another 14 percent said they were employed part time (Table 3.1P,D,Q). Approximately 5 percent of the state's residents were unemployed, laid off, or looking for work. Nearly one-third (30 percent) were out of the labor force that is, not employed and not looking for work. (These include adults who are



T

IOWA TABLE 3.1P

Prose Literacy Levels and Average Proficiencies, by Labor Force Status: Results for Iowa, the Midwest, and the Nation

LABOR FORCE STATUS						P	ercenta	age of a	idults in	each pr	ose lite	eracy lev	el		
				Lev 225 or	el 1 Iower	Lev 226 to	el 2 o 275	Le 276	vel 3 to 325	Lev 326 t	el 4 o 375	Lev 376 or	el 5 higher	Aver	age lency
	n	WGT N (/1000)	PCT	RPCT (SE)	RPCT (SE)	RPCT	(SE)	RPCT (SE)	RPCT (SE)	PROF (SE)
lowa												1			
Employed full-time	741	1,068	52	7(1.3)	22 (3.1)	42	(2.7)	26 (2.6)	41	1.1)	299 (4.2)
Employed part-time	170	276	14	4(1.8)	23 (5.0)	46	(5.9)	24 (6.5)	3(1.9)	300 (6.6)
Unemployed	69	96	5	16 (6.5)!	29 (9.0)!	34	(8.4)!	19 (8.0)!	1(1.6)1	279 (11.2)!
Out of labor force	230	604	30	32 (8.2)	30 (9.8)	27	(12.6)	10 (5.1)	1(0.4)	254 (11.1)
Midwest															
Employed full-time	3,714	20,595	47	8(0.6)	24 (1.2)	40	(1.8)	241	1.3)	41	0.5)	296 (1.7)
Employed part-time	935	5,681	13	9 (1.6)	26 (3.5)	42	4.6)	19 (2.5)	41	1.1)	291 (2.8)
Unemployed	545	3,036	7	19 (2.5)	36 (3.4)	32	3.2)	121	3.1)	11	0.6)	267 (3.5)
Out of labor force	1,782	14,188	33	32 (2.0)	32 (1.9)	25	1.9)	10 (1.3)	1(0.3)	252 (2.3)
Nation															
Employed full-time	12,466	89,723	48	13 (0.6)	24 (0.7)	36	1.0)	23 (0.7)	51	0.3)	288 (0.9)
Employed part-time	3,051	23,600	13	14 (0.8)	26 (1.5)	36	1.8)	20 (1.3)	41	0.5)	284 (1.4)
Unemployed	1,942	13,557	7	24 (1.3)	35 (1.7)	29	2.7)	11 (1.8)	11	0.4)	260 (2.1)
Out of labor force	6,721	58,202	31	35 (0.8)	30 (1.0)	25	0.9)	9(0.7)	10	0.3)	246 (1.1)

n = sample size; WGT N = population size estimate / 1,000 (the sample sizes for subpopulations may not add up to the total sample sizes, due to missing data); PCT = percentage in group; RPCT = row percentage estimate; PROF = average proficiency estimate; (SE) = standard error of the estimate (the reported sample estimate can be said to be within 2 standard errors of the true population value with 95% confidence).

- Percentages less than 0.5 are rounded to zero.
- Sample size is insufficient to permit a reliable estimate (fewer than 45 respondents).
 Interpret with caution the nature of the sample does not allow accurate determination of the variability of this statistic.
- Source: Educational Testing Service, State Adult Literacy Survey, 1992.


in school, keeping house, retired, or doing volunteer work.) The distribution of individuals across these labor force categories was almost identical for the Midwest region and the nation as a whole.

Individuals in Iowa who were working full time and those working part time performed similarly in each of the three dimensions of literacy examined. On the prose scale, the percentages of full-time and part-time employees who performed in each literacy level were almost identical, and their average proficiencies were therefore essentially the same (299 and 300, respectively). Although there appear to be differences between these two groups on the quantitative scale, the gap is not statistically significant.



IOWA TABLE 3.1D

Document Literacy Levels and Average Proficiencies, by Labor Force Status: Results for Iowa, the Midwest, and the Nation

ABOR FORCE STATUS						Perc	entage	of adu	lts in ea	ach docu	ment li	teracy le	vel		
				Level 225 or lo	1 wer	Leve 226 to	el 2 275	Lev 276 t	el 3 o 325	Leve 326 to	el 4 9 375	Leve 376 or l	el 5 higher	Avera Profici	age ency
	n	WGT N (/1000)	PCT	RPCT (S	SE)	RPCT (SE)	RPCT (SE)	RPCT (SE)	RPCT (SE)	PROF (SE)
lowa										05.4			4.43	206 (2 7)
Employed full-time	741	1,068	52	7(1	.4)	25 (3.5)	40 (3.1)	25 (2.9)	3(2.5)	296 (7.9)
Employed part-time	170	276	14	6(2	(8.2	2/(8.2)1	44	7.5)1	16 (6.2)	21	1.8)	278 (10.9)1
Out of labor force	69 230	96 604	30	37 (8	3.6)	33 (6.0)	23	(12.1)	6(2.8)	0†(0.4)	244 (8.0)
Midwost															
Employed full time	3714	20.595	47	9(0	(8.0	27 (1.5)	39	(2.0)	22 (1.4)	3 (0.7)	292 (1.8)
Employed nul-time	935	5.681	13	12(1	1.8)	29 (2.2)	38	(3.7)	18 (2.2)	3 (1.0)	284 (2.7)
Unemployed	545	3,036	7	22 (2	2.6)	36 (3.8)	30	(4.1)	10 (2.2)	2 (0.9)	263 (3.9)
Out of labor force	1,782	14,188	33	36 (2	2.2)	33 (3.1)	22	(2.2)	8 (1.2)	1 (0.2)	245 (2.7)
Nation											-		12.00		
Employed full-time	12,466	89,723	48	14 ((0.7)	26 (0.6)	35	(0.7)	21 (0.7)	4 (0.3)	284 (0.9)
Employed part-time	3,051	23,600	13	17 ((0.9)	29 (1.3)	34	(1.7)	17 (1.0)	3(0.4)	217 (1.3)
Unemployed	1,942	13,557	7	26 (1.2)	34 (1.7)	29	(1.6)	9(1.1)	1(0.4)	237 (13)
Out of labor force	6,721	58,202	31	39 (1.0)	31 (0.9)	22	(0.8)	1 1	0.5)	111	0.1)	201	1.0)

n = sample size; WGT N = population size estimate / 1,000 (the sample sizes for subpopulations may not add up to the total sample sizes, due to missing data); PCT = percentage in group; RPCT = row percentage estimate; PROF = average proficiency estimate; (SE) = standard error of the estimate (the reported sample estimate can be said to be within 2 standard errors of the true population value with 95% confidence).

Percentages less than 0.5 are rounded to zero.

*** Sample size is insufficient to permit a reliable estimate (fewer than 45 respondents).

Interpret with caution -- the nature of the sample does not allow accurate determination of the variability of this statistic.

Source: Educational Testing Service, State Adult Literacy Survey, 1992.

Section III 89

The performance of employed adults differed sharply from that of adults who were either unemployed or out of the labor force. Across the three scales, between 25 and 33 percent of the employed adults in Iowa performed in Levels 1 and 2, in contrast to approximately 45 percent of unemployed adults and roughly two-thirds of adults who were out of the labor force. Conversely, employed adults were much more likely to reach the highest literacy levels; on the document scale, for example, 25 percent of full-time employees attained Level 4, and 3 percent reached Level 5. The proportions of adults who were either unemployed or out of the labor force who reached these uppermost levels were far smaller.



IOWA TABLE 3.1Q

Quantitative Literacy Levels and Average Proficiencies, by Labor Force Status: Results for Iowa, the Midwest, and the Nation

ABOR FORCE STATUS						Perc	entage	of adul	ts in ea	ch quan	titative	literacy	level		
				Lev 225 or	el 1 Iower	Lev 226 t	el 2 o 275	Lev 276 t	rel 3 o 325	Lev 326 to	el 4 5 375	Lev 376 or	el 5 higher	Ave	rage
	n	WGT N (/1000)	PCT	RPCT (SE)	RPCT (SE)	RPGT (SE)	RPCT (SE)	RPCT (SE)	PROF (SE)
Iowa										1					
Employed full-time	741	1,068	52	6 (1.2)	19 (3.2)	39 (2.4)	31 (2.7)	6(1.7)	306 (4.6)
Employed part-time	170	276	14	5 (2.8)	25 (6.2)	45 (7.3)	22 (4.1)	3(1.4)	296 (7.2)
Unemployed	69	96	5	23 (9.0)1	23 (9.0)1	34 (9.4)1	18 (7.0)1	3(2.2)1	275 (10.0)1
Out of labor force	230	604	30	34 (8.9)	29 (7.8)	26 (11.8)	9(4.9)	2 (1.2)	251 (12.8)
Midwest						1000									
Employed full-time	3,714	20,595	47	8 (0.9)	22 (2.1)	38 (1.8)	26 (1.5)	6(0.6)	299 (2.1)
Employed part-time	935	5,681	13	11 (1.9)	28 (2.8)	38 (3.7)	19 (2.5)	4 (1.3)	288 (2.8)
Unemployed	545	3,036	7	26 (2.7)	33 (3.7)	28 (3.3)	10 (2.1)	21	1.1)	261 (4.2)
Out of labor force	1,782	14,188	33	31 (2.4)	29 (1.8)	27 (2.4)	11 (1.5)	1(0.3)	252 (3.7)
Nation															
Employed full-time	12,466	89,723	48	13 (0.6)	23 (0.9)	35 (1.1)	23 (0.6)	6 (0.3)	290 (0.9)
Employed part-time	3,051	23,600	13	15 (1.1)	27 (1.3)	36 (1.6)	18 (1.3)	31	0.5)	280 (1.5)
Unemployed	1,942	13,557	7	28 (1.5)	32 (1.8)	28 (2.0)	10 (1.3)	21	0.4)	256 (1.9)
Out of labor force	6.721	58,202	31	37 (1.0)	27 (0.8)	24 (0.8)	101	0.7)	21	0.3)	241 (1.6)

- n = sample size; WGT N = population size estimate / 1,000 (the sample sizes for subpopulations may not add up to the total sample sizes, due to missing data); PCT = percentage in group; RPCT = row percentage estimate; PROF = average proficiency estimate; (SE) = standard error of the estimate (the reported sample estimate can be said to be within 2 standard errors of the true population value with 95% confidence).
 - Percentages less than 0.5 are rounded to zero.
- *** Sample size is insufficient to permit a reliable estimate (fewer than 45 respondents). 1
- Interpret with caution -- the nature of the sample does not allow accurate determination of the variability of this statistic.
- Source: Educational Testing Service, State Adult Literacy Survey, 1992.

90 Employment, Economic Status . . .

The average proficiency results also reflect this strong association between literacy and labor force status. For example, while full-time employees in Iowa had an average quantitative score of 306 (within the Level 3 range), the average score of unemployed adults was 275, and of adults not in the labor force, 251 (both in the Level 2 range). Similar patterns are found in the regional and national results. The only notable variation is that in the regional and national results there are statistically significant differences between the average scores of unemployed adults and the average scores of those out of the labor force.

Occupation

While it might be useful to know the level of literacy skills required to find, hold, and succeed in various types of jobs, research in this area has been limited. Such questions can be approached, however, by looking at the literacy skills of adults within certain types of occupations.

Accordingly, household survey participants were asked to describe the type of work they performed in their current or most recent jobs, and this information was sorted into occupational categories using the Census Classification for Industries and Occupations. These categories were then recombined into four occupational groupings: professional, managerial, or technical; sales or clerical; craft or service; and labor, assembly, fishing, or farming.

Twenty percent of the adults in Iowa worked in managerial, professional, or technical jobs; 23 percent were in sales or clerical occupations; 29 percent worked in craft or service occupations; and 28 percent were in labor, assembly, fishing, or farming jobs (Table 3.2P,D,Q). These numbers are somewhat different from the national figures. Most notably, the percentages of Iowa adults who reported working in labor, assembly, fishing, or farming jobs are higher than the proportion of adults nationwide who did so.

A strong connection exists between literacy and occupation. Although some individuals in managerial or professional jobs displayed limited literacy skills, they were less likely than respondents in other types of jobs to perform in the lowest literacy levels and more likely to attain the highest levels defined. On the quantitative scale, for example, 1 percent of Iowa adults in professional, managerial, or technical positions performed in Level 1, in contrast to 8 percent of craft or service workers and 13 percent of laborers. Seven percent of the adults in professional jobs performed in Level 2 on this scale, compared with 17 to 27 percent of the adults in each of the other occupational categories. In contrast, 56 percent of the adults in professional, managerial, or technical jobs performed in Levels 4 and 5, compared with one-third of adults in sales or clerical jobs and 23 percent of those in craft, service, or labor jobs.

Section III 91



IOWA TABLE 3.2P

Prose Literacy Levels and Average Proficiencies, by Occupational Category: Results for Iowa, the Midwest, and the Nation

OCCUPATIONAL CATEGORY						P	ercenta	age of a	dults in	each p	rose lit	eracy le	vel		
				Lev 225 or	el 1 Iower	Lev 226 t	el 2 o 275	Lev 276 t	el 3 o 325	Lev 326 1	rel 4 to 375	Lev 376 or	el 5 higher	Aver	rage clency
	n	WGT N (/1000)	PGT	RPCT (SE)	RPCT (SE)	RPCT (SE)	RPCT	SE)	RPCT (SE)	PROF	SE)
Iowa															
Professional, Manager	278	321	20	1(1.3)	71	2.3)	36 (3.6)	46 (4.3)	11(3.1)	330 (4.5)
Sales, Clerical	250	380	23	1(0.6)	18 (4.1)	47 (4.8)	30 (3.5)	4 (1.6)	309 (3.8)
Craft, Service	297	462	29	8 (2.7)	29 (3.5)	46 (3.8)	16 (3.1)	1 (0.9)	286 (4.8)
Laborer, Assembler	272	456	28	15 (4.2)	33 (7.6)	38 (4.5)	13 (3.1)	1(1.1)	276 (5.1)
Midwest															
Professional, Manager	1,560	7,724	22	2(0.8)	11 (1.7)	35 (2.2)	42 (2.4)	11(1.3)	325 (2.3)
Sales, Clerical	1,821	9,498	28	5 (0.8)	23 (1.9)	47 (2.6)	22 (2.0)	3(0.7)	298 (1.7)
Craft, Service	1,632	9,887	29	14 (1.2)	33 (2.1)	38 (3.3)	13 (1.6)	1(0.5)	276 (1.8)
Laborer, Assembler	1,148	7,286	21	19 (2.2)	36 (3.1)	34 (2.8)	10 (1.5)	1(0.5)	266 (2.8)
Nation															
Professional, Manager	5,461	35,599	24	3(0.4)	13 (1.0)	34 (1.2)	39 (1.1)	11(0.7)	322 (1.0)
Sales, Clerical	6,544	41,713	28	8(0.6)	25 (0.9)	43 (1.4)	21 (1.1)	3(0.4)	293 (1.1)
Craft, Service	5,614	42,187	29	22 (0.8)	32 (1.1)	33 (1.1)	12 (0.8)	11	0.2)	264 (1.1)
Laborer, Assembler	3,479	27,671	19	29 (1.3)	33 (1.4)	29 (1.3)	8 (0.7)	11	0.2)	249 (1.8)

n = sample size; WGT N = population size estimate / 1,000 (the sample sizes for subpopulations may not add up to the total sample sizes, due to missing data); PCT = percentage in group; RPCT = row percentage estimate; PROF = average proficiency estimate; (SE) = standard error of the estimate (the reported sample estimate can be said to be within 2 standard errors of the true population value with 95% confidence).

Percentages less than 0.5 are rounded to zero. T

*** Sample size is insufficient to permit a reliable estimate (fewer than 45 respondents).

Interpret with caution - the nature of the sample does not allow accurate determination of the variability of this statistic.

Source: Educational Testing Service, State Adult Literacy Survey, 1992.

As a result of these differences in levels of performance, Iowa residents with professional, managerial, or technical positions had higher average literacy proficiencies than adults working in other types of jobs. Their average score on the prose scale was 330, for example, while that of adults in labor, assembly, fishing, or farming positions was 276; that of adults in craft or service positions was 286; and that of adults in sales or clerical positions was 309.

92 Employment, Economic Status . . .



IOWA TABLE 3.2D

Document Literacy Levels and Average Proficiencies, by Occupational Category: Results for Iowa, the Midwest, and the Nation

OCCUPATIONAL CATEGORY			-			Perc	entage	of adult	ts in ea	ach docu	Iment	literacy le	evel		
				Leve 225 or	el 1 Iower	Lev 226 to	el 2 275	Leve 276 to	el 3 5 325	Lev 326 to	el 4 5 375	Leve 376 or l	el 5 higher	Aver Profici	age ency
	n	WGT N (/1000)	PCT	RPCT (SE)	RPCT (SE)	RPCT (SE)	RPCT (SE)	RPCT (SE)	PROF (SE)
lowa															-
Professional, Manager	278	321	20	2 (1.4)	10 (2.8)	39 (6.3)	41 (5.5)	9(3.6)	323 (5.1)
Sales, Clerical	250	380	23	1(0.8)	23 (5.9)	47 (6.8)	26 (5.1)	3 (1.8)	305 (4.8
Craft, Service	297	462	29	9 (3.7)	31 (4.6)	42 (3.5)	16 (3.2)	1 1(0.9)	285 (5.2
Laborer, Assembler	272	456	28	14 (1.8)	35 (5.9)	35 (4.9)	15 (2.4)	1(0.9)	2/4 (3.6,
Midwest									1.20	1					
Professional, Manager	1,560	7,724	22	3 (1.0)	14 (1.8)	38 (2.4)	37 (2.8)	8 (1.7)	318 (2.2
Sales, Clerical	1,821	9,498	28	6 (1.1)	27 (1.7)	44 (2.4)	20 (2.0)	2 (0.8)	293 (1.5
Craft, Service	1,632	9,887	29	16 (1.7)	35 (2.1)	35 (3.1)	13 (1.8)	1(0.6)	274 (1.9
Laborer, Assembler	1,148	7,286	21	20 (2.2)	36 (3.1)	33 (2.2)	10 (1.7)	1(0.4)	264 (3.1
Nation												1		-	3
Professional, Manager	5.461	35,599	24	4 (0.6)	15	0.8)	37 (1.1)	35	1.3)	9 (0.7)	315 (1.0
Sales, Clerical	6,544	41,713	28	9(0.7)	29	1.0)	40 (1.4)	19	1.0)	2 (0.3)	287 (1.0
Craft, Service	5,614	42,187	29	23 (0.8)	33	1.1)	31 (1.4)	11	0.9)	1(0.2)	262	1.2
Laborer, Assembler	3,479	27,671	19	30 (1.3)	33	1.4)	28 (1.4)	8	0.6)	1(0.2)	247	1.7

n = sample size; WGT N = population size estimate / 1,000 (the sample sizes for subpopulations may not add up to the total sample sizes, due to missing data); PCT = percentage in group; RPCT = row percentage estimate; PROF = average proficiency estimate; (SE) = standard error of the estimate (the reported sample estimate can be said to be within 2 standard errors of the true population value with 95% confidence).

Percentages less than 0.5 are rounded to zero.
*** Sample size is insufficient to permit a reliable estimate (fewer than 45 respondents).
Interpret with caution -- the nature of the sample does not allow accurate determination of the variability of this statistic.

Source: Educational Testing Service, State Adult Literacy Survey, 1992.

In viewing these results, it is important to remember that education is strongly associated with literacy skills, and that adults in professional, managerial, or technical positions are likely to have higher levels of education than adults in other types of positions. It is also true that many of these positions offer or require continuing education and training opportunities that enable individuals to further enhance their proficiencies.¹

¹N.P. Eurich. (1990). The Learning Industry: Education for Adult Workers. Princeton, NJ: The Carnegie Foundation for the Advancement of Teaching.

Section III 93



IOWA TABLE 3.2Q

Quantitative Literacy Levels and Average Proficiencies, by Occupational Catego Results for Iowa, the Midwest, and the Nation

OCCUPATIONAL CATEGORY						Perc	entage	of adult	s in ea	ich quai	ntitative	literacy	level		
				Lev 225 or	el 1 Iower	Lev 226 t	el 2 o 275	Lev 276 to	el 3 o 325	Lev 326 1	rel 4 to 375	Lev 376 or	el 5 higher	Ave: Profic	rage lency
	n	WGT N (/1000)	PCT	RPCT (SE)	RPCT (SE)	RPCT (SE)	RPCT	(SE)	RPCT (SE)	PROF (SE)
Iowa															
Professional, Manager	278	321	20	1(1.3)	7 (3.4)	36 (4.5)	44	4.2)	12 (2.6)	329 (4.9)
Sales, Clerical	250	380	23	1 (0.8)	17 (4.8)	47 (6.2)	30	5.0)	5 (2.4)	311 (3.7)
Craft, Service	297	462	29	8 (2.9)	27 (4.7)	43 (5.4)	20 (4.5)	3 (1.7)	291 (5.7)
Laborer, Assembler	272	456	28	13 (2.4)	26 (4.3)	37 (5.3)	20 (3.1)	3 (1.5)	286 (5.2)
Midwest															
Professional, Manager	1,560	7,724	22	3 (0.7)	13 (2.0)	33 (2.1)	40 (2.7)	12 (1.5)	324 (2.2)
Sales, Clerical	1,821	9,498	28	6 (1.1)	25 (1.7)	43 (2.3)	23 (1.7)	4 (0.8)	297 (1.9)
Craft, Service	1,632	9,887	29	16 (1.5)	31 (2.5)	35 (2.2)	15 (1.0)	2(0.8)	277 (2.2)
Laborer, Assembler	1,148	7,286	21	18 (2.6)	30 (4.8)	36 (3.5)	14 (2.0)	2 (0.5)	272 (2.9)
Nation															
Professional, Manager	5,461	35,599	24	4 (0.5)	14 (0.9)	34 (1.2)	36 (0.8)	13 (0.7)	322 (1.0)
Sales, Clerical	6,544	41,713	28	9 (0.5)	25 (0.8)	41 (1.4)	21 (1.1)	3 (0.3)	292 (1.1)
Craft, Service	5,614	42,187	29	24 ((8.0	30 (1.2)	32 (1.2)	13 (0.7)	2(0.4)	264 (1.3)
Laborer, Assembler	3,479	27,671	19	29 (1.6)	30 (1.6)	30 (1.8)	10 (1.4)	1(0.3)	253 (2.0)

n = sample size; WGT N = population size estimate / 1,000 (the sample sizes for subpopulations may not add up to the total sample sizes, due to missing data); PCT = percentage in group; RPCT = row percentage estimate; PROF = average proficiency estimate; (SE) = standard error of the estimate (the reported sample estimate can be said to be within 2 standard errors of the true population value with 95% confidence).

*** Sample size is insufficient to permit a reliable estimate (fewer than 45 respondents).

Source: Educational Testing Service, State Adult Literacy Survey, 1992.

Weeks worked

Household survey participants, regardless of their current or recent employment status, were asked how many weeks they had worked in the past 12 months. On each scale, and in each population — state, region, and nation — adults who performed in Levels 3, 4, and 5 worked more weeks in the past year than those in Level 2, who, in turn, worked more weeks than those in

Level 1 (Table 3.3).

In fact, the number of weeks worked increases dramatically across the literacy levels. On each scale, Iowa respondents who performed in the lowest level worked, on average, only about 13 or 14 weeks a year. In contrast,

94.... Employment, Economic Status ...



IOWA TABLE 3.3

Average Number of Weeks Worked in the Past 12 Months, by Literacy Level: Results for Iowa, the Midwest, and the Nation

	LITERACY	Lev 225 or	rel 1 lower	Lev 226 t	el 2 o 275	Lev 276 t	el 3 o 325	Lev 326	vel 4 to 375	Lev 376 or	el 5 higher
		WKS	(SE)								
Prose lowa Midwest Nation Document lowa Midwest Nation		14 16 19 14 16 19	(5.3) (0.7) (0.5) (6.4) (1.1) (0.5)	27 26 27 31 27 29	(6.7) (0.7) (0.4) (4.9) (0.6) (0.3)	39 35 35 37 35 35	(1.9) (0.8) (0.4) (1.7) (0.9) (0.4)	39 37 38 41 40 40	(1.8) (1.0) (0.4) (1.5) (0.8) (0.4)	43 44 44 41 42 43	(1.8) (1.8) (0.7) (2.2) (2.1) (0.8)
Quantitative lowa Midwest Nation		13 15 18	(5.9) (1.5) (0.5)	29 28 29	(6.3) (0.7) (0.4)	37 34 34	(1.5) (1.0) (0.4)	40 39 39	(1.9) (0.8) (0.4)	45 42 40	(2.1) (2.5) (0.8)

WKS = average number of weeks worked in the past 12 months; (SE) = standard error of the estimate (the reported sample estimate can be said to be within 2 standard errors of the true population value with 95% confidence).

Source: Educational Testing Service, State Adult Literacy Survey, 1992.

individuals in Level 2 worked an average of 27 to 31 weeks, those in Level 3 worked 37 to 39 weeks, adults in Level 4 worked 39 to 41 weeks, and individuals in Level 5 worked an average of 41 to 45 weeks. Thus, respondents performing in the highest literacy level worked, on average, roughly three times as many weeks as those in the lowest level.

Economic Status

To explore the relationships between literacy and economic status, the State and National Adult Literacy Surveys included a series of questions requesting detailed information about respondents' income. One of the questions asked for information on weekly wages, another asked about annual household income, and another asked about sources of income.

Section III 95

When adults' literacy proficiencies are compared according to their responses to these questions, strong relationships between literacy and economic status are evident. Individuals who earned low wages, had low annual household incomes, and who were either poor or near poor were far more likely than their more affluent peers to perform in the lowest literacy levels.

Weekly wages

Individuals who were working full time or part time or were on leave from their jobs the week before the survey were asked to report their weekly wage or salary before deductions. Given that individuals in professional, managerial, and technical positions were more likely to perform in the higher literacy levels, and that those in the higher literacy levels were likely to have worked more weeks in the past year than individuals in the lower levels, it is not surprising that weekly wages are also higher for adults with greater literacy proficiencies (Table 3.4).



IOWA TABLE 3.4

Median Weekly Wages, by Literacy Level: Results for Iowa, the Midwest, and the Nation

	LITERACY	Le 225 c	ovel 1 or lower	Le [*] 226	vel 2 to 275	Le ⁴ 276	vel 3 to 325	Le 326	vel 4 to 375	376 d	evel 5 or higher
		ww	(SE)	ww	(SE)	ww	(SE)	ww	(SE)	ww	(SE)
<u>Prose</u> lowa Midwest Nation		258 228 240	(42.5) (11.7) (2.2)	242 262 281	(20.6) (26.2) (4.8)	298 309 339	(24.2) (12.5) (16.9)	391 424 465	(42.7) (27.4) (19.0)	514 607 650	(24.2) (116.1) (61.5)
Document Iowa Midwest Nation		261 230 244	(73.8) (10.4) (5.2)	244 276 288	(27.0) (10.5) (8.9)	302 319 350	(5.2) (14.3) (0.6)	400 411 462	(69.5) (19.8) (28.7)	504 567 618	(36.7) (112.7) (34.6)
Quantitative Iowa Midwest Nation		251 220 230	(33.9) (19.2) (10.5)	228 251 274	(16.5) (2.2) (11.4)	283 306 345	(32.3) (14.6) (3.8)	419 427 472	(46.9) (49.6) (14.9)	550 621 681	(58.4) (69.6) (49.5)

WW = median weekly wages; (SE) = standard error of the estimate (the reported sample estimate can be said to be within 2 standard errors of the true population value with 95% confidence).

Source: Educational Testing Service, State Adult Literacy Survey, 1992.

96 Employment, Economic Status . . .

On each literacy scale, the median weekly earnings of Iowa residents who performed in Levels 1 and 2 were between \$228 and \$261. In comparison, those in Level 3 earned \$283 to \$302. Adults in Level 4 reported earning \$391 to \$419, and for Iowa residents who attained Level 5, the financial rewards were even greater. Individuals performing in this level on the prose scale, for example, had median earnings of \$514 each week — approximately twice as much as individuals who demonstrated skills in the Level 1 range on that scale.

Annual household income

Household survey respondents were asked to indicate their total family incomes from all sources in the year preceding the survey. They were instructed to consider as family anyone who lives with them and is related by blood, marriage, or adoption.

The pattern observed in the weekly wages data is repeated in the median annual household income data: Adults who performed in the highest literacy levels reported much larger annual household income than adults in the lowest levels (Table 3.5). On the document scale, for example, the median annual household income of Iowa residents who performed in the two highest proficiency levels was approximately \$40,000, compared with about \$30,000 for respondents who performed in Level 3 and \$12,000 for respondents in the lowest level.

These strong relationships between literacy and family income are also evident in the regional and national data, where the gap in median annual earnings between the highest and lowest proficiency level was between \$33,000 and \$40,000.

Sources of nonwage income

Household survey participants were given a list of various types of nonwage income and support and asked to identify each type that they or anyone in their families had received in the year preceding the survey. The skills of individuals who reported receiving three types of nonwage income and support that reflect socioeconomic status are examined here: Aid to Families with Dependent Children (AFDC), food stamps, and interest from savings or other bank accounts.



IOWA TABLE 3.5 Median Annual Household Income, by Literacy Level: Results for Iowa, the Midwest, and the Nation

	LITERACY LEVEL	Lev 225 or	rel 1 lower	Lev 226 t	el 2 o 275	Lev 276 t	el 3 o 325	Lev 326 (vel 4 to 375	Lev 376 or	el 5 higher
len li		н	(SE)	ні	(SE)	н	(SE)	н	(SE)	н	(SE)
Prose Iowa Midwest Nation		10,630 16,630 15,550	(5,700) (1,210) (1,650)	24,390 24,710 25,010	(3,360) (720) (300)	30,010 34,190 35,020	(70) (1,320) (300)	39,580 44,590 45,610	(960) (610) (1,330)	40,620 52,400 55,400	(3,440) (2,740) (7,120)
Document Iowa Midwest Nation		12,380 17,280 16,300	(10,880) (1,340) (1,850)	25,600 27,020 27,580	(3,890) (1,320) (610)	30,800 35,210 36,700	(890) (460) (1,560)	38,610 44,000 46,180	(3,850) (1,310) (3,020)	43,190 50,410 51,100	(8,310) (6,700) (1,250)
Quantitative Iowa Midwest Nation		12,320 15,630 15,180	(3,890) (1,670) (280)	25,270 26,090 25,820	(3,040) (2,630) (2,550)	29,380 33,020 35,010	(2,570) (4,170) (300)	37,400 43,490 44,980	(5,310) (3,510) (400)	45,010 50,410 53,910	(3,560) (970) (2,940)

HI = median annual household income; (SE) = standard error of the estimate (the reported sample estimate can be said to be within 2 standard errors of the true population value with 95% confidence).

Source: Educational Testing Service, State Adult Literacy Survey, 1992.

Five percent of the adults in Iowa reported having received AFDC or public assistance in the year before the survey (Table 3.6P,D,Q). Across the literacy scales, nearly half of these individuals demonstrated skills in the two lowest literacy levels; 15 to 20 percent performed in Level 1, and another 25 to 31 percent performed in Level 2. At the other end of the scale, 7 to 15 percent reached Level 4, and only 1 percent attained Level 5.

The pattern of results for food stamp recipients is similar. Eight percent of Iowa residents said they or someone in their family had received food stamps in the past year. On each scale, 22 to 25 percent of these adults performed in Level 1, and 29 to 32 percent performed in Level 2, while just 8 to 11 percent

reached the two highest literacy levels on each scale.

98.... Employment, Economic Status...



IOWA TABLE 3.6P

Prose Literacy Levels and Average Proficiencies, by Sources of Nonwage Income and Support: Results for Iowa, the Midwest, and the Nation

SOURCES OF NONWAGE					Pe	rcenta	ge of ac	ults in	each pro	nse lite	racy leve	nl		
			Level 225 or le	l 1 ower	Leve 226 to	1 2 275	Leve 276 to	al 3 325	Leve 326 to	el 4 375	Leve 376 or l	el 5 higher	Aver Profici	age ency
	n	WGT N (/1000)	RPCT (SE)	RPCT (SE)	RPCT (SE)	RPCT (SE)	RPCT (SE)	PROF (SE)
<u>lowa</u> AFDC, public assistance Food stamps Interest from savings	75 100 735	115 166 1,230	15 (22 (11 (5.0) 9.3) 5.3)	31 (32 (19 (9.2) 8.0) 2.7)	39 (35 (40 (7.4) 8.2) 6.9)	15 (11 (26 (5.9) 3.2) 2.4)	0†(0†(4 (0.5) 0.0) 1.0)	278 (264 (296 (7.1) 7.8) 5.5)
Midwest AFDC, public assistance Food stamps Interest from savings	621 837 3,578	3,150 4,283 23,329	24 (29 (12 (3.7) 3.7) 1.1)	40 (38 (21 (3.2) 3.7) 1.6)	28 (25 (37 (2.6) 2.4) 1.8)	8 (7 (25 (1.8) 1.9) 1.1)	1 (0†(5 (1.1) 0.8) 0.5)	259 (250 (294 (3.0) 2.5) 1.2)
<u>Nation</u> AFDC, public assistance Food stamps Interest from savings	2,070 3,001 10,884	11,995 17,953 88,365	34 (38 (11 (1.7) 1.6) 0.4)	36 (36 (21 (1.6) 1.4) 0.9)	24 (21 (36 (1.7) 1.4) 1.0)	6 (5 (26 (1.1) 0.9) 0.7)	0 [†] (0 [†] (6 (0.3) 0.4) 0.5)	243 (236 (297 (2.2) 1.8) 0.7)

n = sample size; WGT N = population size estimate / 1,000 (the sample sizes for subpopulations may not add up to the total sample sizes, due to missing data); RPCT = row percentage estimate; PROF = average proficiency estimate; (SE) = standard error of the estimate (the reported sample estimate can be said to be within 2 standard errors of the true population value with 95% confidence).

† Percentages less than 0.5 are rounded to zero.

*** Sample size is insufficient to permit a reliable estimate (fewer than 45 respondents).

Source: Educational Testing Service, State Adult Literacy Survey, 1992.

Fifty-nine percent of the adults in Iowa reported having received interest from savings in the past year. These individuals were generally less likely than AFDC or food stamp recipients to perform in the lowest levels on each scale and much more likely to attain the highest levels, though the large standard errors prevent some of the differences from being statistically significant.

Section III 99



IOWA TABLE 3.6D

Document Literacy Levels and Average Proficiencies, by Sources of Nonwage Income and Support: Results for Iowa, the Midwest, and the Nation

SOURCES OF NONWAGE					Per	centag	e of adu	lts in e	ach docu	ument	literacy le	evel		
			Lev 225 or	el 1 Iower	Lev 226 to	el 2 o 275	Lev 276 tr	el 3 o 325	Lev 326 to	el 4 o 375	Lev 376 or	el 5 higher	Aver Profic	age lency
	n	WGT N (/1000)	RPCT (SE)	RPCT (SE)	RPCT (SE)	RPCT (SE)	RPCT (SE)	PROF (SE)
Iowa AFDC, public assistance Food stamps Interest from savings	75 100 735	115 166 1,230	17 (25 (12 (6.8) 9.2) 4.0)	27 (31 (24 (8.8) 8.7) 2.2)	48 (36 (37 (8.3) 8.2) 3.8)	7 (8 (23 (4.4) 4.1) 2.3)	1 (0 [†] (3 (0.7) 0.4) 1.4)	274 (256 (290 (5.8) 10.3) 4.3)
Midwest AFDC, public assistance Food stamps Interest from savings	621 837 3,578	3,150 4,283 23,329	30 (35 (13 (3.2) 2.4) 1.0)	39 (35 (25 (2.8) 2.5) 1.5)	24 (23 (36 (2.7) 1.9) 1.9)	7 (7 (22 (2.3) 1.7) 1.4)	1 (0 [†] (4 (1.2) 0.8) 0.5)	253 (245 (287 (3.5) 3.2) 1.5)
<u>Nation</u> AFDC, public assistance Food stamps Interest from savings	2,070 3,001 10,884	11,995 17,953 88,365	37 (41 (13 (1.5) 1.4) 0.5)	35 (33 (24 (1.2) 1.4) 0.7)	23 (20 (35 (1.5) 1.1) 0.6)	5 (5 (23 (0.9) 0.6) 0.6)	0 [†] (0 [†] (5 (0.3) 0.3) 0.3)	239 (232 (289 (2.0) 1.9) 0.9)

n = sample size; WGT N = population size estimate / 1,000 (the sample sizes for subpopulations may not add up to the total sample sizes, due to missing data); RPCT = row percentage estimate; PROF = average proficiency estimate; (SE) = standard error of the estimate (the reported sample estimate can be said to be within 2 standard errors of the true population value with 95% confidence).

[†] Percentages less than 0.5 are rounded to zero.

*** Sample size is insufficient to permit a reliable estimate (fewer than 45 respondents).

Source: Educational Testing Service, State Adult Literacy Survey, 1992.

The average literacy proficiencies of adults who received interest from savings or another type of bank account in the past year were higher than those of residents who said they or someone in their families had received public assistance or food stamps. These differences are particularly evident in the national data. For instance, the average prose score of adults nationwide who had received AFDC or public assistance was 243, and the average score of food stamp recipients was 236, while for adults who had income from savings it was 297.





IOWA TABLE 3.6Q

Quantitative Literacy Levels and Average Proficiencies, by Sources of Nonwage Income and Support: Results for Iowa, the Midwest, and the Nation

SOURCES OF NONWAGE					Perc	entage	of adult	s in ea	ch quant	itative	literacy l	evel		
			Leve 225 or	el 1 lower	Leve 226 to	el 2 275	Leve 276 to	el 3 5 325	Leve 326 to	el 4 0 375	Leve 376 or	el 5 higher	Avera Profici	age ency
	n	WGT N (/1000)	RPCT (SE)	RPCT (SE)	RPCT (SE)	RPCT (SE)	RPCT (SE)	PROF (SE)
lowa									1					
AFDC, public assistance	75	115	20 (6.4)	25 (9.3)	47 (7.8)	8(4.0)	1(1.6)	273 (6.9)
Food stamps Interest from savings	100 735	166 1,230	25 (9.1) 5.5)	17 (9.3) 2.0)	37 (4.1)	28 (2.9)	6(1.2)	299 (6.6)
Midwest														
AFDC, public assistance	621	3,150	32 (3.5)	36 (4.5)	23 (3.9)	8 (1.5)	1(1.0)	251 (3.3)
Food stamps	837	4,283	36 (3.3)	36 (3.3)	21 (3.2)	7(1.7)	1(0.7)	242 (3.3)
Interest from savings	3,578	23,329	10 (1.0)	21 (1.3)	36 (1.5)	26 (1.4)	0(0.5)	298 (1.8)
Nation					1							12.12.0		
AFDC, public assistance	2,070	11,995	40 (1.7)	32 (1.4)	21 (2.0)	6(1.0)	1(0.4)	235 (2.3)
Food stamps	3,001	17,953	44 (1.5)	32 (1.4)	20 (1.4)	5(0.7)	1(0.4)	228 (1.9)
Interest from savings	10,884	88,365	11 (0.5)	20 (0.7)	30 (0.7)	21(0.0)	1	0.4)	230 (0.9)

n = sample size; WGT N = population size estimate / 1,000 (the sample sizes for subpopulations may not add up to the total sample sizes, due to missing data); RPCT = row percentage estimate; PROF = average proficiency estimate; (SE) = standard error of the estimate (the reported sample estimate can be said to be within 2 standard errors of the true population value with 95% confidence).

Percentages less than 0.5 are rounded to zero.

*** Sample size is insufficient to permit a reliable estimate (fewer than 45 respondents).

Source: Educational Testing Service, State Adult Literacy Survey, 1992.

Poverty status

Adults who participated in the household component of the National and State Adult Literacy Surveys were divided into two categories — poor or near poor, and not poor — based on both their household income and family size. (The criteria are provided in the appendices.) For example, adults whose household size is one and whose annual household income is at or below \$8,665 are classified as poor or near poor. For adults in a four-person family, those whose annual household income is \$17,405 or less are assigned to that category. Nineteen percent of the adults in Iowa and nationwide were designated as poor or near poor (Table 3.7P,D,Q). Across the three literacy scales, approximately half of the Iowa residents who were classified as poor or near poor performed in the two lowest levels; 22 to 27 percent performed in Level 1, and another 27 to 29 percent performed in Level 2. In comparison, 25 to 31 percent of the adults classified as not poor were in the two lowest levels.



IOWA TABLE 3.7P

Prose Literacy Levels and Average Proficiencies, by Poverty Status: Results for Iowa, the Midwest, and the Nation

POVERTY STATUS						F	Percent	age of a	dults in	each pr	ose lite	eracy lev	el		
				Leve 225 or	el 1 Iower	Lev 226 to	el 2 o 275	Lev 276 t	vel 3 o 325	Lev 326 to	el 4 o 375	Lev 376 or	el 5 higher	Aver Profic	rage lency
1	ŭ	WGT N (/1000)	PCT	RPCT (SE)	RPCT (SE)	RPCT (SE)	RPCT (SE)	RPCT (SE)	PROF (SE)
lowa															
Not poor	846	1,258	81	5(1.1)	20 (2.6)	43 (4.4)	27 (2.6)	4 (1.3)	302 (3.1)
Midwest	1/0	232	15	24 (0.3)	29 (0.4)	33 (5.6)	13 (3.2)	2(1.0)	268 (6.6)
Not poor	4,516	28,196	83	10 (0.8)	25 (1.3)	38 (1.3)	22 (1.1)	4(0.5)	292 (1.6)
Poor or near poor	1,040	5,814	17	30 (2.7)	34 (3.5)	27 (2.0)	9 (1.4)	1(0.6)	252 (3.3)
Nation										1					
Not poor	14,868	113,929	81	12 (0.4)	24 (0.7)	37 (1.0)	23 (0.5)	5(0.3)	290 (0.7)
Poor or near poor	3,968	26,353	19	38 (1.3)	31 (1.3)	22 (0.8)	8 (0.9)	1(0.3)	239 (2.2)

n = sample size; WGT N = population size estimate / 1,000 (the sample sizes for subpopulations may not add up to the total sample sizes, due to missing data); PCT = percentage in group; RPCT = row percentage estimate; PROF = average proficiency estimate; (SE) = standard error of the estimate (the reported sample estimate can be said to be within 2 standard errors of the true population value with 95% confidence).

Percentages less than 0.5 are rounded to zero.

*** Sample size is insufficient to permit a reliable estimate (fewer than 45 respondents).

Source: Educational Testing Service, State Adult Literacy Survey, 1992.



As a result of these differences in the distribution of performance, the average prose, document, and quantitative proficiency scores of Iowa adults who were poor or near poor are more than 30 points lower than the scores of adults who were not poor. These results underscore literacy's strong connection to economic status.



IOWA TABLE 3.7D

Document Literacy Levels and Average Proficiencies, by Poverty Status: Results for Iowa, the Midwest, and the Nation

POVERTY STATUS					Percentag	e of adults in ea	ich document l	iteracy level	
				Level 1 225 or lower	Level 2 226 to 275	Level 3 276 to 325	Level 4 326 to 375	Level 5 376 or higher	Average Proficiency
	n	WGT N (/1000)	PCT	RPCT (SE)	RPCT (SE)	RPCT (SE)	RPCT (SE)	RPCT (SE)	PROF (SE)
Iowa Not poor Poor or near poor	846 176	1,258 292	81 19	7 (1.5) 27 (8.9)	24 (3.0) 29 (5.9)	41 (2.4) 32 (5.8)	24 (2.7) 11 (3.9)	3 (1.5) 1 (1.5)	296 (2.8) 264 (5.9)
Midwest Not poor Poor or near poor	4,516 1,040	28,196 5,814	83 17	13 (1.0) 35 (3.3)	29 (1.2) 32 (3.7)	36 (1.5) 24 (2.6)	20 (1.2) 9 (2.1)	3 (0.5) 1 (0.8)	285 (1.6) 246 (4.4)
Not poor Poor or near poor	14,868 3,968	113,929 26,353	81 19	14 (0.5) 42 (1.5)	27 (0.6) 29 (1.4)	35 (0.6) 21 (1.2)	20 (0.5) 8 (0.9)	4 (0.3) 1 (0.3)	284 (0.8) 234 (2.3)

- n = sample size; WGT N = population size estimate / 1,000 (the sample sizes for subpopulations may not add up to the total sample sizes, due to missing data); PCT = percentage in group; RPCT = row percentage estimate; PROF = average proficiency estimate; (SE) = standard error of the estimate (the reported sample estimate can be said to be within 2 standard errors of the true population value with 95% confidence).
- † Percentages less than 0.5 are rounded to zero.
- *** Sample size is insufficient to permit a reliable estimate (fewer than 45 respondents).

Source: Educational Testing Service, State Adult Literacy Survey, 1992.



IOWA TABLE 3.7Q

Quantitative Literacy Levels and Average Proficiencies, by Poverty Status: Results for Iowa, the Midwest, and the Nation

POVERTY STATUS	-					Perc	centage	e of adul	ts in ea	ich quan	titative	literacy I	evel		
				Lev 225 or	el 1 Iower	Lev 226 to	el 2 o 275	Lev 276 to	el 3 o 325	Lev 326 to	el 4 5 375	Lev 376 or	el 5 higher	Aver Profic	age lency
	n	WGT N (/1000)	PCT	RPCT (SE)	RPCT (SE)	RPCT (SE)	RPCT (SE)	RPCT (SE)	PROF (SE)
lowa															
Not poor	846	1,258	81	7(2.6)	18 (2.6)	40 (3.6)	30 (2.6)	6(1.9)	305 (2.7)
Poor or near poor	176	292	19	22 (5.2)	27 (9.6)	34 (6.5)	15 (4.9)	2(1.5)	269 (6.7)
Midwest								1							
Not poor	4,516	28,196	83	10 (0.9)	24 (1.9)	37 (1.5)	23 (1.5)	5(0.5)	294 (1.9)
Poor or near poor	1,040	5,814	17	34 (3.0)	31 (3.1)	25 (2.9)	9(2.0)	11	0.7)	246 (4.6)
Nation												1			
Not poor	14,868	113,929	81	12 (0.4)	23 (0.8)	36 (0.8)	23 (0.5)	6(0.3)	291 (0.7)
Poor or near poor	3,968	26,353	19	42 (1.5)	28 (1.5)	21 (1.2)	81	1.0)	11	0.4)	233 (2.4)

n = sample size; WGT N = population size estimate / 1,000 (the sample sizes for subpopulations may not add up to the total sample sizes, due to missing data); PCT = percentage in group; RPCT = row percentage estimate; PROF = average proficiency estimate; (SE) = standard error of the estimate (the reported sample estimate can be said to be within 2 standard errors of the true population value with 95% confidence).

† Percentages less than 0.5 are rounded to zero.

*** Sample size is insufficient to permit a reliable estimate (fewer than 45 respondents).

Source: Educational Testing Service, State Adult Literacy Survey, 1992.

Civic Responsibility

Another area of interest in the survey was that of civic responsibility. Accordingly, one of the background questions asked household survey participants whether or not they had voted in a state or national election in the past five years. Their answers make it possible to investigate the connection between civic responsibilities and demonstrated literacy proficiencies. Are voters more likely than nonvoters to display advanced literacy skills, or are the proficiencies of the two groups essentially the same? The answers to these questions vary according to the population examined, as seen in the results

discussed below.

104 Employment, Economic Status . . .

Voting

Two-thirds of the adults in Iowa who were eligible to vote said they had voted in a recent election, compared with 72 percent of the eligible voters in the Midwest and 67 percent of those nationwide (Table 3.8P,D,Q). Thus, it appears that adults in Iowa are as likely to vote as adults nationwide, and are less likely to vote than adults in the Midwest.



IOWA TABLE 3.8P

Prose Literacy Levels and Average Proficiencies, by Voting in Recent Elections: Results for Iowa, the Midwest, and the Nation

VOTED IN					Percent	age of adults in	each prose litera	acy level	
FIVE YEARS	FIVE YEARS			Level 1 225 or lower	Level 2 226 to 275	Level 3 276 to 325	Level 4 326 to 375	Level 5 376 or higher	Average Proficiency
	n	WGT N (/1000)	PCT	RPCT (SE)	RPCT (SE)	RPCT (SE)	RPCT (SE)	RPCT (SE)	PROF (SE)
<u>lowa</u> Yes No	824 372	1,346 661	67 33	13 (2.8) 18 (5.5)	20 (2.8) 31 (3.5)	38 (5.5) 36 (5.3)	25 (2.2) 13 (2.3)	4 (1.0) 1 (0.6)	292 (3.1) 272 (4.2)
<u>Midwest</u> Yes No	4,847 1,962	30,461 11,854	72 28	15 (1.0) 20 (1.5)	25 (1.4) 35 (1.8)	35 (1.3) 34 (2.0)	21 (1.0) 11 (1.2)	4 (0.4) 1 (0.4)	286 (1.5) 267 (1.4)
Nation Yes No	15,484 7,616	117,379 58,510	67 33	16 (0.4) 26 (0.6)	24 (0.7) 32 (1.0)	34 (0.8) 30 (1.0)	22 (0.6) 11 (0.6)	5 (0.3) 1 (0.2)	285 (0.7) 257 (1.0)

n = sample size; WGT N = population size estimate / 1,000 (the sample sizes for subpopulations may not add up to the total sample sizes, due to missing data); PCT = percentage in group; RPCT = row percentage estimate; PROF = average proficiency estimate; (SE) = standard error of the estimate (the reported sample estimate can be said to be within 2 standard errors of the true population value with 95% confidence).

† Percentages less than 0.5 are rounded to zero.

*** Sample size is insufficient to permit a reliable estimate (fewer than 45 respondents).

Source: Educational Testing Service, State Adult Literacy Survey, 1992.

Section III 105



IOWA TABLE 3.8D

Document Literacy Levels and Average Proficiencies, by Voting in Recent Elections: Results for Iowa, the Midwest, and the Nation

VOTED IN THE PAST					Percenta	ge of adults in e	àch document li	teracy level	
FIVE YEARS				Level 1 225 or lower	Level 2 226 to 275	Level 3 276 to 325	Level 4 326 to 375	Level 5 376 or higher	Average Proficiency
	n	WGT N (/1000)	PGT	RPCT (SE)	RPCT (SE)	RPCT (SE)	RPCT (SE)	RPCT (SE)	PROF (SE)
lowa	-								
Yes	824	1,346	67	15 (2.8)	25 (2.2)	36 (3.2)	21 (2.3)	3 (1.2)	285 (2.5)
No	372	661	33	20 (4.2)	31 (2.9)	35 (4.0)	14 (3.1)	1 (0.6)	270 (5.0)
Midwest									
Yes	4,847	30,461	72	18 (1.1)	28 (1.5)	33 (1.7)	18 (1.2)	3 (0.4)	278 (1.7)
No	1,962	11,854	28	22 (1.3)	34 (2.0)	32 (1.7)	12 (0.9)	1 (0.4)	265 (1.5)
Nation									
Yes	15,484	117,379	67	19 (0.5)	27 (0.6)	32 (0.7)	19 (0.5)	4 (0.2)	277 (0.8)
No	7,616	58,510	33	27 (0.6)	31 (0.7)	30 (0.7)	10 (0.5)	1 (0.2)	255 (1.0)

n = sample size; WGT N = population size estimate / 1,000 (the sample sizes for subpopulations may not add up to the total sample sizes, due to missing data); PCT = percentage in group; RPCT = row percentage estimate; PROF = average proficiency estimate; (SE) = standard error of the estimate (the reported sample estimate can be said to be within 2 standard errors of the true population value with 95% confidence).

† Percentages less than 0.5 are rounded to zero.

*** Sample size is insufficient to permit a reliable estimate (fewer than 45 respondents).

Source: Educational Testing Service, State Adult Literacy Survey, 1992.

Literacy skills do appear to be related to voting practices. On all three literacy scales, Iowa residents who had voted in a recent state or national election demonstrated higher proficiencies, on average, than those who had not. There were also significant differences between the literacy skills of voters and nonvoters in the Midwest and the nation. On each of the literacy scales, voters were less likely than nonvoters to demonstrate skills in Levels 1 and 2 and more likely to attain Levels 4 and 5. As a result, the average prose, document, and quantitative proficiencies of voters in the Midwest and the

nation as a whole were significantly higher than those of nonvoters.

106 Employment, Economic Status . . .

IOWA TABLE 3.8Q

Quantitative Literacy Levels and Average Proficiencies, by Voting in Recent Elections: Results for Iowa, the Midwest, and the Nation

VOTED IN					Percentage	e of adults in eac	ch quantitative li	teracy level	
FIVE YEARS	FIVE YEARS				Level 2 226 to 275	Level 3 276 to 325	Level 4 326 to 375	Level 5 Average 376 or higher Proficience	
	'n	WGT N (/1000)	PCT	RPCT (SE)	PROF (SE)				
<u>lowa</u> Yes No	824 372	1,346 661	67 33	13 (4.3) 19 (2.2)	19 (2.0) 28 (5.0)	35 (3.2) 36 (4.9)	27 (2.1) 15 (2.7)	5 (1.2) 2 (0.9)	294 (3.9) 274 (4.0)
Midwest Yes No	4,847 1,962	30,461 11,854	72 28	16 (1.0) 21 (1.9)	23 (1.5) 33 (3.2)	34 (1.2) 32 (2.6)	22 (1.3) 12 (1.2)	5 (0.5) 1 (0.5)	286 (1.9) 266 (2.1)
Nation Yes No	15,484 7,616	117,379 58,510	67 33	17 (0.5) 28 (0.8)	23 (0.6) 30 (0.9)	33 (0.7) 29 (0.8)	22 (0.5) 11 (0.4)	6 (0.3) 1 (0.3)	284 (1.0) 255 (1.1)

n = sample size; WGT N = population size estimate / 1,000 (the sample sizes for subpopulations may not add up to the total sample sizes, due to missing data); PCT = percentage in group; RPCT = row percentage estimate; PROF = average proficiency estimate; (SE) = standard error of the estimate (the reported sample estimate can be said to be within 2 standard errors of the true population value with 95% confidence).

† Percentages less than 0.5 are rounded to zero.

*** Sample size is insufficient to permit a reliable estimate (fewer than 45 respondents).

Source: Educational Testing Service, State Adult Literacy Survey, 1992.

Summary

Approximately half of the adults in Iowa reported that they were employed full time, and another 14 percent said they were employed part time. Five percent were unemployed, laid off, or looking for work, and nearly one-third were in school, keeping house, retired, or doing volunteer work — that is, out of the labor force. In each of the three dimensions of literacy, full-time and part-time employees performed similarly. The average proficiencies of employed adults differed sharply from those of adults who were either unemployed or out of the labor force, however. Employed adults were much more likely to reach the highest literacy levels and much less likely to perform in the lowest levels.

Section III 107

Twenty percent of the adults in Iowa said they worked in managerial, professional, or technical jobs; 23 percent were in sales or clerical occupations; 29 percent worked in craft or service occupations; and 28 percent were in labor, assembly, fishing, or farming jobs. Although some individuals in managerial and professional jobs displayed limited literacy skills, they were less likely than respondents in other types of jobs to perform in the lowest literacy levels and more likely to attain the highest levels.

On each literacy scale, adults who performed in Levels 3, 4, and 5 worked more weeks in the past year than those in Level 2, who worked more weeks than those in Level 1. The average number of weeks worked climbs steadily across the literacy levels. Adults with higher literacy proficiencies were also likely to earn higher weekly wages than adults with more limited skills. Similarly, individuals who performed in the highest literacy levels reported much higher annual household incomes, on average, than adults in the lowest levels.

Adults in Iowa whose families had received AFDC, public assistance, or food stamps in the past year demonstrated lower average proficiencies than adults who reported having received interest from savings.

Individuals who were classified as poor or near poor as a result of their income and household size were much more likely than those who were not poor to demonstrate limited literacy skills. Across the three literacy scales, approximately half of the Iowa residents classified as poor or near poor had proficiencies in the two lowest levels. As a result, their average literacy scores were considerably lower than those of individuals who were not poor.

Respondents in Iowa were as likely as respondents nationwide but less likely than those in the Midwest to have voted in a recent state or national election. Two-thirds of the eligible voters in Iowa said they had voted recently. Literacy appears to be related to voting in Iowa as well as in the Midwest and nation, as voters demonstrated higher average literacy proficiencies than nonvoters.

108.... Employment, Economic Status ...











SECTION IV

Language Use and Literacy Practices

revious studies have identified certain practices and conditions that are related to literacy. Accordingly, the State and National Adult Literacy Surveys included an extensive set of questions that asked respondents about their early language experiences and the frequency with which they engage in various reading and writing practices. This section of the report examines responses to these questions and their relationship to demonstrated literacy proficiencies.

Language Use

One area of primary interest in the survey was that of language use. What proportions of the adults in Iowa and nationwide were bilingual or spoke a language other than English as children? What languages do respondents speak now, in various contexts? Do adults who demonstrate limited skills in the English language perceive themselves as having limited proficiency? These and other questions are explored in the beginning of this section.



Language learned before starting school

Survey participants were asked what language or languages they learned to speak before they started school, and their responses were analyzed to determine the percentages of adults who spoke English only, who spoke another language only, and who spoke English and another language.

Virtually all (96 percent) of the respondents in Iowa said they spoke only English before beginning their schooling. Two percent said they spoke a different language only, and another 2 percent spoke both English and another language as a child (Table 4.1).

Nationwide, 85 percent of the respondents reported speaking English only, 10 percent said they spoke a different language only, and 5 percent were bilingual before they began their schooling. Thus, the proportions of adults in

Section IV 111



IOWA TABLE 4.1

Average Literacy Proficiencies, by Languages Learned Before Starting School: Results for Iowa, the Midwest, and the Nation

LANGUAGES LEARNED BEFORE STARTING SCHOOL			[Average proficie	ency of adults on e	ach literacy scale
	1		[Prose	Document	Quantitative
	n	WGT N (/1000)	PCT	PROF (SE)	PROF (SE)	PROF (SE)
lowa						
English only	1,194	2,011	96	286 (3.0)	280 (2.9)	288 (3.7)
Spanish/Other only	22	43	2	*** (****)	*** (****)	*** (****)
English and Spanish/Other	29	40	2	*** (****)	*** (****)	*** (****)
Midwest	A Real Property lies					
English only	6,917	41,386	91	283 (1.0)	277 (1.2)	283 (1.4)
Spanish/Other only	331	2,034	4	215 (7.0)	218 (7.0)	222 (7.4)
English and Spanish/Other	234	1,839	4	264 (11.9)	254 (11.1)	260 (14.2)
Nation						
English only	21,980	162,016	85	282 (0.7)	275 (0.8)	280 (0.8)
Spanish/Other only	2,794	19,569	10	200 (2.1)	200 (2.3)	204 (2.4)
English and Spanish/Other	1,271	9,408	5	264 (3.5)	257 (3.4)	260 (4.2)

n = sample size; WGT N = population size estimate / 1,000 (the sample sizes for subpopulations may not add up to the total sample sizes, due to missing data); PCT = percentage in group; PROF = average proficiency estimate; (SE) = standard error of the estimate (the reported sample estimate can be said to be within 2 standard errors of the true population value with 95% confidence).

Percentages less than 0.5 are rounded to zero.

*** Sample size is insufficient to permit a reliable estimate (fewer than 45 respondents).

Interpret with caution -- the nature of the sample does not allow accurate determination of the variability of this statistic.

Source: Educational Testing Service, State Adult Literacy Survey, 1992.

Iowa who as children spoke another language in place of or in addition to English were far lower than the proportions in the United States population as a whole.

Adults who reported speaking only English before they began school tended to display better English literacy skills than adults who were bilingual as children. Because the number of adults in Iowa who spoke another language before starting school is so small, reliable proficiency estimates are not

112 Language Use and Literacy Practices

÷

available. Nationwide, however, the gap in average scores between Englishonly and bilingual individuals was roughly 20 points on each literacy scale. Individuals who spoke only English as children performed, on average, toward the low end of the Level 3 range, while those who spoke English and another language tended to perform in the Level 2 range.

Adults who were bilingual as children, in turn, performed far better than adults who spoke only a different language before starting school. Here, the gap was larger than between English-only and bilingual respondents. On the prose scale, for example, adults nationwide who spoke English and another language as children had an average score of 264; in contrast, those who spoke only a different language had an average score of just 200 - within the range for Level 1. Similarly, on the document and quantitative scales, the difference in average scores between bilingual individuals and those who spoke only a different language was approximately 60 points.

Language usually spoken now

Respondents who said they learned a language other than English before starting school were asked what language they usually speak now. In Iowa, nearly all of these individuals (92 percent) said they usually speak English now, while 4 percent said they usually speak Spanish (Table 4.2). Another 4 percent reported that they usually speak a language other than English or Spanish.

Nationwide, 65 percent of the respondents who spoke another language before starting school reported they usually speak English now, while 27 percent said they usually speak Spanish and 8 percent said they usually speak some other language.

In the national and regional samples alike, adults who learned another language as children but now usually speak English performed far better than those who usually speak the other language. Nationwide, for example, there is a gap of approximately 100 points between the average proficiencies of those who usually speak English and those who usually speak Spanish. Stated differently, adults who usually speak Spanish had average scores in the Level 1 range on each scale, while those who usually speak English performed in the Level 2 range.

Further, adults who said they usually speak a language other than English or Spanish generally performed better than adults who usually speak Spanish, but worse than adults who usually speak English. On the document scale, for example, adults nationwide who reported usually speaking a language other than Spanish or English had an average proficiency score of 187 - 35 points higher than Spanish speakers but 63 points lower than English speakers.

Section IV 113



IOWA TABLE 4.2

Average Literacy Proficiencies of Adults Who Learned a Non-English Language, By Language Spoken Now: Results for Iowa, the Midwest, and the Nation

LANGUAGE USUALLY				Average prof	ficiency of adults on eacl	h literacy scale
SPOREN NOW				Prose	Document	Quantitative
	n	WGT N (/1000)	PCT	PROF (SE)	PROF (SE)	PROF (SE)
lowa						
English	45	75	92	285 (7.1)	281 (9.3)	291 (11.6)
Spanish	2	3	4	*** (****)	*** (****)	*** (****)
Other	3	3	4	*** (****)	*** (****)	*** (****)
Midwest						
English	394	3,120	83	251 (7.5)	245 (7.6)	250 (9.4)
Spanish	133	522	14	171 (12.3)	179 (9.9)	179 (9.9)
Other	20	130	3	*** (****)	*** (****)	*** (****)
Nation						
English	2,456	18,404	65	254 (2.2)	250 (2.4)	254 (2.7)
Spanish	1,311	7,634	27	153 (3.8)	152 (3.8)	150 (3.9)
Other	226	2,385	8	175 (6.8)	187 (5.8)	195 (8.3)

n = sample size; WGT N = population size estimate / 1,000 (the sample sizes for subpopulations may not add up to the total sample sizes, due to missing data); PCT = percentage in group; PROF = average proficiency estimate; (SE) = standard error of the estimate (the reported sample estimate can be said to be within 2 standard errors of the true population value with 95% confidence).

† Percentages less than 0.5 are rounded to zero.

*** Sample size is insufficient to permit a reliable estimate (fewer than 45 respondents).

I Interpret with caution -- the nature of the sample does not allow accurate determination of the variability of this statistic.

Source: Educational Testing Service, State Adult Literacy Survey, 1992.

Language use in various contexts

Survey respondents who said they learned a language other than English before starting school were also asked how often they use English or another language in various contexts. Seventy percent of the Iowa residents who spoke another language as a child said they always use English at home, while the remaining 30 percent said they sometimes speak their other language. Virtually none of the respondents said they always speak their other language (Table 4.3). Nearly all (97 percent) of the Iowa residents who learned another language before starting school said they always speak English while shopping

114 Language Use and Literacy Practices

<u>IOWA</u>

IOWA TABLE 4.3

Use of English or Another Language in Various Contexts by Adults Who Learned a Non-English Language: Results for Iowa

CONTEXT			Percentage of ad langua	dults who use Er age in various co	nglish or another ntexts
			Always English	Sometimes non-English language	Always non-English language
	n	WGT N (/1000)	RPCT (SE)	RPCT (SE)	RPCT (SE)
At home At work While shopping in own neighborhood When visiting relatives or friends	50 48 50 50	81 76 81 81	70 (6.8) 89 (8.0) 97 (2.9) 60 (7.3)	30 (6.8) 11 (8.0) 3 (2.9) 34 (7.5)	0 [†] (0.0) 0 [†] (0.0) 0 [†] (0.0) 7 (5.6)

n = sample size; WGT N = population size estimate / 1,000 (the sample sizes for subpopulations may not add up to the total sample sizes, due to missing data); RPCT = row percentage estimate; (SE) = standard error of the estimate (the reported sample estimate can be said to be within 2 standard errors of the true population value with 95% confidence).

† Percentages less than 0.5 are rounded to zero.

*** Sample size is insufficient to permit a reliable estimate (fewer than 45 respondents).

Source: Educational Testing Service, State Adult Literacy Survey, 1992.

in their neighborhoods, and 89 percent said they always use English at work. In contrast, a smaller proportion of these respondents (60 percent) said they always use English when visiting relatives or friends.

In sum, Iowa residents who learned a non-English language as a child are more likely to use only English at work and when shopping in their neighborhood than at home or when visiting relatives or friends.

Self-reported proficiency in the English language

One question of interest in this survey is that of self-perception. Do adults who display more limited skills in the English language perceive themselves as having restricted skills? To address this question, respondents were asked how well they understand the English language when it is spoken to them, and how well they speak, read, and write English.

Section IV 115

Virtually all adults in Iowa described themselves as understanding (99 percent), speaking (99 percent), reading (98 percent), and writing (96 percent) English either well or very well (Table 4.4).

Because so few Iowa residents described themselves as not understanding, speaking, or reading English well, reliable proficiency estimates are not available for this group. In examining the proficiency results by responses to the question about writing skills, however, we see that those who described themselves as not writing well (or at all) had average prose, document, and



IOWA TABLE 4.4

Average Literacy Proficiencies, by Self-reported English Literacy: **Results for Iowa**

SELF-REPORTED ENGLISH LITERACY			L	Average profic	iency of adults on ea	ch literacy scale
				Prose	Document	Quantitative
	n	WGT N (/1000)	PCT	PROF (SE)	PROF (SE)	PROF (SE)
Understand						
Very well or well	1,239	2,080	99	286 (2.9)	281 (2.7)	288 (3.3)
Not well or not at all	7	15	1	*** (****)	*** (****)	*** (****)
Speak						
Very well or well	1,236	2,076	99	286 (3.0)	281 (2.7)	288 (3.4)
Not well or not at all	10	19	1	*** (****)	*** (****)	*** (****)
Read						1 1 1
Very well or well	1,224	2.045	98	288 (2.4)	282 (2.1)	290 (2.6)
Not well or not at all	20	40	2	*** (****)	*** (****)	*** (****)
Write			100			
Very well or well	1,195	1,991	96	288 (2.5)	283 (2.3)	291 (2.9)
Not well or not at all	47	82	4	228 (14.6)!	223 (15.0)!	226 (16.2)!

- n = sample size; WGT N = population size estimate / 1,000 (the sample sizes for subpopulations may not add up to the total sample sizes, due to missing data); PCT = percentage in group; PROF = average proficiency estimate; (SE) = standard error of the estimate (the reported sample estimate can be said to be within 2 standard errors of the true population value with 95% confidence).
- Percentages less than 0.5 are rounded to zero. ***

t

- Sample size is insufficient to permit a reliable estimate (fewer than 45 respondents).
- Interpret with caution -- the nature of the sample does not allow accurate determination of the variability of this statistic. L

Source: Educational Testing Service, State Adult Literacy Survey, 1992.

116 Language Use and Literacy Practices

quantitative proficiency scores that were about 60 points lower than the scores of those who said they do.

Stated differently, Iowa residents who said they did not write English well tended to perform in the high end of the Level 1 range or the low end of the Level 2 range on each literacy scale, while residents who said they write well or very well performed, on average, in the Level 3 range.

It is intriguing to note that such large numbers of adults described themselves as having strong literacy skills, particularly when so many individuals performed in the lowest levels of literacy defined in this survey. In Iowa, for example, almost all adults described themselves as reading and writing English well or very well — yet 37 to 43 percent performed in Levels 1 and 2 on each literacy scale.

It therefore appears that many adults who displayed limited proficiencies in this assessment perceive that they have adequate literacy skills in English. It may be that their skills do, in fact, enable them to meet some or all of the literacy demands they encounter at work, at home, and in the community.

Literacy Practices

Previous studies have found strong connections between adults' literacy skills and their reading and writing practices — for example, the frequency with which they read the newspaper and other materials.¹ In this survey, similar connections are found, and these are discussed in the pages that follow. While reviewing the results, readers should keep in mind that the relationship between literacy skills and practices is complex. While it may be true that individuals with better skills are more likely to pursue an array of literacy activities, the experience of pursuing these activities is, in turn, also likely to strengthen their skills.

Reliance on print and nonprint sources of information

Survey participants were asked to indicate how much information about current events, public affairs, and government they get from different sources, such as newspapers, magazines, television, radio, and family or friends. For analysis purposes, these sources were grouped into three categories: print media (newspapers or magazines), nonprint media (television or radio), and personal sources (family or friends).

¹ I.S. Kirsch and A. Jungeblut. (1986). Literacy: Profiles of America's Young Adults. Princeton, NJ: Educational Testing Service.

Section IV 117

Most adults in Iowa (97 percent) said they get either some or a lot of information about current events, public affairs, and government from nonprint media, while 86 percent said they get much of their information from print media (Table 4.5). Approximately two-thirds (67 percent) reported getting some or a lot of information from personal sources, such as family or friends.



IOWA TABLE 4.5

Average Literacy Proficiencies, by Reliance on Various Sources of Information About Current Events: Results for Iowa

ARIOUS SOURCES			L	Average profi	ciency of adults on eac	ch literacy scale
ABOUT CURRENT EVENTS				Prose	Document	Quantitative
	n	WGT N (/1000)	PCT	PROF (SE)	PROF (SE)	PROF (SE)
Print media						
A lot or some	1,097	1,799	86	290 (3.0)	284 (31)	292 (37)
A little or none	149	296	14	256 (5.9)	254 (6.9)	258 (6.4)
Nonprint media						
A lot or some	1,208	2.037	97	286 (3 2)	280 (3.0)	288 (3 7)
A little or none	38	58	3	*** (****)	*** (****)	*** (****)
Personal sources			20			
A lot or some	830	1,406	67	287 (4 7)	282 (5 1)	297 (6 7)
A little or none	416	689	33	282 (3.3)	277 (4.8)	288 (5.8)

n = sample size; WGT N = population size estimate / 1,000 (the sample sizes for subpopulations may not add up to the total sample sizes, due to missing data); PCT = percentage in group; PROF = average proficiency estimate; (SE) = standard error of the estimate (the reported sample estimate can be said to be within 2 standard errors of the true population value with 95% confidence).

† Percentages less than 0.5 are rounded to zero.

*** Sample size is insufficient to permit a reliable estimate (fewer than 45 respondents).

Interpret with caution -- the nature of the sample does not allow accurate determination of the variability of this statistic.

Source: Educational Testing Service, State Adult Literacy Survey, 1992.

118 Language Use and Literacy Practices

Because virtually all adults rely on television and radio for information about current events, reliable proficiency estimates are unavailable for those who do not get a lot of information from these sources.

In comparing the proficiencies of adults in the other groups, however, one sees that literacy skills are not a good predictor of respondents' reliance on family or friends for information. On each literacy scale, the average scores of adults who said they get some or a lot of information about current events, public affairs, or government from personal sources were equivalent to those of respondents who said they get little or no information from these sources.

On the other hand, Iowa residents who said they get some or a lot of information from print media had considerably higher proficiencies (284 to 292 on each scale) than those who get little or no information from newspapers or magazines (254 to 258). On the prose and quantitative scales the difference between the two groups is 34 points, and on the document scale it is 30 points.

Frequency of newspaper reading

Many different types of newspapers are published in this country, ranging from long, comprehensive daily papers to shorter and less frequent community papers. Together these print media keep readers informed about current events in their communities, the nation, and the world. Because the newspaper plays such an important role in disseminating information in this society, adults who participated in the National and State Adult Literacy Surveys were asked to indicate how often they read one.

The responses indicate that newspaper reading is quite common (Table 4.6). Slightly more than half (56 percent) of the adults in Iowa said they read the newspaper every day, and another 24 percent said they read it a few times a week. Eleven percent said they read the paper once a week, 5 percent reported reading it less than once a week, and 4 percent said they never read a newspaper.

Though the literacy proficiencies of adults who said they rarely read a newspaper appear to be lower than the proficiencies of those who often read one, these differences are not statistically significant. In other words, the literacy skills of individuals who read the newspaper regularly and those who do not appear to be comparable, on average.

Aspects of newspaper reading

Survey participants were asked to indicate not only how often they read a newspaper, but also what parts they generally read. Together, the responses to these two questions were used to determine the percentages of newspaper readers (that is, of those who read a newspaper at least once a week) who read

Section IV 119



IOWA TABLE 4.6

Average Literacy Proficiencies, by Frequency of Newspaper Reading: Results for Iowa

FREQUENCY OF NEWSPAPER READING				Average p	proficiency of adu literacy scale	ults on each
				Prose	Document	Quantitative
	n	WGT N .(/1000)	PCT	PROF (SE)	PROF (SE)	PROF (SE)
Frequency of newspaper reading						
Every day	707	1,172	56	294 (2.7)	285 (2.6)	296 (3.6)
A few times a week	305	505	24	284 (5.8)	282 (7.3)	288 (7.6)
Once a week	142	238	11	276 (8.4)	279 (7.6)	279 (8.3)
Less than once a week	63	105	5	273 (11.1)!	269 (12.4)!	276 (11.9)!
Never	29	76	4	*** (****)	*** (****)	*** (****)

n = sample size; WGT N = population size estimate / 1,000 (the sample sizes for subpopulations may not add up to the total sample sizes, due to missing data); PCT = percentage in group; PROF = average proficiency estimate; (SE) = standard error of the estimate (the reported sample estimate can be said to be within 2 standard errors of the true population value with 95% confidence).

† Percentages less than 0.5 are rounded to zero.

*** Sample size is insufficient to permit a reliable estimate (fewer than 45 respondents).

Interpret with caution -- the nature of the sample does not allow accurate determination of the variability of this statistic. Source: Educational Testing Service, State Adult Literacy Survey, 1992.

> certain parts. The 10 categories listed in the survey questionnaire were grouped into five categories for reporting purposes: the news, editorials, and financial pages; home, health, fashion, and reviews of books, movies, and art; television, movie, and concert listings, as well as classified ads and other advertisements; comics, horoscopes, and advice columns; and sports.

Virtually all (97 percent) of the adults in Iowa who read the newspaper frequently (at least once a week) said they read the news, editorials, or financial pages (Table 4.7).

Roughly three-quarters of the newspaper readers in the state said they

usually read the home, fashion, health, or reviews sections; 85 percent said they read the advertisements or listings; and 77 percent reported reading the comics, horoscopes, or advice columns. Half the state's newspaper readers said they generally read the sports pages.

120 Language Use and Literacy Practices

IOWA TABLE 4.7

Average Literacy Proficiencies of Adults Who Read the Newspaper Regularly, by Parts Read: Results for Iowa

PARTS OF THE NEWSPAPER GENERALLY READ				Average p	roficiency of adul literacy scale	ts on each
			[Prose	Document	Quantitative
	n	WGT N (/1000)	PCT	PROF (SE)	PROF (SE)	PROF (SE)
News, editorials, financial news Yes No	1,117 37	1,852 63	97 3	289(3.8) ***(****)	283(3.9) ***(****)	292(4.8) ***(****)
Home, fashion, health, reviews Yes No	929 225	1,492 423	78 22	293 (2.5) 275 (11.1)	287 (2.4) 272 (12.9)	293 (3.2) 285 (12.6)
Advertisements, listings Yes No	1,018 136	1,627 288	85 15	292 (2.5) 275 (34.4)!	287 (2.8) 264 (35.6)!	294 (2.5) 277 (39.3)!
<u>Comics, horoscope, advice</u> Yes No	887 267	1,474 441	77 23	291 (2.0) 282 (13.6)	286 (2.0) 274 (15.3)	293 (2.5) 286 (16.9)
<u>Sports</u> Yes No	561 593	956 959	50 50	291 (2.6) 287 (6.9)	286 (3.0) 281 (8.2)	297 (3.3) 285 (9.1)

- n = sample size; WGT N = population size estimate / 1,000 (the sample sizes for subpopulations may not add up to the total sample sizes, due to missing data); PCT = percentage in group; PROF = average proficiency estimate; (SE) = standard error of the estimate (the reported sample estimate can be said to be within 2 standard errors of the true population value with 95% confidence).
- Percentages less than 0.5 are rounded to zero.
- *** Sample size is insufficient to permit a reliable estimate (fewer than 45 respondents).
- I Interpret with caution -- the nature of the sample does not allow accurate determination of the variability of this statistic.

Source: Educational Testing Service, State Adult Literacy Survey, 1992.

There are no significant differences in average prose, document, or quantitative proficiencies between individuals who reported that they usually read these sections of the newspaper and those who do not. Although some of the gaps between readers and nonreaders appear to be quite large, the variability in the samples (as reflected in the standard errors) prevents the differences from reaching statistical significance.

Section IV 121

Magazine and book reading practices

In addition to asking respondents about their newspaper reading practices, the survey requested information on the extent to which they engaged in other types of reading - in particular, reading magazines and books. Fourteen percent of the adults in Iowa said they do not read any magazines in English on a regular basis (Table 4.8). Another 35 percent read one or two, 39 percent read three to five, and 12 percent read six or more magazines regularly.



IOWA TABLE 4.8

Average Literacy Proficiencies, by Magazine and **Book Reading Practices: Results for Iowa**

MAGAZINE AND BOOK READING (IN ENGLISH)				Average proficiency of adults on each literacy scale		
				Prose	Document	Quantitative
	n	WGT N (/1000)	PCT	PROF (SE)	PROF (SE)	PROF (SE)
Number of different magazines looked at or read regularly	12					
0 1 or 2 3 to 5	153 438 484	290 723	14 35	253 (9.4) 283 (6.2) 293 (2.0)	253 (10.5) 278 (8.0)	258 (11.9) 284 (7.3)
6 or more	171	261	12	306 (7.8)	298 (5.8)	310 (8.5)
Read a book in the past six months	1.00		1.1			
Yes No	1,100 144	1,731 363	83 17	294 (3.0) 243 (6.1)!	288 (3.5) 240 (9.0)!	295 (2.7) 249 (6.8)!
Types of books read in the past six months	1.000				1	
Fiction Recreation or entertainment	664 406	1,014 604		305 (2.9) 297 (9.1)	299 (2.7) 291 (12.1)	303 (2.3) 298 (10.8)
Current affairs or history	414	582		304 (9.9)	297 (12.4)	304 (10.5)
Science or social science Reference	409 284 759	651 389 1.131		292 (3.6) 313 (5.6) 304 (3.0)	282 (3.6) 307 (6.8) 299 (3.3)	289 (5.8) 312 (6.1) 305 (3.0)
Manuals Any other types	843 305	1,301 425		300 (2.8) 307 (5.9)	294 (2.8) 300 (6.9)	303 (2.5) 306 (6.7)

n = sample size; WGT N = population size estimate / 1,000 (the sample sizes for subpopulations may not add up to the total sample sizes, due to missing data); PCT = percentage in group; PROF = average proficiency estimate; (SE) = standard error of the estimate (the reported sample estimate can be said to be within 2 standard errors of the true population value with 95% confidence).

- Percentages less than 0.5 are rounded to zero.
- Sample size is insufficient to permit a reliable estimate (fewer than 45 respondents).
- Interpret with caution -- the nature of the sample does not allow accurate determination of the variability of this statistic.

Source: Educational Testing Service, State Adult Literacy Survey, 1992.

122 Language Use and Literacy Practices

Survey participants were also asked what types of books they had read in English within the past six months. (The response percentages are not all shown in the table, as respondents were asked to indicate each type of book they had read, thus yielding a total of more than 100 percent.) Seventeen percent of the respondents said they had not read any books, while the remainder (83 percent) had read at least one. The types of books most commonly cited were manuals, reference books, and fiction. Among those adults in Iowa who had reportedly read a book in the past half-year, threequarters had read a manual for cooking, operating, repairing, or building, and 65 percent had read a reference book (such as an encyclopedia or dictionary). Fifty-eight percent of the readers had read a work of fiction.

Somewhat smaller percentages of adults said they had read other types of books in English. About one-third (35 percent) of the adults who had read a book within the past half-year had read one on recreation or entertainment, on current affairs or history (34 percent), or on inspiration or religion (38 percent). Twenty-two percent had read a science or social science book, and 25 percent reported having read some other type of book.

The relationship between adults' literacy skills and their magazine and book reading practices is quite clear. Individuals who said they read at least a few magazines on a regular basis performed far better than those who do not, and the more magazines they read, the higher their average literacy scores tended to be. On the prose scale, Iowa residents who do not read any magazines had an average score of 253, compared with 306 for adults who read six or more magazines on a regular basis.

Similarly, adults who had read a book in English in the past six months

performed better in the assessment than those who had not; on each literacy scale, the gap between these two groups was approximately 50 points. In general, respondents who had not read any books in English performed in the Level 2 range, while those who had done so displayed skills in the Level 3 range. There were no significant differences among respondents according to the types of books they had read.

Frequency of library use

Survey participants were asked how often they use the services of a library. Thirty-one percent of the adults in Iowa said they never do so, and another 30 percent said they do so only once or twice a year (Table 4.9). Nineteen percent estimated that they use library services monthly, 17 percent said they do so weekly, and 3 percent said they do so every day.

Section IV 123



IOWA TABLE 4.9

Average Literacy Proficiencies, by Frequency of Library Use: Results for Iowa

FREQUENCY OF LIBRARY USE				Average proficiency of adults on each literacy scale			
				Prose	Document	Quantitative	
	n	WGT N (/1000)	PCT	PROF (SE)	PROF (SE)	PROF (SE)	
Frequency of library use							
Daily	43	60	3	*** (****)	*** (****)	*** (****)	
Weekly	246	352	17	310 (6.5)	303 (6.8)	304 (5.9)	
Monthly	260	400	19	304 (3.9)	298 (2.8)	304 (5.9)	
Once or twice a year	406	626	30	295 (2.5)	290 (2.8)	299 (3.1)	
Never	291	657	31	249 (5.6)	244 (5.5)	254 (6.4)	

n = sample size; WGT N = population size estimate / 1,000 (the sample sizes for subpopulations may not add up to the total sample sizes, due to missing data); PCT = percentage in group; PROF = average proficiency estimate; (SE) = standard error of the estimate (the reported sample estimate can be said to be within 2 standard errors of the true population value with 95% confidence).

† Percentages less than 0.5 are rounded to zero.

*** Sample size is insufficient to permit a reliable estimate (fewer than 45 respondents).

I Interpret with caution -- the nature of the sample does not allow accurate determination of the variability of this statistic. Source: Educational Testing Service, State Adult Literacy Survey, 1992.

> The number of adults in Iowa who said they used the library every day was too small to provide reliable proficiency estimates, but in general, those who reported frequent use of the library demonstrated better literacy skills than less frequent users. The differences among the groups are most evident on the prose scale, where individuals who use the library at least once a week had an average score of 310, compared with only 249 — about 60 points lower — for adults who never use the library.

Amount of television watching

When asked how much television they watch each day, virtually all of the adults in Iowa (98 percent) said they watch at least some, although 23 percent said they spend no more than an hour on this pastime (Table 4.10). Twenty-four percent reported that they generally watch two hours of television a day, while 20 percent watch three hours, 13 percent watch four hours, 8 percent watch

124 Language Use and Literacy Practices

IOWA TABLE 4.10

Average Literacy Proficiencies, by Amount of Television Usually Watched Each Day: Results for Iowa

AMOUNT OF TELEVISION USUALLY WATCHED				Average proficiency of adults on each literacy sca		
EACH DAT				Prose	Document	Quantitative
	n	WGT N (/1000)	PCT	PROF (SE)	PROF (SE)	PROF (SE)
Amount of television usually watched each day None 1 hour or less 2 hours 3 hours 4 hours 5 hours 6 hours or more	28 313 336 247 152 74 96	34 478 505 424 264 164 226	2 23 24 20 13 8 11	*** (****) 306 (3.9) 294 (3.9) 282 (5.7) 277 (6.5) 261 (22.9)! 253 (10.8)!	*** (****) 298 (2.8) 292 (3.8) 274 (6.4) 270 (6.2) 262 (17.6)! 250 (11.3)!	*** (****) 309 (2.9) 301 (4.8) 283 (6.8) 276 (6.3) 263 (25.5)! 250 (12.2)!

n = sample size; WGT N = population size estimate / 1,000 (the sample sizes for subpopulations may not add up to the total sample sizes, due to missing data); PCT = percentage in group; PROF = average proficiency estimate; (SE) = standard error of the estimate (the reported sample estimate can be said to be within 2 standard errors of the true population value with 95% confidence).

Percentages less than 0.5 are rounded to zero.

*** Sample size is insufficient to permit a reliable estimate (fewer than 45 respondents).

I Interpret with caution -- the nature of the sample does not allow accurate determination of the variability of this statistic.

Source: Educational Testing Service, State Adult Literacy Survey, 1992.

five hours, and 11 percent watch six hours or more. In all, approximately onethird of the adults in the state (32 percent) spend four hours or more every day watching television.

There were substantial differences in literacy proficiency between adults who watch the most television and those who watch the least. Across the literacy scales, individuals who watch no more than one hour of television each day had average scores of 298 to 309. The average scores of respondents who said they watch four hours daily ranged from 270 to 277 — or about 30 points lower.

Personal and job-related use of prose materials

Survey respondents were asked how often they read various types of materials in English, either for their personal use or for their current or most recent jobs. One set of questions asked how often they read or use prose materials such as
letters, memos, reports, and articles. Half the adults in Iowa (51 percent) said they read letters or memos every day, 19 percent read them a few times a week, 12 percent read them once a week, 13 percent read them less than once a week, and 6 percent never read them (Table 4.11). Fewer respondents in the state reported reading reports or articles every day (35 percent); still, 30 percent said they read these materials a few times a week, and 15 percent said they did so once a week. Ten percent reported never reading reports or articles.

When asked how often they write letters or memos either for their own use or as part of their jobs, 39 percent of the Iowa respondents said they do so every day, and 22 percent do so a few times a week. Twenty percent said they write these materials less than once a week. Thus, writing letters or memos seems to be slightly less common than reading letters or memos.



IOWA TABLE 4.11

Types of Prose Materials Used for Personal or Job-related Reading and Writing: Results for Iowa

USE, TYPE OF PROSE MATERIAL			2	Per	centage of	adults	s who u	se ea	ich type	of ma	terial	
			Every	day	A few tim a week	ies o	nce a v	week	Less t once a	han week	Nev	/er
	n	WGT N (/1000)	RPCT (SE)	RPCT (SE	i) R	RPCT (S	SE)	RPCT (SE)	RPCT (SE)
Reads or uses:	1.014	0.000	54.4	0.01	10/01		10/0	-	10.1		~ (1.5
Reports articles	1,244	2,088	51 (2.6)	19 (3.	1)	12 (2	2.5)	13 (5.5)	6(1.5)
Writes or fills out:	1,211	2,000	001	,	00111	,	10 (2		10 (1.4)	10 (2.0)
Letters, memos	1,242	2,076	39 (4.7)	22 (1.2	2)	13 (1	.0)	20 (3	2.7)	6(2.4)
Reports, articles	1,241	2,067	19 (1.7)	13 (1.1	1)	13 (1	.0)	22 (1	2.6)	34 (3.9)

n = sample size; WGT N = population size estimate / 1,000 (the sample sizes for subpopulations may not add up to the total sample sizes, due to missing data); RPCT = row percentage estimate; (SE) = standard error of the estimate (the reported sample estimate can be said to

be within 2 standard errors of the true population value with 95% confidence).

† Percentages less than 0.5 are rounded to zero.

*** Sample size is insufficient to permit a reliable estimate (fewer than 45 respondents).

Source: Educational Testing Service, State Adult Literacy Survey, 1992.

126 Language Use and Literacy Practices

Report and article writing was, not surprisingly, far less common than letter or memo writing, and also far less common than report or article reading. Approximately one-third (34 percent) of the adults in Iowa said they never write these types of materials. Though nearly half said they do so at least once a week, this was still far smaller than the percentage who said they read these materials that often (80 percent).

Very large performance differences are found between adults who read and write prose frequently and those who do not (Table 4.12). Regardless of the activity or type of material, individuals who reported engaging in reading or writing every day had average prose scores (302 to 310) that were far higher than the scores of individuals who never engage in these activities (211 to 249).



IOWA TABLE 4.12

Average Prose Proficiency, by Types of Prose Materials Used for Personal or Job-related Reading and Writing: Results for Iowa

PROSE MATERIAL	Average pro	ose proficiency of adults who use each type of materia						
	Every day	A few times a week	Once a week	Less than once a week	Never			

and the second se	n	(/1000)	Thor (02)	11101 (02)	11101 (02)		1
<u>Reads or uses:</u> Letters, memos Reports, articles	1,244 1,244	2,088 2,088	302 (4.0) 304 (6.0)	290 (4.3) 296 (3.3)	272 (10.5) 284 (3.8)	249 (26.1) 268 (9.7)	236 (18.2) 211 (6.8)
<u>Writes or fills out:</u> Letters, memos Reports, articles	1,242 1,241	2,076 2,067	304 (8.1) 310 (4.5)	298 (3.4) 313 (6.8)	275 (5.1) 297 (3.8)	269 (15.6) 303 (4.5)	213 (10.7) 249 (6.4)

n = sample size; WGT N = population size estimate / 1,000 (the sample sizes for subpopulations may not add up to the total sample sizes, due to missing data); PROF = average proficiency estimate; (SE) = standard error of the estimate (the reported sample estimate can be said to be within 2 standard errors of the true population value with 95% confidence).

† Percentages less than 0.5 are rounded to zero.

*** Sample size is insufficient to permit a reliable estimate (fewer than 45 respondents).

Source: Educational Testing Service, State Adult Literacy Survey, 1992.

Personal and job-related use of documents

A second set of questions asked respondents to indicate how often they read or use various types of documents. Twenty-seven to 30 percent of the adults in Iowa said they read or use reference books, catalogs, lists, directions, instructions, bills, or spreadsheets every day, while 23 to 26 percent reported reading these materials a few times a week (Table 4.13). On the other hand, 7 to 13 percent of the Iowa respondents said they never use these types of documents. The use of diagrams or schematics was less common. Fourteen percent of the respondents said they read or use these types of documents every day, while 34 percent reported never using them. When asked how often they write or fill out forms, bills, or budgets, 29 percent of the adults in Iowa said they do so every day, and another 24 percent reported doing so a few times a week, while 9 percent said they never do.



t

IOWA TABLE 4.13

Types of Documents Used for Personal or Job-related Reading and Writing: Results for Iowa

USE, TYPE OF DOCUMENT	2		Pe	rcentage of adu	lts who use eac	h type of docun	nent
			Every day	A few times a week	Once a week	Less than once a week	Never
	n	WGT N (/1000)	RPCT (SE)	RPCT (SE)	RPCT (SE)	RPCT (SE)	RPCT (SE)
Reads or uses:			1				
Reference books, catalogs, lists	1,243	2,087	27 (1.9)	25 (0.9)	15 (1.3)	19 (1.6)	13 (3.8)
Directions, instructions	1,244	2,088	27 (2.2)	26 (2.0)	19 (3.0)	20 (1.1)	7 (1.9)
Bills, spreadsheets	1,243	2,078	30 (2.3)	23 (3.0)	11 (0.7) 18 (1.4)	26 (1.5) 22 (6.6)	7 (1.3)
Writes or fills out:							
Forms, bills, budgets	1,241	2,067	29 (2.1)	24 (1.6)	17 (1.7)	22 (3.0)	9 (0.6)

n = sample size; WGT N = population size estimate / 1,000 (the sample sizes for subpopulations may not add up to the total sample sizes, due to missing data); RPCT = row percentage estimate; (SE) = standard error of the estimate (the reported sample estimate can be said to be within 2 standard errors of the true population value with 95% confidence).

Percentages less than 0.5 are rounded to zero.

** Sample size is insufficient to permit a reliable estimate (fewer than 45 respondents).

Source: Educational Testing Service, State Adult Literacy Survey, 1992.

128 Language Use and Literacy Practices

Once again, adults who use various types of documents frequently outperformed those who do not (Table 4.14). For example, Iowa survey participants who said they read or use reference books, catalogs, or lists either every day or a few times a week had average document proficiencies that were substantially higher than those of participants who never read or use these materials.



IOWA TABLE 4.14

Average Document Proficiency, by Types of Documents Used for Personal or Job-related Reading and Writing: Results for Iowa

USE, TYPE OF DOCUMENT			Average document proficiency of adults who use each type of document								
			Every day	A few times a week	Once a week	Less than once a week	Never				
	n	WGT N (/1000)	PROF (SE)	PROF (SE)	PROF (SE)	PROF (SE)	PROF (SE)				
Reference books, catalogs, lists Directions, instructions Diagrams, schematics Bills, spreadsheets	1,243 1,244 1,243 1,242	2,087 2,088 2,078 2,086	304 (5.7) 284 (4.0) 301 (6.2) 301 (5.0)	297 (3.7) 289 (12.7) 309 (6.5) 302 (6.1)	286 (2.8) 279 (8.8) 295 (5.5) 281 (3.4)	263 (3.3) 279 (3.6) 294 (4.2) 250 (25.2)	218 (6.6) 242 (10.0) 244 (4.5) 212 (29.0)				
Writes or fills out:	1 241	2 067	302 (44)	298 (4.3)	286 (5.0)	257 (14.8)	217 (21.2)				

Forms, bins, budgets	11-11 -1-11				
		in the second se	 	the second se	

n = sample size; WGT N = population size estimate / 1,000 (the sample sizes for subpopulations may not add up to the total sample sizes, due to missing data); PROF = average proficiency estimate; (SE) = standard error of the estimate (the reported sample estimate can be said to be within 2 standard errors of the true population value with 95% confidence).

† Percentages less than 0.5 are rounded to zero.

*** Sample size is insufficient to permit a reliable estimate (fewer than 45 respondents).

Source: Educational Testing Service, State Adult Literacy Survey, 1992.

Personal use of mathematics

Respondents were also asked to indicate how often they use arithmetic or mathematics — that is, addition, subtraction, multiplication, division, or measurement — for their own use. Slightly more than half (52 percent) of the survey respondents in Iowa said they use mathematics every day, and 29 percent reported using it a few times a week (Table 4.15). Eight percent said they use it once a week, 10 percent said they use it less than once a week, and 1 percent said they never use mathematics.



IOWA TABLE 4.15

Quantitative Literacy Levels and Average Proficiencies, by Frequency of Arithmetic or Mathematics Use: Results for Iowa

FREQUENCY OF ARITHMETIC OR					Percentage	of adults in ea	ich quantitative	literacy level	
MATHEMATICS USE				Level 1 225 or lower	Level 2 226 to 275	Level 3 276 to 325	Level 4 326 to 375	Level 5 376 or higher	Average Proficiency
	n	WGT N (/1000)	PCT	RPCT (SE)	RPCT (SE)	RPCT (SE)	RPCT (SE)	RPCT (SE)	PROF (SE)
Every day A few times a week Once a week Less than once a week Never	676 373 108 75 12	1,085 606 173 203 23	52 29 8 10	6 (1.2) 14 (7.0) 24 (9.2) 47 (5.5)	22 (2.5) 22 (3.9) 24 (8.4) 25 (8.9)	40 (3.5) 36 (5.8) 35 (7.4) 19 (8.4)	27 (2.7) 23 (3.5) 14 (3.2) 8 (4.0)	5 (1.3) 5 (2.0) 3 (1.5) 1 (1.9)	301 (2.7) 293 (5.2) 271 (8.8) 229 (10.6)

n = sample size; WGT N = population size estimate / 1,000 (the sample sizes for subpopulations may not add up to the total sample sizes, due to missing data); PCT = percentage in group; RPCT = row percentage estimate; PROF = average proficiency estimate; (SE) = standard error of the estimate (the reported sample estimate can be said to be within 2 standard errors of the true population value with 95% confidence).

Percentages less than 0.5 are rounded to zero.

*** Sample size is insufficient to permit a reliable estimate (fewer than 45 respondents).

Source: Educational Testing Service, State Adult Literacy Survey, 1992.

130 Language Use and Literacy Practices

As expected, adults who said they rarely use mathematics were far more likely than those who use it frequently to perform in the lowest levels of quantitative literacy. Almost half the respondents who said they used math less than once a week were in Level 1 on the quantitative scale, and another 25 percent were in Level 2; conversely, only 8 percent reached Level 4 and 1 percent attained Level 5. As a result, their average quantitative proficiencies are quite low (229).

In contrast, far lower percentages of respondents who use mathematics every day performed in the lowest levels of quantitative literacy (6 percent in Level 1; 22 percent in Level 2), and far higher percentages reached the two highest levels (27 percent in Level 4; 5 percent in Level 5). As a result, their average quantitative proficiencies are quite high (301). Still, it is interesting to note that so many of the individuals who use mathematics every day performed in Levels 1 and 2. It appears that ability is not the sole predictor of mathematics use, and that many adults with limited quantitative skills are called upon to use these skills often.

Summary

Ninety-six percent of the survey respondents in Iowa and 85 percent of those nationwide said they spoke only English before beginning their schooling. In the national sample, these individuals demonstrated higher average proficiencies than adults who spoke another language either in addition to or instead of English.

Nearly all (92 percent) of the Iowa respondents who learned a language other than English before starting school said they usually speak English now, while 4 percent said they usually speak Spanish and another 4 percent said they usually speak a language other than English or Spanish.

Virtually all of the adults in Iowa described themselves as understanding (99 percent), speaking (99 percent), and reading (98 percent) English either well or very well, and 96 percent perceived themselves as writing it either well or very well. Those who described themselves as having limited writing skills did, in fact, demonstrate lower proficiencies than those who rated their skills more highly.

Ninety-seven percent of Iowa respondents said they get some or a lot of information about current events and public affairs from nonprint media, and 86 percent said they get some or a lot of information from print media. Approximately two-thirds reported getting some or a lot of information from

personal sources, such as family or friends. Iowa residents who said they get some or a lot of information from print media had considerably higher literacy scores than those who get little or no information from these sources.

The survey results indicate that newspaper reading is quite common, even among adults who displayed relatively limited literacy skills. Fifty-six percent of the adults in Iowa said they read the newspaper every day, and another onequarter reported reading it a few times a week. Five percent said they read a newspaper less than once a week, and 4 percent said they never read one.

Fourteen percent of the adults in Iowa said they do not read any magazines in English on a regular basis, while the remainder reported reading at least one or two. Similarly, 17 percent of the survey respondents said they had not read any books in English in the past six months, while the remainder had read at least one. Individuals who said they read a few magazines on a regular basis performed far better than those who do not read any, and the more magazines they read the better their performance was likely to be. Similarly, adults who had read a book in English in the past six months performed better in the assessment, on average, than those who had not.

When asked how often they use a library, 31 percent of the adults in Iowa said they never do and another 30 percent said they do so only once or twice a year. Twenty percent of the state's respondents reported using a library on a weekly or daily basis. In general, those who reported frequent use of a library demonstrated better literacy skills than infrequent users.

Virtually all respondents in Iowa (98 percent) said they watch at least some television each day, although 23 percent said they spend no more than an hour on this pastime. The remainder watch at least two hours of television a day. Adults who watch four hours of television demonstrated far lower proficiencies in the assessment, on average, than individuals who watch an hour or less.

Finally, survey respondents were asked how often they read or use various types of materials in English, either for their personal use or for their jobs. Proficiency differences are found on the prose scale between adults who read and write prose frequently and those who do not. Similarly, adults who often use various types of documents had higher average document proficiencies than those who do not. Adults who said they rarely use mathematics were far more likely than those who use it frequently to perform in the lowest levels of













SECTION V

Interpreting the Literacy Scales

Building on the two earlier literacy surveys conducted by Educational Testing Service (ETS), the performance results from the National and State Adult Literacy Surveys are reported on three literacy scales — prose, document, and quantitative — rather than on a single conglomerate scale. Each of the three literacy scales ranges from 0 to 500.

The purpose of this section of the report is to give meaning to the literacy scales — or, more specifically, to interpret the numerical scores that are used to represent adults' proficiencies on these scales. Toward this end, the section begins with a brief summary of the task development process and of the way in which the literacy levels are defined. A detailed description of the prose, document, and quantitative scales is then provided. The five levels on each scale are defined, and the skills and strategies needed to successfully perform the tasks in each level are discussed. Sample tasks are presented to illustrate the types of materials and task demands that characterize the levels on each scale. The section ends with a brief summary of the probabilities of successful performance on tasks within each level for individuals who demonstrated different proficiencies.



Building the Literacy Tasks

The literacy scales make it possible not only to summarize the literacy proficiencies of the total population and of various subpopulations, but also to determine the relative difficulty of the literacy tasks administered in the survey. That is, just as an individual receives a score according to his or her performance on the assessment tasks, each task receives a value according to its difficulty as determined by the performance of the adults who participated in the survey. Previous research conducted at ETS has shown that the difficulty of a literacy task, and therefore its placement on a particular literacy scale, is determined by three factors: the structure or linguistic format of the material,

the content and/or the context from which it is selected, and the nature of the task, or what the individual is asked to do with the material.

Materials. The materials selected for inclusion in the survey reflect a variety of linguistic formats that adults encounter in their daily activities. Most of the prose materials used in the survey are expository — that is, they describe, define, or inform — since most of the prose that adults read is expository in nature; however, narratives and poetry are included, as well. The prose materials include an array of linguistic structures, ranging from texts that are highly organized both topically and visually to those that are loosely organized. They also include texts of varying lengths, from multiple-page magazine selections to short newspaper articles. All prose materials included in the survey were reproduced in their original format.

The document materials represent a wide variety of structures, which are characterized as tables, charts and graphs, forms, and maps, among other categories. Tables include matrix documents in which information is arrayed in rows and columns — for example, bus or airplane schedules, lists, or tables of numbers. Documents categorized as charts and graphs include pie charts, bar graphs, and line graphs. Forms are documents that require information to be filled in, while other structures include such materials as advertisements and coupons.

The quantitative tasks require the reader to perform arithmetic operations using numbers that are embedded in print. Since there are no materials that are unique to quantitative tasks, these tasks were based on prose materials and documents. Most quantitative tasks were, in fact, based on document structures.

Content and/or Contexts. Adults do not read printed or written materials in a vacuum. Rather, they read within a particular context or for a particular purpose. Accordingly, the survey materials represent a variety of contexts and contents. Six such areas were identified: home and family; health and safety; community and citizenship; consumer economics; work; and leisure and recreation.

In selecting materials to represent these areas, efforts were made to include as broad a range as possible, as well as to select universally relevant contexts and contents. This was to ensure that the materials would not be so specialized as to be familiar only to certain groups. In this way, disadvantages for individuals with limited background knowledge were minimized.

Types of Tasks. After the materials were selected, tasks were developed to accompany the materials. These tasks were designed to simulate the ways in which people use various types of materials and to require different strategies for successful task completion. For both the prose and document scales, the tasks can be organized into three major categories: *locating, integrating,* and

generating information. In the locating tasks, readers are asked to match information that is given in a question or directive with either literal or synonymous information in the text or document. Integrating tasks require the reader to incorporate two or more pieces of information located in different parts of the text or document. Generating tasks require readers not only to process information located in different parts of the material, but also to go beyond that information by drawing on their knowledge about a subject or by making broad text-based inferences.

Quantitative tasks require readers to perform arithmetic operations addition, subtraction, multiplication, or division — either singly or in combination. In some tasks, the type of operation that must be performed is obvious from the wording of the question, while in other tasks the readers must infer which operation is to be performed. Similarly, the numbers that are required to perform the operation can, in some cases, be easily identified, while in others, the numbers that are needed are embedded in text. Moreover, some quantitative tasks require the reader to explain how the problem would be solved rather than perform the calculation, and on some tasks the use of a simple four-function calculator is required.

Defining the Literacy Levels

The relative difficulty of the assessment tasks reflects the interactions among the various task characteristics described here. As shown in Figure 1 in the Introduction to this report, the score point assigned to each task is the point at which the individuals with that proficiency score have a high probability of responding correctly. In this survey, an 80 percent probability of correct response was the criterion used. While some tasks were at the very low end of the scale and some at the very high end, most had difficulty values in the 200 to 400 range. By assigning scale values to both the individuals and tasks, it is possible to see how well adults with varying proficiencies performed on tasks of varying difficulty. While individuals with low proficiency tend to perform well on tasks with difficulty values equivalent to or below their level of proficiency, they are less likely to succeed on tasks with higher difficulty values. This does not mean that individuals with low proficiency can never succeed on more difficult literacy tasks — that is, on tasks whose difficulty values are higher than their proficiencies. They may do so some of the time. Rather, it means that their probability of success is not as high. In other words, the more difficult the task relative to their proficiency, the lower their likelihood of responding correctly.

The response probabilities for two tasks on the prose scale are displayed in Figure 5.1. The difficulty of the first task is measured at the 250 point on the scale, and the second task is at the 350 point. This means that an individual would have to score at the 250 point on the prose scale to have an 80 percent chance (that is, a .8 probability) of responding correctly to Task 1. Adults scoring at the 200 point on the prose scale have only about a 40 percent chance of responding correctly to this task, whereas those scoring at the 300 point and above would be expected to rarely miss this task and others like it.

In contrast, an individual would need to score at the 350 point to have an 80 percent chance of responding correctly to Task 2. While individuals performing at the 250 point would have an 80 percent chance of success on the first task, their probability of answering the more difficult second task correctly is only 20 percent. An individual scoring at the 300 point is likely to succeed on this more difficult task only half the time.

NALS

Figure 5.1

Probabilities of Successful Performance on Two Prose Tasks by Individuals at Selected Points on the Prose Scale





An analogy may help clarify the information presented for the two prose tasks. The relationship between task difficulty and individual proficiency is much like the high jump event in track and field, in which an athlete tries to jump over a bar that is placed at increasing heights. Each high jumper has a height at which he or she is proficient. That is, he or she is able to clear the bar at that height with a high probability of success, and can clear the bar at lower

levels almost every time. When the bar is higher than their level of proficiency, however, they can be expected to have a much lower chance of clearing it successfully.

Once the literacy tasks are placed on their respective scales, using the criterion described here, it is possible to see how well the interactions among the task characteristics explain the placement of various tasks along the scales.¹ In investigating the progression of task characteristics across the scales, certain questions are of interest. Do tasks with similar difficulty values (that is, with difficulty values near one another on a scale) have certain shared characteristics? Do these characteristics differ in systematic ways from tasks in either higher or lower levels of difficulty? Analyses of the interactions between the materials read and the tasks based on these materials reveal that an ordered set of information-processing skills appears to be called into play to perform the range of tasks along each scale.

To capture this ordering, each scale was divided into five levels that reflect the progression of information-processing skills and strategies: Level 1 (0 to 225), Level 2 (226 to 275), Level 3 (276 to 325), Level 4 (326 to 375), and Level 5 (376 to 500). These levels were determined not as a result of any statistical property of the scales, but rather as a result of shifts in the skills and strategies required to succeed on various tasks along the scales, from simple to complex.

The remaining pages of this section describe each scale in terms of the nature of the task demands at each of the five levels. After a brief introduction to each scale, sample tasks in each level are presented and the factors contributing to their difficulty are discussed. The aim of these discussions is to give meaning to the scales and to facilitate interpretation of the results provided in the first and second sections of this report.

Interpreting the Literacy Levels

Prose literacy

The ability to understand and use information contained in various kinds of textual material is an important aspect of literacy. Most of the prose materials administered in this assessment were expository — that is, they inform, define, or describe — since these constitute much of the prose that adults read. Some narrative texts and poems were included, as well. The prose materials were drawn from newspapers, magazines, books, brochures, and pamphlets and reprinted in their entirety, using the typography and layout of the original source. As a result, the materials vary widely in length, density of information,

¹ I.S. Kirsch and P.B. Mosenthal. (1990). "Exploring Document Literacy: Variables Underlying the Performance of Young Adults." *Reading Research Quarterly*, 25. pp. 5-30.

and the use of structural or organizational aids such as section or paragraph headings, italic or boldface type, and bullets.

Each prose selection was accompanied by one or more questions or directives which asked the reader to perform specific tasks. These tasks represent three major aspects of information-processing: locating, integrating, and generating. Locating tasks require the reader to find information in the text based on conditions or features specified in the question or directive. The match may be literal or synonymous, or the reader may need to make a textbased inference in order to perform the task successfully. Integrating tasks ask the reader to compare or contrast two or more pieces of information from the text. In some cases the information can be found in a single paragraph, while in others it appears in different paragraphs or sections. In the generating tasks, readers must produce a written response by making text-based inferences or drawing on their own background knowledge.

In all, the prose literacy scale includes 41 tasks with difficulty values ranging from 149 to 468. It is important to remember that the locating, generating, and integrating tasks extend over a range of difficulty as a result of interactions with other variables including:

- the number of categories or features of information that the reader must process
- the number of categories or features of information in the text that can distract the reader, or that may seem plausible but are incorrect
- the degree to which information given in the question is obviously related to the information contained in the text
- the length and density of the text

The five levels of prose literacy are defined, and sample tasks provided, in the following pages.

Prose Level 1Scale range: 0 to 225Most of the tasks in this level require the reader to read relatively
short text to locate a single piece of information which is identical to
or synonymous with the information given in the question or
directive. If plausible but incorrect information is present in the text,

it tends not to be located near the correct information.

Percentage of adults in the state performing in this level: 14% Percentage of adults in the nation performing in this level: 21%

Tasks in this level require the reader to locate and match a single piece of information in the text. Typically the match between the question or directive and the text is literal, although sometimes synonymous matches may be necessary. The text is usually brief or has organizational aids such as paragraph headings or italics that suggest where in the text the reader should search for the specified information. The word or phrase to be matched appears only once in the text.

One task in Level 1 with a difficulty value of 210 asks respondents to read a newspaper article about a marathon swimmer and to underline the sentence that tells what she ate during a swim. Only one reference to food is contained in the passage, and it does not use the word "ate." Rather, the article says the swimmer "kept up her strength with banana and honey sandwiches, hot chocolate, lots of water and granola bars." The reader must match the word "ate" in the directive with the only reference to foods in the article.

Underline the sentence that tells what Ms. Chanin ate during the swim.

Swimmer completes Manhattan marathon

The Associated Press NEW YORK—University of Maryland senior Stacy Chanin on Wednesday became the first person to swim three 28-mile laps Chanin has twice circled Manhattan before and trained for the new feat by swimming about 28.4 miles a week. The Yonkers native has competed as a swimmer since she was 15 and hoped to persuade Olympic authorities to add a long-distance swimming event. The Leukemia Society of America solicited pledges for each mile she swam. In July 1983, Julie Ridge became the first person to swim around Manhattan twice. With her three laps, Chanin came up just short of Diana Nyad's distance record, set on a Florida-to-Cuba swim.

around Manhattan.

Chanin, 23, of Virginia, climbed out of the East River at 96th Street at 9:30 p.m. She began the swim at noon on Tuesday.

A spokesman for the swimmer, Roy Brunett, said Chanin had kept up her strength with "banana and honey" sandwiches, hot chocolate, lots of water and granola bars."

Reduced from original copy.

Prose Level 2

Scale range: 226 to 275

Some tasks in this level require readers to locate a single piece of information in the text; however, several distractors or plausible but incorrect pieces of information may be present, or low-level inferences may be required. Other tasks require the reader to integrate two or more pieces of information or to compare and contrast easily identifiable information based on a criterion provided in the question or directive.

Percentage of adults in the state performing in this level: 24% Percentage of adults in the nation performing in this level: 27%

Like the tasks in Level 1, most of the tasks in this level ask the reader to locate information. However, these tasks place more varied demands on the reader. For example, they frequently require readers to match more than a single piece of information in the text and to discount information that only partially satisfies the question. If plausible but incomplete information is included in the text, such distractors do not appear near the sentence or paragraph that contains the correct answer. For example, a task based on the sports article reproduced earlier asks the reader to identify the age at which the marathon swimmer began to swim competitively. The article first provides the swimmer's current age of 23, which is a plausible but incorrect answer. The correct information, age 15, is found toward the end of the article.

In addition to directing the reader to locate more than a single piece of information in the text, low-level inferences based on the text may be required to respond correctly. Other tasks in Level 2 (226 to 275) require the reader to identify information that matches a given criterion. For example, in one task with a difficulty value of 275, readers were asked to identify specifically what was wrong with an appliance by choosing the most appropriate of four statements describing its malfunction.

A manufacturing company provides its customers with the following instructions for returning appliances for service:

When returning appliance for servicing, include a note telling as clearly and as specifically as possible what is wrong with the appliance.

A repair person for the company receives four appliances with the following notes attached. Circle the letter next to the note which best follows the instructions supplied by the company.

A

B

The clock does not run correctly on this clock radio. I tried fixing it, but I couldn't.

С

The alarm on my clock radio doesn't go off at the time I set. It rings 15-30 minutes later.

My clock radio is not working. It stopped working right after I

D

This radio is broken. Please repair and return by United Parcel Service to the address on my slip.

Readers in this level may also be asked to infer a recurring theme. One task with a difficulty value of 262 asks respondents to read a poem that uses several metaphors to represent a single, familiar concept and to identify its theme. The repetitiveness and familiarity of the allusions appear to make this "generating" task relatively easy.

Prose Level 3

Scale range: 276 to 325

Tasks in this level tend to require readers to make literal or synonymous matches between the text and information given in the task, or to make matches that require low-level inferences. Other tasks ask readers to integrate information from dense or lengthy text that contains no organizational aids such as headings. Readers may also be asked to generate a response based on information that can be easily identified in the text. Distracting information is present, but is not located near the correct information.

Percentage of adults in the state performing in this level: 37% Percentage of adults in the nation performing in this level: 32%

One of the easier Level 3 tasks requires the reader to write a brief letter explaining that an error has been made on a credit card bill. This task is at 288 on the prose scale. Other tasks in this level require the reader to search fairly dense text for information. Some of the tasks ask respondents to make a literal or synonymous match on more than a single feature, while other tasks ask them to integrate multiple pieces of information from a long passage that does not contain organizational aids.

One of the more difficult Level 3 tasks (with a difficulty value of 316) requires the reader to read a magazine article about an Asian-American woman and to provide two facts that support an inference made from the text. The question directs the reader to identify what Ida Chen did to help resolve conflicts due to discrimination.

List two things that Chen became involved in or has done to help resolve conflicts due to discrimination.

IDA CHEN is the first Asian-American woman to become a judge of the Commonwealth of Pennsylvania.

She understands discrimination because she has experienced it herself.

Soft-spoken and eminently dignified, Judge Ida Chen prefers hearing about a new acquaintance rather than talking about herself. She wants to know about career plans, hopes, dreams, fears. She gives unsolicited advice as well as encouragement. She instills confidence.

Her father once hoped that she would become a professor. And she would have also made an outstanding social worker or guidance counselor. The truth is that Chen wears the caps of all these professions as a Family Court judge of the Court of Common Pleas of Philadelphia County, as a participant in public advocacy for minorities, and as a particularly sensitive, caring person.

She understands discrimination because she has experienced it herself. As an elementary school student, Chen tried to join the local Brownie troop. "You can't be a member," she was told. "Only American girls are in the Brownies."

Originally intent upon a career as a journalist, she selected Temple University because of its outstanding journalism department and affordable tuition. Independence being a personal need, she paid for her tuition by working for Temple's Department of Criminal Justice. There she had her first encounter with the legal world and it turned her career plans in a new direction law school. Through meticulous planning, Chen was able to earn her undergraduate degree in two and a half years and she continued to work three jobs. But when she began her first semester as a Temple law student in the fall of 1973, she was barely able to stay awake. Her teacher Lynne Abraham, now a Common Pleas Court judge herself, couldn't help but notice Chen yawning in the back of the class, and when she determined that this student was not a party animal but a workhorse, she arranged a teaching assistant's job for Chen on campus. After graduating from Temple Law School in 1976, Chen worked for the U.S. Equal Employment Opportunity Commission where she was a litigator on behalf of plaintiffs who experienced discrimination in the workplace, and

then moved on to become the first Asian-American to serve on the Philadelphia Commission on Human Relations.

Appointed by Mayor Wilson Goode, Chen worked with community leaders to resolve racial and ethnic tensions and also made time to contribute free legal counsel to a variety of activist groups.

The "Help Wanted" section of the newspaper contained an entry that aroused Chen's curiosity — an ad for a judge's position. Her application resulted in her selection by a state judicial committee to fill a seat in the state court. And in July of 1988, she officially became a judge of the Court of Common Pleas. Running as both a Republican and Democratic candidate, her position was secured when she won her seat on the bench at last November's election.

At Family Court, Chen presides over criminal and civil cases which include adult sex crimes, domestic violence, juvenile delinquency, custody, divorce and support. Not a pretty picture.

Chen recalls her first day as judge, hearing a juvenile dependency case — "It was a horrifying experience. I broke down because the cases were so depressing," she remembers.

Outside of the courtroom, Chen has made a name for herself in resolving interracial conflicts, while glorying in her Chinese-American identity. In a 1986 incident involving the desecration of Korean street signs in a Philadelphia neighborhood, Chen called for a meeting with the leaders of that community to help resolve the conflict. Chen's interest in community advocacy is not limited to Asian communities. She has been involved in Hispanic, Jewish and Black issues, and because of her participation in the Ethnic Affairs Committee of the Anti-Defamation League of B'nai B'rith, Chen was one of 10 women nationwide selected to take part in a mission to Israel. With her recently won mandate to judicate in the affairs of Pennsylvania's citizens, Chen has pledged to work tirelessly to defend the rights of its people and contribute to the improvement of human welfare. She would have made a fabulous Brownie.

Jessica Schultz

Prose Level 4

Scale range: 326 to 375

These tasks require readers to perform multiple-feature matches and to integrate or synthesize information from complex or lengthy passages. More complex inferences are needed to perform successfully. Conditional information is frequently present in tasks in this level and must be taken into consideration by the reader.

Percentage of adults in the state performing in this level: 21% Percentage of adults in the nation performing in this level: 17%

A prose task with a difficulty value of 328 requires the reader to synthesize the repeated statements of an argument from a newspaper column in order to generate a theme or organizing principle. In this instance, the supporting statements are elaborated in different parts of a lengthy text.

A more challenging task (with a difficulty value of 359) directs the reader to contrast the two opposing views stated in the newspaper feature reprinted here that discusses the existence of technologies that can be used to produce more fuel-efficient cars.

Contrast Dewey's and Hanna's views about the existence of technologies that can be used to produce more fuel-efficient cars while maintaining the size of the cars.

Face-Off: Getting More Miles Per Gallon

Demand cars with better gas mileage

By Robert Dewey Guest columnist

Section

. 147

WASHINGTON - Warning: Automakers are resurrecting their heavymetal dinosaurs, aka gas guzzlers.

Government reports show that average new-car mileage has declined to 28.2 miles per gallon - the 1986 level. To reverse this trend, Congress must significantly increase existing gas-mileage standards. More than half our Nobel laureates

and 700 members of the National Academy of Sciences recently called global warming "the most serious environmental threat of the 21st century." In 1989, oil imports climbed to a near-record 46% of U.S. consumption. Increasing gas mileage is the single biggest step we can take to reduce oil imports and curb global warming. Greater efficiency also lowers our trade deficit (oil imports represent 40% of it) and decreases the need to drill in pristine areas.

Bigger engines and bigger cars mean bigger profits for automakers, who offer us the products they want us to buy. More than ever, Americans want products that have less of an environmental impact. But with only a few fuel-efficient cars to choose from, how do we find ones that meet all our needs?

Government studies show automakers have the technology to dramatically im- Reprinted by permission of USA Today.

prove gas mileage - while maintaining the 1987 levels of comfort, performance and size mix of vehicles. Automakers also have the ability to make their products safer. The cost of these improvements will be offset by savings at the gas pump!

Cars can average 45 mpg and light trucks 35 mpg primarily by utilizing engine and transmission technologies already on a few cars today. Further improvements are possible by using technologies like the two-stroke engine and better aerodynamics that have been developed but not used.

When the current vehicle efficiency standards were proposed in 1974, Ford wrongly predicted that they "would require either all sub-Pinto-sized vehicles or some mix of vehicles ranging from a sub-subcompact to perhaps a Maverick." At that time, Congress required a 100% efficiency increase; raising gas mileage to 45 mpg requires only a 60% increase.

Americans want comfortable, safe and efficient cars. If automakers won't provide them, Congress must mandate them when it considers the issue this summer.

Let's hope lawmakers put the best interest of the environment and the nation ahead of the automakers' lobbyists and political action committees.

Robert Dewey is a conservation analyst for the Environmental Action Foundation.

Guest columnist

DETROIT - Do Americans look forward to the day when they'll have to haul groceries, shuttle the kids to and from school or take family vacations in compact and subcompact cars?

will be made.

But to achieve a U.S. fleet average of 40 mpg to 45 mpg, carmakers would have to sharply limit the availability of familysize models and dramatically trim the size and weight of most cars. There simply are not magic technolo-

Don't demand end to cars people want

By Thomas H. Hanna

I doubt it - which is why U.S. and import carmakers oppose the 40-milesper-gallon to 45 mpg corporate average fuel economy mandates that some are pushing in Congress, either to curb tailpipe carbon dioxide emissions because of alleged global warming or for energy conservation. Since the mid-1970s, automakers have doubled the fleet average fuel economy of new cars to 28 mpg — and further progress

Compact and subcompact cars with mileage of 40 mpg or better are now available, yet they appeal to only 5% of U.S. car buyers.

gies to meet such a standard.

Almost every car now sold in the USA

would have to be drastically downsized, and many would be obsolete.

As a result, Americans each year would be unable to buy the vehicles most suited for their needs: mid- and family-size models, luxury automobiles, mini-vans, small trucks and utility vehicles.

The fleet shift to compacts and subcompacts could also force the closing of assembly plants, supplier firms and dealerships, at a cost of thousands of U.S. jobs.

Although a growing number of scientists are skeptical of global warming, the issue deserves thorough international scientific evaluation, not premature unilateral U.S. action.

Carbon dioxide emissions from U.S. vehicles total less than 2.5% of worldwide "greenhouse" gases. Even doubling today's corporate average fuel economy for U.S. cars - if technically possible - would cut those gases about .5%

Whatever the motivation - alleged global warming or energy conservation the stakes are high for millions of Americans and thousands of U.S. jobs in unrealistic corporate average fuel economy mandates.

Thomas H. Hanna is president and chief executive officer of the Motor Vehicle Manufacturers Association of the United States. Reprinted by permission of USA Today.

Reduced from original copy.

Two other tasks in Level 4 on the prose scale require the reader to draw on background knowledge in responding to questions asked about two poems. In one they are asked to generate an unfamiliar theme from a short poem (difficulty value of 362), and in the other they are asked to compare two metaphors (value of 374).

Prose Level 5

Scale range: 376 to 500

Some tasks in this level require the reader to search for information in dense text which contains a number of plausible distractors. Others ask readers to make high-level inferences or use specialized background knowledge. Some tasks ask readers to contrast complex information.

Percentage of adults in the state performing in this level: 3% Percentage of adults in the nation performing in this level: 3%

Two tasks in Level 5 require the reader to search for information in dense text containing several plausible distractors. One such task (difficulty value of 410) requires the respondent to read information about jury selection and service. The question requires the reader to interpret information to identify two ways in which prospective jurors may be challenged.

Identify and summarize the two kinds of challenges that attorneys use while selecting members of a jury.



DO YOU HAVE A QUESTION?

QUESTION: What is the new program for scheduling jurors?

ANSWER: This is a new way of organizing and scheduling jurors that is being introduced all over the country. The goals of this program are to save money, increase the number of citizens who are summoned to serve and decrease the inconvenience of serving.

The program means that instead of calling jurors for two weeks, jurors now serve only one day, or for the length of one trial if they are selected to hear a case. Jurors who are not selected to hear a case are excused at the end of the day, and their obligations to serve as jurors are fulfilled for three years. The average trial lasts two days once testimony begins.

An important part of what is called the One Day - One Trial program is the "standby" juror. This is a person called to the Courthouse if the number of cases to be tried requires more jurors than originally estimated. Once called to the Courthouse, the standby becomes a "regular" juror, and his or her service is complete at the end of one day or one trial, the same as everyone else.

Q. How was I summoned?

A. The basic source for names of eligible

ensure that all of the jurors who are selected to hear the case will be unbiased, objective and attentive.

In most cases, prospective jurors will be asked to raise their hands when a particular question applies to them. Examples of questions often asked are: Do you know the Plaintiff, Defendant or the attorneys in this case? Have you been involved in a case similar to this one yourself? Where the answer is yes, the jurors raising hands may be asked additional questions, as the purpose is to guarantee a fair trial for all parties. When an attorney believes that there is a legal reason to excuse a juror, he or she will challenge the juror for cause. Unless both attorneys agree that the juror should be excused, the Judge must either sustain or override the challenge.

After all challenges for cause have been ruled upon, the attorneys will select the trial jury from those who remain by exercising peremptory challenges. Unlike challenges for cause, no reason need be given for excusing a juror by peremptory challenge. Attorneys usually exercise these challenges by taking turns striking names from a list until both are satisfied with the jurors at the top of the list or until they use up the number of challenges allowed. Challenged jurors and any extra jurors will then be excused and asked to return to the jury selection room.

jurors is the Driver's License list which is supplemented by the voter registration list. Names are chosen from these combined lists by a computer in a completely random manner.

Once in the Courthouse, jurors are selected for a trial by this same computer and random selection process.

- Q. How is the Jury for a particular trial selected?
- A. When a group of prospective jurors is selected, more than the number needed for a trial are called. Once this group has been seated in the courtroom, either the Judge or the attorneys ask questions. This is called voir dire. The purpose of questions asked during voir dire is to

Jurors should not feel rejected or insulted if they are excused for cause by the Court or peremptorily challenged by one of the attorneys. The voir dire process and challenging of jurors is simply our judicial system's way of guaranteeing both parties to a lawsuit a fair trial.

Q. Am I guaranteed to serve on a jury?

A. Not all jurors who are summoned actually hear a case. Sometimes all the Judges are still working on trials from the previous day, and no new jurors are chosen. Normally, however, some new cases begin every day. Sometimes jurors are challenged and not selected.

A somewhat more demanding task (difficulty value of 423) involves the magazine article on Ida Chen reproduced earlier. This more challenging task requires the reader to explain the phrase "recently won mandate" used at the end of the text. To explain this phrase, the reader needs to understand the concept of a political mandate as it applies to Ida Chen and the way she is portrayed in this article.

Document literacy

Another important aspect of being literate in modern society is having the knowledge and skills needed to process information from documents. We often encounter tables, schedules, charts, graphs, maps, and forms in everyday life, both at home and at work. In fact, researchers have found that many of us spend more time reading documents than any other type of material.² The ability to locate and use information from documents is therefore essential.

Success in processing documents appears to depend at least in part on the ability to locate information in complex arrays and to use this information in the appropriate ways. Procedural knowledge may be needed to transfer information from one source or document to another, as is necessary in completing applications or order forms.

The document literacy scale contains 81 tasks with difficulty values that range from 69 to 396 on the scale. By examining tasks associated with various proficiency levels, we can identify characteristics that appear to make certain types of document tasks more or less difficult for readers. Questions and directives associated with these tasks are basically of four types: *locating, cycling, integrating,* and *generating.* Locating tasks require the readers to match one or more features of information stated in the question to either identical or synonymous information given in the document. Cycling tasks require the reader to locate and match one or more features, but differ in that they require the reader to engage in a series of feature matches to satisfy conditions given in the question. The integrating tasks typically require the reader to compare and contrast information in adjacent parts of the document. In the generating tasks, readers must produce a written response by processing information found in the document and also making text-based inferences or drawing on their own background knowledge.

² J.T. Guthrie, M. Seifert, and I.S. Kirsch. (1986). "Effects of Education, Occupation, and Setting on Reading Practices." American Educational Research Journal, 23. pp. 151-60.

As with the prose tasks, each type of question or directive extends over a range of difficulty as a result of interactions among several variables or task characteristics that include:

- the number of categories or features of information in the question that the reader has to process or match
- the number of categories or features of information in the document that can serve to distract the reader or that may seem plausible but are incorrect
- the extent to which the information asked for in the question is obviously related to the information stated in the document
- the structure of the document

A more detailed discussion of the five levels of document literacy is provided in the following pages.

Document Level 1

Scale range: 0 to 225

Tasks in this level tend to require the reader either to locate a piece of information based on a literal match or to enter information from personal knowledge onto a document. Little, if any, distracting information is present.

Percentage of adults in the state performing in this level: 16% Percentage of adults in the nation performing in this level: 23%

Some of the Level 1 tasks require the reader to match one piece of

information in the directive with an identical or synonymous piece of information in the document. For example, readers may be asked to write a piece of personal background information — such as their name or age — in the appropriate place on a document. One task with a difficulty value of 69 directs individuals to look at a Social Security card and sign their name on the line marked "signature." Tasks such as this are quite simple, since only one piece of information is required, it is known to the respondent, and there is only one logical place on the document where it may be entered.

Here is a Social Security card. Sign your name on the line that reads "signature."



Other tasks in this level are slightly more complex. For example, in one task, readers were asked to complete a section of a job application by providing several pieces of information. This was more complicated than the previous task described, since respondents had to conduct a series of one-feature matches. As a result, the difficulty value of this task was higher (218).

You have gone to an employment center for help in finding a job. You know that this center handles many different kinds of jobs. Also, several of your friends who have applied here have found jobs that appeal to you.

The agent has taken your name and address and given you the rest of the form to fill out. Complete the form so the employment center can help you get a job.

_ ngc	SCA. IVIAIC	remaie
Taialat	TT	
		Nec Bealth

Last grade completed in school

Kind of work wanted:



Other tasks in this level ask the reader to locate specific elements in a document that contains a variety of information. In one task, for example, respondents were given a form providing details about a meeting and asked to indicate the date and time of the meeting, which were stated in the form. The difficulty values associated with these tasks were 183 and 180, respectively. The necessary information was referred to only once in the document.

Document Level 2

Scale range: 226 to 275

Tasks in this level are more varied than those in Level 1. Some require the reader to match a single piece of information; however, several distractors may be present, or the match may require low-level inferences. Tasks in this level may also ask the reader to cycle through information in a document or to integrate information from various parts of a document.

Percentage of adults in the state performing in this level: 27% Percentage of adults in the nation performing in this level: 28%

Some tasks in Level 2 ask readers to match two pieces of information in the text. For example, one task with a difficulty value of 261 directs the respondent to look at a pay stub and to write "the gross pay for this year to date." To perform the task successfully, respondents must match both "gross pay" and "year to date" correctly. If readers fail to match on both features, they are likely to indicate an incorrect amount.

What is the gross pay for this year to date?

			1200	PERIOD	ENDING	the second second	-			-	-	NET DAY
-	HOURS			03/1	5/85	REGULAR	OVER	TIME	GROSS	DEF. AN	111	NETPAT
REGULAR	2ND SHIFT C		TAL	CURRENT		62500)	1	625:00		1	45988
-		5	00	YEART	O DATE			426885				
500			AX DEDUC	TIONS		2 1			OTHER DED	UCTIONS	-	
-	FED. W/H	BTATE W/H	C	TY W/H	FICA		CR UNION	10	UNITED FD	PERS INS.	MISC.	MISC
CURRENT	10894	137	5		383	1						
YEAR TO DATE	73498	825	0		2616	7			OTHER DEL	UCTIONS	TYPE	AMOUNT
							07	DEN	412			

A second question based on this document — What is the current net pay? — was also expected to require readers to make a two-feature match. Accordingly, the difficulty values of the two items were expected to be similar. The task anchored at about the 200 point on the scale, however, and an analysis of the pay stub reveals why its difficulty was lower than that of the previous task. To succeed on the second task, the reader only needs to match on the feature "net pay." Since the term appears only once on the pay stub and there is only one number in the column, this task requires only a one-feature match and receives a difficulty value that lies within the Level 1 range on the document scale.

Tasks in Level 2 may also require the reader to integrate information from different parts of the document by looking for similarities or differences. For example, a task with a difficulty value of 268 asks respondents to study a line graph showing a company's seasonal sales over a three-year period, then predict the level of sales for the following year, based on the seasonal trends shown in the graph.

You are a marketing manager for a small manufacturing firm. This graph shows your company's sales over the last three years. Given the seasonal pattern shown on the graph, predict the sales for Spring 1985 (in thousands) by putting an "x" on the graph.



Document Level 3

Scale range: 276 to 325

Some tasks in this level require the reader to integrate multiple pieces of information from one or more documents. Others ask readers to cycle through rather complex tables or graphs which contain information that is irrelevant or inappropriate to the task.

Percentage of adults in the state performing in this level: 36% Percentage of adults in the nation performing in this level: 31%

Tasks within the range for Level 3 ask the reader to locate particular features in complex displays, such as tables that contain nested information. Typically, distractor information is present in the same row or column as the correct answer. For example, the reader might be asked to use a table that summarizes appropriate uses for a variety of products, and then choose which product to use for a certain project. One such task had a difficulty value of 305. To perform this task successfully, the respondent uses a table containing nested information to determine the type of sandpaper to buy if one needs "to smooth wood in preparation for sealing and plans to buy garnet sandpaper." This task requires matching not only on more than a single feature of information but also on features that are not always superordinate categories in the document. For example, "preparation for sealing" is subordinated or nested under the category "wood," while the type of sandpaper is under the main heading of "garnet." In addition, there are three other types of sandpaper that the reader might select that partially satisfy the directive.

You need to smooth wood in preparation for sealing and plan to buy garnet sandpaper. What type of sandpaper should you buy?

and the other design of th	-	-		A	BRASI	VE SE	LECT	ION G	UIDE	-			-	-		_		-
MATERIAL & OPERATION	EC	PRO	DUCT	MEFF			C M E EE			ur	WETO	RDRY	Luc	FRE-CUT*		EMERY		
VOOD alint Removal leavy Stock Removal Moderate Stock Removal reparation for Sealing dier Sealer letween Coats dier Final Coat IETAL hust and Paint Removal ight Stock Removal reparation for Priming inishing and Polishing dier Primer letween Coats dier Final Coat LASTIC & FIBERGLASS chaping light Stock Removal inishing & Southing																C		
EC = Extra Coarse C = C SAFETY INFORM/ Wear approved safety when sanding.	oarse ATIOI goggle	M = I N: HS	Mediur Us mea sand	n F se par ins to ding d	e Fine ticle/d preve lust.	VF lust m nt Inhi	= Very ask or alation	Fine other of	EF	= Extra What manu proce	a Fine en usi factur dures	SF and s	super to comma fety	or Fine tols, fo nender instruc	UF : flow d	= Ultra	Fine	

At the same level of difficulty (306), another task directs the reader to a stacked bar graph depicting estimated power consumption by source for four different years. The reader is asked to select an energy source that will provide more power in the year 2000 than it did in 1971. To succeed on this task, the reader must first identify the correct years and then compare each of the five pairs of energy sources given.

Document Level 4

Scale range: 326 to 375

Tasks in this level, like those in the previous levels, ask readers to perform multiple-feature matches, cycle through documents, and integrate information; however, they require a greater degree of inferencing. Many of these tasks require readers to provide numerous responses but do not designate how many responses are needed. Conditional information is also present in the document tasks in this

level and must be taken into account by the reader.

Percentage of adults in the state performing in this level: 19% Percentage of adults in the nation performing in this level: 15%

One task in this level (348) combines many of the variables that contribute to difficulty in Level 4. These include: multiple-feature matching, complex displays involving nested information, numerous distractors, and conditional information that must be taken into account in order to arrive at a correct response. Using the bus schedule shown here, readers are asked to select the time of the next bus on a Saturday afternoon, if they miss the 2:35 bus leaving Hancock and Buena Ventura going to Flintridge and Academy. Several departure times are given, from which respondents must choose the correct one.

On Saturday afternoon, if you miss the 2:35 bus leaving Hancock and Buena Ventura going to Flintridge and Academy, how long will you have to wait for the next bus?



VISTA GRANDE

This bus line operates Monday through Saturday providing "local service" to most neighborhoods in the northeast section. Buses run thirty minutes apart during the morning and afternoon rush hours Monday through Friday. Buses run one hour apart at all other times of day and Saturday.

No Sunday, holiday or night service.

OUT from Terminal	BC		NC)			IN	BOL		to another headed anywhere else in the city bus system		
Leave Downtown Terminal	Leave Hancock and Buena	Leave Citadel	Leave Rustic Hills	Leave North Carefree and Oro Blanco	Arrive Flintridge and Academy	Leave Flintridge and Academy	Leave North Carefree and Oro Blanco	Leave Rustic Hills	Leave Citadel	Leave Hancock and Buena Ventura	Arrive Downtown Terminal	
6:20 6:50 7:20 7:50 8:20 8:50 9:20 10:20 11:20	6:35 7:05 7:35 8:05 8:35 9:05 9:35 10:35 11:35	6:45 7:15 7:45 8:15 8:45 9:15 9:45 10:45 11:45	6:50 7:20 7:50 8:20 8:50 9:20 9:50 10:50 11:50	7:03 7:33 8:03 8:33 9:03 9:33 10:03 11:03 12:03	7:15 7:45 8:15 8:45 9:15 9:45 10:15 11:15 12:15	6:15 6:45 7:15 7:45 8:15 8:45 9:15 9:45 10:15 11:15 12:15	6:27 6:57 7:27 7:57 8:27 8:57 9:27 9:27 9:57 10:27 11:27 12:27	6:42 7:12 7:42 8:12 8:42 9:12 9:42 10:12 10:42 11:42 12:42 p.m.	6:47 7:17 7:47 8:17 8:47 9:17 9:47 10:17 10:47 11:47 12:47 p.m.	6:57 7:27 7:57 8:27 8:57 9:27 9:57 10:27 10:57 11:57 12:57 p.m.	7:15 7:45 Monday through Friday only 8:15 8:45 Monday through Friday only 9:15 9:45 Monday through Friday only 10:15 10:45 Monday through Friday only 11:15 12:15 1:15 p.m.	
12:20 1:20 2:20 2:20 2:20 3:20 3:20 3:20 3:20 3	12:35 1:35 2:35 3:05 3:35 3:35 4:05 4:35 5:05 5:35 6:05 6:35	12:45 1:45 2:45 3:15 3:45 4:15 4:45 5:15 5:45 6:15 6:45	12:50 1:50 2:50 3:20 3:50 4:20 4:50 5:50 6:20 6:50	1:03 2:03 3:03 3:33 4:03 4:33 5:03 5:33 6:03 6:03 6:33 7:03	1:15 2:15 3:15 3:45 4:15 4:45 5:15 5:45 6:15 6:45 7:15	1:15 2:15 3:15 3:45 4:15 4:45 5:15 5:45	1:27 2:27 3:27 3:57 4:27 4:57 5:27 5:57	1:42 2:42 3:42 4:12 4:42 5:12 5:42 6:12	1:47 2:47 3:47 4:17 4:47 4:17 5:47 6:17	1:57 2:57 3:57 4:27 4:57 5:27 5:57 6:27	2:15 3:15 4:15 4:45 Monday through Friday only 5:15 5:45 Monday through Friday only 6:15 6:45 Monday through Friday only Monday through Friday only To be sure of a smooth transfer tall the driver of this bus the name of the second bus you need.	

Other tasks involving this bus schedule are found in Level 3. These tasks require the reader to match on fewer features of information and do not involve the use of conditional information.

Document Level 5

Scale range: 376 to 500

Tasks in this level require the reader to search through complex displays that contain multiple distractors, to make high-level textbased inferences, and to use specialized knowledge.

Percentage of adults in the state performing in this level: 2% Percentage of adults in the nation performing in this level: 3%

A task receiving a difficulty value of 395 involves reading and understanding a table depicting the results from a survey of parents and teachers evaluating parental involvement in their school. Respondents were asked to write a brief paragraph summarizing the results. This particular task requires readers to integrate the information in the table to compare and contrast the viewpoints of parents and teachers on a selected number of school issues.

Using the information in the table, write a brief paragraph summarizing the extent to which parents and teachers agreed or disagreed on the statements about issues pertaining to parental involvement at their school.



Parents and Teachers Evaluate Parental Involvement at Their School					
Do you agree or disagree that ?	Total	Level of School			
		Elementary	Junior High	High School	
Our school does a good job of encouraging parental involvement in sports, arts, and other nonsubject areas			percent agreein	g	
Parents	77	76	74	79	
Teachers	77	73	77	85	
Our school does a good job of encouraging parental involvement in educational areas	73	82	71	64	
Parents	80	84	78	70	
Our school only contacts parents when there is a problem with their child	55	46	62	63	
Teachers	23	18	22	33	
Our school does not give parents the opportunity for any meaningful roles					
Parents	22	18	22	28	
Tanakan	8	8	12	7	

Quantitative literacy

Since adults are often required to perform numerical operations in everyday life, the ability to perform quantitative tasks is another important aspect of literacy. These abilities may seem, at first glance, to be fundamentally different from the types of skills involved in reading prose and documents and, therefore, to extend the concept of literacy beyond its traditional limits. However, research indicates that the processing of printed information plays a critical role in affecting the difficulty of tasks along this scale.³

³ I.S. Kirsch and A. Jungeblut. (1986). Literacy: Profiles of America's Young Adults, Final Report. Princeton, NJ: Educational Testing Service. I.S. Kirsch, A. Jungeblut, and A. Campbell. (1992). Beyond the School Doors: The Literacy Needs of Job Seekers Served by the U.S. Department of Labor. Princeton, NJ: Educational Testing Service.

The quantitative literacy scale contains some 39 tasks with difficulty values that range from 191 to 436. The difficulty of these tasks appears to be a function of several factors, including:

- the particular arithmetic operation called for
- the number of operations needed to perform the task
- the extent to which the numbers are embedded in printed materials
- the extent to which an inference must be made to identify the type of operation to be performed

In general, it appears that many individuals can perform simple arithmetic operations when both the numbers and operations are made explicit. However, when the numbers to be used must be located in and extracted from different types of documents that contain similar but irrelevant information, or when the operations to be used must be inferred from printed directions, the tasks become increasingly difficult.

A detailed discussion of the five levels of quantitative literacy is provided on the following pages.

Quantitative Level 1

Scale range: 0 to 225

Tasks in this level require readers to perform single, relatively simple arithmetic operations, such as addition. The numbers to be used are provided and the arithmetic operation to be performed is specified.

Percentage of adults in the state performing in this level: 15% Percentage of adults in the nation performing in this level: 22%

The least demanding task on the quantitative scale (191) requires the reader to total two numbers on a bank deposit slip. In this task, both the numbers and the arithmetic operation are judged to be easily identified and the operation involves the simple addition of two decimal numbers that are set up in column format.



You wish to use the automatic teller machine at your bank to make a deposit. Figure the total amount of the two checks being deposited. Enter the amount on the form in the space next to TOTAL.

Availability of Deposits

Funds from deposits may not be available for immediate withdrawal. Please refer to your institution's rules governing funds availability for details.

Crediting of deposits and payments is subject to verification and collection of actual amounts deposited or paid in accordance with the rules and regulations of your financial institution.

OUR MAC CARD NUMBER (No PINS PLEASE)	CASH	\$	00
111 222 3334	LIST CHECKS BY BANK NO.	ENDORSE WITH NAME & ACCOUNT NUMBER	
OUR FINANCIAL INSTITUTION Union Bank		557	19
OUR ACCOUNT NUMBER 987 555 674		75	00
Chris Jones			
CHECK ONE DEPOSIT or PAYMENT	TOTAL		

Quantitative Level 2

Scale range: 226 to 275

Tasks in this level typically require readers to perform a single operation using numbers that are either stated in the task or easily located in the material. The operation to be performed may be stated in the question or easily determined from the format of the material (for example, an order form).

Percentage of adults in the state performing in this level: 22% Percentage of adults in the nation performing in this level: 25%

In the easier tasks in Level 2, the quantities are also easy to locate. In one such task at 246 on the quantitative scale, the cost of a ticket and bus is given for each of two shows. The reader is directed to determine how much less attending one show will cost in comparison to the other.

The price of one ticket and bus for "Sleuth" costs how much less than the price of one ticket and bus for "On the Town"?

THEATER TRIP

A charter bus will leave from the bus stop (near the Conference Center) at 4 p.m., giving you plenty of time for dinner in New York. Return trip will start from West 45th Street directly following the plays. Both theaters are on West 45th Street. Allow about 1½ hours for the return trip.

Time:	4 p.m., Saturday, Novem	ber 20	
Price:	"On the Town"	Ticket and bus	\$11.00
	"Sleuth"	Ticket and bus	\$8.50
Limit:	Two tickets per person		

In a more complex set of tasks, the reader is directed to complete an order form for office supplies using a page from a catalogue. No other specific instructions as to what parts of the form should be completed are given in the directive. One task (difficulty value of 270) requires the reader to use a table on the form to locate the appropriate shipping charges based on the amount of a specified set of office supplies, to enter the correct amount on an order form, and then to calculate the total price of the supplies.

Quantitative Level 3

Scale range: 276 to 325

In tasks in this level, two or more numbers are typically needed to solve the problem, and these must be found in the material. The operation(s) needed can be determined from the arithmetic relation terms used in the question or directive.

Percentage of adults in the state performing in this level: 36% Percentage of adults in the nation performing in this level: 31%
In general, tasks within the range for Level 3 ask the reader to perform a single operation of addition, subtraction, multiplication, or division. However, the operation is not stated explicitly in the directive or made clear by the format of the document. Instead, it must be inferred from the terms used in the directive. These tasks are also more difficult because the reader must locate the numbers in various parts of the document in order to perform the operation.

From a bar graph showing percentages of population growth for two groups across six periods, a task at the 278 point on the scale directs the reader to calculate the difference between the groups for one of the years.

A more difficult task in Level 3 (321) requires the use of a bus schedule to determine how long it takes to travel from one location to another on a Saturday. To respond correctly, the reader must match on several features of information given in the question to locate the appropriate times.

Suppose that you took the 12:45 p.m. bus from U.A.L.R. Student Union to 17th and Main on a Saturday. According to the schedule, how many minutes is the bus ride?

Section V 163

16		So	uth High	land	1(
-		Going	TOWARD Do	www.town	Can the
					O CAPITOL
				DOWNTOWN LITTLE ROCK	HTBO
					NIN
		S ELI	EMENTARY	DUNBAR JR. HIGH	
	RISO	EMIS			1717H
	HAR	91-00 - 20TH	9/-	WRIGHT &	
		22ND YO		HORY	Governor's) /
	Catherine and	Will-CS			7 1
	Alledina Pdg	WA	dia	/	NORTH
TEAVE	E STATE	Arkansas Enterprises	1,	1	
LEAVE	R	for the Blind	1	1	10
		4	6	6	6
		BUS LEAVES	Bus arrives	Bus arrives	BUS ENDS
		from	at	at 17th P	at Conitol P
		Student Union	Woodrow	Main	Louisiana
-	-	Student Onion	Woodrow	Within	Louisiana
		and the second second	WEEKDAYS		
A.M.	é.	5:38	5:51	6:00	6:09
	1	6:41	6:55	7:05	7:15
	6	7:11	7:25	7:35	7:45
	E	7:41	7:55	8:05 8:35	8:15 8:45
	¢.	8:41	8:55	9:05	9:15
		9:14	9:27	9:36	9:45
	Ġ.	10:14	10:27	10:36	10:45
	Ġ.	10:44	10:57	11:06	11:15
	,	11:44	11:57	12:06	12:15
P.M.	G	12:14	12:27	12:36	12:45
	e.	12:44	12:57	1:36	1:15
	6	1:44	1:57	2:06	2:15
		2:14	2:27	2:36	3:15
	6	3:14	3:27	3:36	3:45
	6	3:43	3:56	4:05	4:15
	A	4:43	4:56	5:05	5:15
	a	5:13	5:26	5:35	5:45
	6	6:11	6:22	6:30	0.11
	6	6:46	6:57	7:05	-
			SATURDAY		
A.M.	6	5:38	5:51	6:00	6:09
	6	6:45	6:57	7:06	7:15
	6	7:45	7:57	8:06	8:15
	6	8:45	8:57	9:06	9:15
	6	9:45	9:57	11:06	11:15
	6	11:45	11:57	12:06	12:15
-	Ĩ	10.15	10.57	1.00	1.15



Quantitative Level 4

Scale range: 326 to 375

These tasks tend to require readers to perform two or more sequential operations or a single operation in which the quantities are found in different types of displays, or the operations must be inferred from semantic information given or drawn from prior knowledge.

Percentage of adults in the state performing in this level: 23% Percentage of adults in the nation performing in this level: 17%

One task in this level, with a difficulty value of 332, asks the reader to estimate, based on information in a news article, how many miles per day a driver covered in a sled-dog race. The respondent must know that to calculate a "per day" rate requires the use of division.

A more difficult task (355) requires the reader to select from two unit price labels to estimate the cost per ounce of creamy peanut butter. To perform this task successfully, readers may have to draw some information from prior knowledge.

Estimate the cost per ounce of the creamy peanut butter. Write your estimate on the line provided.





Section V 165

Quantitative Level 5

Scale range: 376 to 500

These tasks require readers to perform multiple operations sequentially. They must disembed the features of the problem from text or rely on background knowledge to determine the quantities or operations needed.

Percentage of adults in the state performing in this level: 4% Percentage of adults in the nation performing in this level: 4%

One of the most difficult tasks on the quantitative scale (433) requires readers to look at an advertisement for a home equity loan and then, using the information given, explain how they would calculate the total amount of interest charges associated with the loan.

You need to borrow \$10,000. Find the ad for Home Equity Loans on page 2 in the newspaper provided. Explain to the interviewer how you would compute the total amount of interest charges you would pay under this loan plan. Please tell the interviewer when you are ready to begin.



Estimating Performance Across the Literacy Levels

The literacy levels not only provide a way to explore the progression of information-processing demands across the scales; they can also be used to explore the likelihood that individuals in each level will succeed on tasks of varying difficulty.

The following graphs (Figure 5.2) display the probability that individuals performing at selected points on each scale will give a correct response to tasks with varying difficulty values. We see, for example, that a person whose prose proficiency is 150 has less than a 50 percent chance of giving a correct response to the Level 1 tasks. Individuals whose proficiency scores were at the 200 point, on the other hand, have an almost 80 percent probability of responding correctly to these tasks.

In terms of task demands, we can infer that adults performing at the 200 point on the prose scale are likely to be able to locate a single piece of information in a brief piece of text where there is no distracting information, or when any distracting information is located apart from the desired information. They are likely to have far more difficulty with the types of tasks that occur in Levels 2 through 5, however. For example, they would have only about a 30 percent chance of performing the average task in Level 2 correctly and only about a 10 percent chance of success, or less, on the more challenging tasks found in Levels 3, 4, and 5.

In contrast, readers at the 300 point on the prose scale have an 80 percent (or higher) likelihood of success on tasks in Levels 1, 2, and 3. This means that they demonstrate skill identifying information in fairly dense text without organizational aids. They can also integrate, compare, and contrast information that is easily identified in the text. On the other hand, they are likely to have difficulty with tasks that require them to make higher-level inferences, to take conditional information into account, and to use specialized knowledge. The probabilities of their performing these Level 4 tasks successfully are just under 50 percent, and on the Level 5 tasks their likelihood of responding correctly falls to under 20 percent.

Section V 167

NALS

Average Probabilities of Successful Performance by Individuals with Selected Proficiency Scores on the Tasks in Each Literacy Level



QUANTITATIVE



Similar interpretations can be made using the performance results on the document and quantitative scales. For example, an individual with a proficiency of 150 on the quantitative scale is estimated to have only a 50 percent chance of responding correctly to tasks in Level 1 and less than a 30 percent chance of responding to tasks in each of the other levels. Such an individual demonstrates little or no proficiency in performing the range of quantitative tasks found in this assessment. In contrast, someone with a proficiency of 300 meets or exceeds the 80 percent criterion for the average tasks in Levels 1, 2, and 3. They can be expected to encounter more difficulty with tasks in Levels 4 and 5.

Section V 169

APPENDICES



-



APPENDIX A

Variable Definitions [in order of presentation]

State

The state sample includes state residents age 16 to 64 who participated in the State Adult Literacy Survey as well as state residents age 16 and older who participated in the National Adult Literacy Survey. The two samples are combined to increase the numbers of adults in various population groups and thus provide more robust estimates of literacy proficiencies.

Region

Census definitions of regions are used in the National and State Adult Literacy Surveys. The four regions analyzed are the Northeast, Midwest, South, and West. The states in each region are identified below.

- Northeast: Maine, New Hampshire, Vermont, Massachusetts, Rhode Island, Connecticut, New York, New Jersey, Pennsylvania
- Midwest: Ohio, Indiana, Illinois, Michigan, Wisconsin, Minnesota, Iowa, Missouri, North Dakota, South Dakota, Nebraska, Kansas
- South: Delaware, Maryland, District of Columbia, Virginia, West Virginia, North Carolina, South Carolina, Georgia, Florida,



Kentucky, Tennessee, Alabama, Mississippi, Arkansas, Louisiana, Oklahoma, Texas

West: Montana, Idaho, Wyoming, Colorado, New Mexico, Arizona, Utah, Nevada, Washington, Oregon, California, Alaska, Hawaii

The regional samples encompass adults who participated in the state and national surveys, including individuals living in households and those in prison.

Nation

The national sample includes adults age 16 and older who participated in the national household survey, the state surveys, and the survey of prisoners.

Age

All survey respondents were asked to report their birthdates, and this information was used to calculate their ages. Typically, the age groups reported

Appendices 173

are: 16 to 18, 19 to 24, 25 to 39, 40 to 54, 55 to 64, and 65 and older. For some analyses, the ages are grouped differently. Because adults age 65 and older were not included in the State Adult Literacy Survey, the state results for adults in the 65 and older age group are based only on those state residents who participated in the national survey. These results may not be representative and should therefore be interpreted with caution.

Country of Birth

All survey respondents were asked to indicate whether they were born in the United States (50 states or Washington, D.C.), a U.S. territory, or another country. Based on their responses, they were divided into two groups: adults born in this country or a United States territory, and those born in another country.

Years Lived in the United States

Survey respondents who were born in a U.S. territory or in another country were asked how many years they had lived in the United States: 1 to 5, 6 to 10, 11 to 15, 16 to 20, 21 to 30, 31 to 40, 41 to 50, 51 or more. They were divided into three groups: adults who had lived in the United States for 1 to 5 years, for 6 to 10 years, and for more than 10 years.

Race/Ethnicity

All survey respondents were asked two questions about their race and ethnicity. One question asked them to indicate which of the following best describes them. The interviewer recorded the races of respondents who refused to answer the question.

White	
Black (African American)	
American Indian	
Alaskan Native	

Pacific Islander Asian Other

The other question asked respondents to indicate whether they were of Spanish or Hispanic origin or descent. Those who responded "yes" were asked to identify which of the following groups best describes their Hispanic origin:

> Mexicano, Mexican, Mexican American, Chicano Puerto Rican Cuban Central/South American

Other Spanish/Hispanic

Adults of Pacific Islander origin were grouped with those of Asian origin, and Alaskan Natives, American Indians, and Other adults are grouped together, due to the small sample sizes. All other racial/ethnic groups are reported separately.

174 Variable Definitions

In some analyses, however, the Latino subpopulations are combined to provide reliable estimates. The race/ethnicity categories are mutually exclusive.

Number of Years Lived in Iowa

Survey respondents in Iowa were asked how many years they have lived in Iowa. They were given the following response options: less than one year, 1 to 5 years, 6 to 10 years, 11 to 15 years, 16 to 20 years, more than 20 years.

Likelihood of Moving Out of Iowa

Survey respondents in Iowa were asked how likely it is that they will move out of Iowa in the next five years: not likely, somewhat likely, or very likely.

Type of Physical, Mental, or Other Health Condition

All survey respondents were asked to identify whether they have a physical, mental, or other health condition that keeps them from participating fully in work, school, housework, or other activities.

Sex

The interviewers recorded the sex of each respondent.

Level of Education Attained in the United States

All survey respondents were asked to indicate the highest level of education they completed in this country. The following options were given:

Still in high school
Less than high school
Some high school
GED or high school equivalency
High school graduate
Vocational, trade, or business school after high school
College: less than 2 years
College: associate's degree (A.A.)
College: 2 or more years, no degree
College graduate (B.S. or B.A.)
Postgraduate, no degree
Postgraduate degree (M.S., M.A., Ph.D., M.D., etc.)

For certain analyses, some of these groups were collapsed. For example, respondents who had completed postgraduate studies but no degree were generally combined with those who had completed a postgraduate degree.

Average Years of Schooling

Responses to the question on the highest level of education attained in the United States were used to calculate the average number of years of schooling attained. Individuals who were still in high school at the time of the survey were left out of this analysis. Adults who had not graduated from high school were asked to indicate exactly how many years of schooling they had completed (0 through 12). Individuals who did not provide this information were assigned a value equal to the average number of years of schooling completed by those who did provide the information. For adults in the category "0 to 8 years of education," the average number of years of schooling was 6.10, and for adults in the category "9 to 12 years of education," the average was 10.11. The remaining adults were assigned values representing the number of years of schooling completed, as follows:

GED, high school equivalency	12
High school graduate	12
Vocational, trade, or business school	13
College: less than 2 years	13
College: associate's degree (A.A.)	14
College: 2 or more years, no degree	14.5
College graduate (B.S. or B.A.)	16
Postgraduate, no degree	17
Postgraduate degree	18

Using these values, the average number of years of schooling was calculated for various reporting groups (such as age, race/ethnicity, and sex).

Parents' Level of Education

All survey respondents were asked to indicate the highest level of education completed by their mother (or stepmother or female guardian) and by their father (or stepfather or male guardian). The response options provided were identical to those provided in the question about respondents' own level of education. A new variable was then constructed, reflecting the highest level of education attained by either parent.

Highest Level of Education Completed Before Coming to the United States

Survey respondents who were born in a United States territory or in another country were asked to indicate the highest level of education they had completed before coming to the United States.

Participation in a GED or High School Equivalency Program Survey respondents who did not graduate from high school (and were not still in high school) were asked if they had ever studied for a GED or high school equivalency. Combined with their responses to the question about the highest level of education they had completed in the United States, their responses

176.... Variable Definitions

were used to create two new variables: one reflecting whether or not they had ever studied for a GED, and another indicating whether program participants had actually earned their diplomas.

Current Educational Enrollment and Goals

Household survey respondents except those still in high school were asked whether they were currently enrolled in school or college either full time or part time. Those who were enrolled were asked what diploma, certificate, degree, or accreditation they expected to earn: a high school diploma or equivalency; vocational, trade, or business; two years of college (associate's degree); four- or five-year college degree (B.S., B.A.); Master's, Ph.D., M.D., or other advanced degree; other; or none.

Enrollment in a Basic Skills Program

All survey respondents were asked whether they were currently enrolled in or had ever taken part in a program other than regular school in order to improve their basic skills - that is, basic reading, writing, and arithmetic skills.

Most Important Reason for Not Taking Part in a Basic Skills Program Survey respondents in Iowa were asked which one reason would be the most important one for them not to take part in a basic skills program. They were given the following list of statements and asked to choose one: I don't need to improve my basic skills; I am too old to go back to school; I think school is too hard; I don't have the time; I don't like school; I have too many conflicts; It would take too long to finish a basic skills program; I don't have any information about available basic skills programs.

Opinion as to the Effect of a State's Literacy Rate on Employer Decisions

Survey respondents in Iowa were asked whether they feel that a state's literacy rate affects an out-of-state employer's decision to establish a new location there. They were given the following response options: yes, no, no opinion.

Opinion as to Employer Obligation to Provide Literacy Education

Survey respondents in Iowa were asked whether they feel that an employer has an obligation to provide literacy education to its employees who need assistance. They were given the following response options: yes, no, no opinion.

Labor Force Status

Household survey respondents were asked what they were doing the week before the survey:

- 1) working at a full-time job for pay or profit (35 hours or more)
- 2) working two or more part-time jobs for pay, totaling 35 or more hours
- 3) working for pay or profit part time (1 to 35 hours)
- 4) unemployed, laid off, or looking for work
- 5) with a job but not at work
- 6) with a job but on family leave (maternity or paternity leave)
- 7) in school
- 8) keeping house
- 9) retired
- 10) doing volunteer work

They were then divided into four groups: adults working full time (or working two or more part-time jobs); those working part time; those unemployed, laid off, or looking for work; and those out of the labor force. Adults in categories 1 and 2 above were counted as being employed full time; those in category 3 were counted as being employed part time; those in category 4 were counted as unemployed; those in categories 5 and 6 were counted as being not at work (and therefore omitted from the analyses); and those in categories 7 through 10 were counted as being out of the labor force.

Occupational Category

All survey respondents were asked two questions about their current or most recent jobs, whether full time or part time. The first question asked them to identify the type of business or industry in which they worked — for example, television manufacturing, retail shoe store, or farm. The second question asked them to indicate their occupation, or the name of their job — for example, electrical engineer, stock clerk, typist, or farmer. Their responses were used to create four occupational categories: professional, management, or technical; sales or clerical; craft or service; and labor, assembly, fishing, or farming.

Average Number of Weeks Worked

Household survey respondents (including those unemployed or out of the labor force the week before the survey) were asked to indicate how many weeks they had worked for pay or profit during the past 12 months, including paid leave (such as vacation and sick leave).

Median Weekly Wages

178 Variable Definitions

Household survey respondents who were employed or on leave the week before the survey were asked to report their average wages or salaries (including tips and commissions) before deductions. They reported their wages or salaries per hour, day, week, two-week period, month, year, or other unit of time, and these data were used to calculate their weekly wages. The median, rather than the arithmetic mean, is used in these analyses due to the wide variability in wages among adults at the lowest and highest literacy levels.

Median Annual Household Income

Household survey respondents were asked to indicate their family's total income from all sources in 1991. They were instructed to consider as family anyone who lives with them and is related by blood, marriage, or adoption.

Sources of Nonwage Income and Support

Household survey respondents were asked to indicate which of the following types of income and support they or anyone in their family received during the past 12 months: Social Security, Supplemental Security Income, retirement payments, Aid to Families with Dependent Children, food stamps, interest from savings or other bank accounts, dividend income, and income from other sources. Each source was treated as a separate variable, and respondents were divided into two groups: those who had received this type of income or support, and those who had not. This report analyzes results for adults who reported receiving food stamps, Aid to Families with Dependent Children (or public assistance), and interest from savings.

Poverty Status

Household survey respondents were asked to report the number of persons living in their households as well as the family's total income from all sources during the previous calendar year. Their responses to these two questions were used to construct the poverty status variable. Based on the 1991 poverty income thresholds of the federal government, the following criteria were used to identify respondents who were poor or near poor:

Respondents whose	And whose annual household
family size was:	income was at or below:
1	\$ 8,665
2	\$11,081
3	\$13,575
4	\$17,405
5	\$20,570
6	\$23,234
7	\$26,322
8	\$29,506
9	\$34,927

Appendices 179

Voting

Household survey respondents were asked whether or not they had voted in a national or state election in the past five years. Some participants reported being ineligible to vote, and they were excluded from the analyses. The results reported herein reflect the percentages of adults who voted, of those who were eligible to vote.

Language Learned Before Starting School

All survey respondents were asked what language or languages they had learned to speak before they started school: English, Spanish, or Other. Their responses were used to divide respondents into three groups: those who spoke English only, those who spoke English and Spanish or another language, and those who spoke Spanish or another language only.

Language Usually Spoken Now

Survey respondents who had learned to speak a language other than English before starting school (instead of or in addition to English) were asked what language they usually speak now: English, Spanish, or Other.

Use of English or Another Language in Various Contexts

Survey respondents who had learned to speak a language other than English before starting school (either instead of or in addition to English) were asked what language they use in the following situations: at home, at work, while shopping in their neighborhoods, and when visiting relatives or friends. The options given were: always English, more English than another language, English and another language equally, more another language than English, or always another language. These were collapsed into the following categories: always English, sometimes a non-English language, and always a non-English language.

Self-reported English Literacy

All survey respondents were asked four questions about their English literacy skills, concerning how well they speak, understand, read, and write English. Four response options were given: very well, well, not well, and not at all. These were combined into two categories: "very well or well" and "not well or not at all."

Reliance on Various Sources of Information

Household survey respondents were asked how much information about current

events, public affairs, and the government they usually get from newspapers, magazines, radio, television, and family members, friends, or coworkers. The



responses to these questions were used to construct a new variable that reflects the extent to which adults get information from different sources:

Print media: Adults who get "some" or "a lot" of information from either newspapers or magazines, and those who do not
Nonprint media: Adults who get "some" or "a lot" of information from either television or radio, and those who do not
Personal sources: Adults who get "some" or "a lot" of information from family, friends, or coworkers, and those who do not

Frequency of Newspaper Reading

All survey respondents were asked how often they read a newspaper in English: every day, a few times a week, once a week, less than once a week, or never.

Aspects of Newspaper Reading

All survey respondents were given a list of different parts of the newspaper and asked to identify which parts they generally read. A long list of parts was given, and these were grouped as follows:

news, editorial pages, financial news and stock listings home, fashion, and health sections, and book, movie, or art reviews classified ads, other ads, and TV, movie, or concert listings comics, horoscopes or advice columns sports

The responses to this question and the prior question on the frequency of newspaper reading were then combined to determine the percentage of newspaper readers (that is, of adults who read the newspaper at least once a week) who read various parts.

Magazine Reading Practices

All survey respondents were asked how many different magazines they look at or read in English on a regular basis: 0, 1, 2, 3 to 5, or 6 or more.

Book Reading Practices

All survey respondents were asked what types of books they had read in English in the past six months, if any. They were given the following options and instructed to code all that apply:

fiction
recreation or entertainment
current affairs or history
inspiration or religion
science or social science
reference, such as encyclopedias or dictionaries

Appendices 181

manuals for cooking, operating, repairing, or building any other types of books none

In addition to analyzing the results for each type of book, we created a second variable which indicated whether respondents had read at least one book (coding any response option except "none") or had not read any books (coding "none").

Frequency of Library Use

Household survey respondents were asked how often they use the services of a library, for any reason: daily, weekly, monthly, once or twice a year, or never.

Amount of Television Watched

Household survey respondents were asked how many hours they watch television each day: none, 1 hour or less, 2 hours, 3 hours, 4 hours, 5 hours, or 6 hours or more.

Personal and Job-related Use of Prose Materials and Documents

Household survey respondents were given a list of prose materials (letters or memos; reports, articles, magazines, or journals) and documents (manuals or reference books, including catalogs or parts lists; directions or instructions for medicines, recipes, or other products; diagrams or schematics; bills, invoices, spreadsheets, or budget tables) and asked how often they used each type for personal reading, job-related reading, personal writing, and job-related writing: every day, a few times a week, once a week, less than once a week, and never. These questions were used to construct four new variables:

personal or job-related reading of prose materials personal or job-related writing of prose materials personal or job-related reading of documents personal or job-related writing of documents

Personal Use of Mathematics

Household survey respondents were asked how often they use arithmetic or mathematics (that is, add, subtract, multiply, divide, or measure) for their own use: every day, a few times a week, once a week, less than once a week, or never.



APPENDIX B



Technical Notes

his appendix provides information about the methods and procedures used in the State and National Adult Literacy Surveys. The forthcoming technical report will provide more extensive information.

Sampling

Sampling activities for the State and National Adult Literacy Surveys were conducted by Westat, Inc., under a subcontract with Educational Testing Service.

The sampling for these surveys included three components: a national household sample; 11 individual state household samples; and a national prison sample. The national and state household components were based on a fourstage stratified area sample. The first stage involved the selection of primary sampling units, consisting of counties or groups of counties; the second stage involved the selection of segments consisting of Census blocks or groups of blocks; the third stage involved the selection of households; and the fourth stage involved the selection of age-eligible individuals. In all, 12 area samples were drawn: one national area sample for the national component, and 11 independent, state-specific area samples for the 11 states that participated in the state component (California, Illinois, Indiana, Iowa, Louisiana, New Jersey, New York, Ohio, Pennsylvania, Texas, Washington). The sample designs used for all 12 samples were similar, except for two principal differences. In the national sample, African American and Latino respondents were sampled at a higher rate than the remainder of the population in order to increase their representation in the sample, whereas the state samples used no oversampling. Also, the target population for the national sample consisted of adults 16 years of age or older, whereas the target population for the state samples consisted of adults 16 to 64 years of age.

Each of the four stages of the sampling process addressed a finer level of geographic detail than the preceding stage. In the first stage, primary sampling

units (counties or groups of counties) were selected. These were stratified on the basis of region, metropolitan status, percent African American, percent Latino, and, whenever possible, per capita income. In the national household survey, 101 primary sampling units (PSUs) were used. The national frame was also used to construct individual state frames for the state household survey. Eight to 12 PSUs were selected within each state that participated in the state survey. All PSUs were selected with probability proportional to the PSU's 1990 population.

In the second stage of sampling, segments consisting of Census blocks or groups of blocks were sampled within the selected PSUs. The segments were selected with probability proportional to their size, where the measure of size was a function of the number of year-round housing units within the segment. The oversampling of African American and Latino respondents for the national component was carried out at the segment level. Accordingly, segments were classified as high minority (segments with more than 25 percent African American or Latino population) or not high minority. The measure of size for high minority segments was defined as the number of White non-Latino households plus three times the number of African American or Latino households. High minority segments were therefore oversampled at up to three times the rate of comparable, non-high-minority segments. The measure of size for non-minority segments was simply the number of year-round housing units within the segment.

One in seven of the national survey segments was selected at random to be included in a "no incentive" sample. Respondents from the remaining segments in the national survey received a monetary incentive for participation, as did respondents in the state survey.

The third stage of sampling involved the selection of households within the selected segments. Westat interviewers canvassed all selected segments and prepared lists of all housing units within the boundaries of each segment as determined by the 1990 Census block maps. The lists were used to construct the sampling frame for households. Households were selected with equal probability within each segment, except for White non-Latino households in high minority segments in the national component, which were subsampled so that the sampling rates for White non-Latino respondents would be about the same overall.

The fourth stage of sampling involved the selection of one or two adults within each selected household. A list of age-eligible household members (16 and older for the national component, 16 to 64 for the state component) was constructed for each selected household. One person was selected at random from households with fewer than four eligible members; two persons were

184 Technical Notes

selected from households with four or more eligible members. The interviewers were instructed to list the eligible household members in descending order by age, then to select the one or two household member(s) to interview, as specified on computer-generated sampling messages attached to each questionnaire.

Sampling in this State

The following Iowa counties made up the primary sampling units selected for participation in either the State Adult Literacy Survey or the National Adult Literacy Survey.

> Black Hawk County **Bremer County Clinton County** Dallas County **Delaware** County **Fayette County** Hamilton County Harrison County Jackson County Jasper County Johnson County Jones County

Kossuth County Monona County Mitchell County Polk County **Poweshiek County** Shelby County Warren County Webster County Winnebago County Woodbury County Worth County

The Data Collection Instruments

Three types of data collection instruments were used in the national and state surveys: the household screener (used to enumerate household members and select survey respondents), the background questionnaires (household and prison), and the literacy exercise booklets. These instruments are described below.

Screener

The screener was used to collect the names, relationships, sex, age, and race/ ethnicity of all household members at the selected dwelling unit. For the national sample, household members age 16 years and older were eligible for selection. For the state sample, household members 16 to 64 years of age were eligible. The procedures described earlier (see Sampling) were used to select eligible participants.

Appendices 185

Background Questionnaires

One of the primary goals of this survey is to relate the literacy skills of the nation's adults to a variety of demographic characteristics and explanatory variables. Accordingly, survey respondents were asked to complete background questionnaires designed to gather information on their characteristics and experiences. The background questionnaires required approximately 20 minutes to complete. To ensure standardized administration, the questionnaires were read to the respondent by trained interviewers. The background questionnaire could be conducted in English or Spanish only.

As recommended by the Literacy Definition Committee that guided the National Adult Literacy Survey, the development of the background questionnaire was guided by two goals: to ensure the usefulness of the data by addressing issues of concern, and to ensure comparability with the young adult and Department of Labor job-seeker surveys by including some of the same questions. With these goals in mind, the background questionnaire addressed the following areas:

- general and language background
- educational background and experiences
- political and social participation
- labor force participation
- literacy activities and collaboration
- demographic information

186 Technical Notes

In addition to these questions, the household background questionnaire included a small set of questions asked only of respondents in the state samples. Each of the 11 states that participated in the State Adult Literacy Survey developed five state-specific questions of particular interest to state decision makers, and these were printed at the end of the questionnaire. The state-specific questions gathered information on topics such as the following:

- length of residency in the state and primary reason for moving there
- likelihood of moving out of the state in the next five years
- levels of schooling completed in the state
- type of adult education best suited to personal needs
- factors that affect participation in a course or training program

- reasons for being denied a job or promotion
- training needs for enhanced job productivity
- employers' responsibility for providing literacy education
- home support for reading and education

Exercise Booklets

A total of 26 different exercise booklets were prepared for the survey, each with a corresponding interview guide, which the interviewer used to facilitate the respondent's completion of tasks in the booklet. Each booklet consisted of three sections, and every respondent was asked to complete one booklet. This required approximately 45 minutes.

The State and National Adult Literacy Surveys measure literacy along three scales — prose, document, and quantitative — composed of literacy tasks that simulate the types of demands that adults encounter in everyday life. In all, 166 literacy tasks were administered in this survey, including 81 new tasks and 85 tasks that were administered in the previous young adult and job-seeker surveys. The administration of a common pool of tasks in each of the three surveys allows for valid comparisons of results across time for different populations.

The new literacy tasks developed for the survey serve to refine and extend the three existing literacy scales and provide a better balance of tasks across the scales. The framework used to develop these tasks reflects research on the processes and strategies that respondents used to perform the literacy tasks administered in the young adult survey. In creating the new tasks, one goal was to include diverse materials and to frame questions and directives that represent a broad range of skills and processes. Another goal was to reflect the kinds of reading, writing, and computational demands that adults encounter in work, community, and home settings. Because the tasks are meant to simulate real-life literacy activities, they are open-ended — that is, individuals must produce a written or oral response, rather than simply choose the correct response from a list of options.

The new literacy tasks were developed with attention to the following elements:

- the structure of the stimulus material for example, exposition, narrative, table, graph, map, or advertisement
- the content represented and/or the context from which the stimulus is drawn — for example, work, home, or community

Appendices 187

 the nature of what the individual is asked to do with the material — that is, the purpose for using the material — which in turn guides the strategies needed to complete the task successfully

These factors, operating in various combinations, affect the difficulty of a task relative to others administered in the survey.

The printed and written materials selected for the survey reflect a variety of structures and formats. After these materials were selected, accompanying tasks were developed. The tasks were designed to simulate the way in which people use various types of materials and to require different strategies for successful performance.

Survey Design: Balanced-Incomplete-Block Spiraling

No individual could be expected to respond to the entire set of 166 simulation tasks administered as part of the survey. Accordingly, the survey design gave each respondent a subset of the total pool of literacy tasks, while at the same time ensuring that each of the 166 tasks was administered to a nationally representative sample of the adult population. Literacy tasks were assigned to blocks or sections that could be completed in about 15 minutes, and these blocks were then compiled into booklets so that each block appeared in each position (first, middle, and last) and each block was paired with every other block. Thirteen blocks of simulation tasks were assembled into 26 booklets, each of which could be completed in about 45 minutes. During a personal interview, each participant was asked to complete one booklet of literacy tasks and to respond to the background questionnaire, which required approximately 20 minutes.

Training the Data Collection Staff

The field staff who would be responsible for conducting the state and national surveys was recruited and trained in January and February of 1992 by Westat, Inc. In total, this field staff consisted of 24 supervisors, 24 editors, and 421 interviewers. Supervisors and interviewers were trained first, during a seven-day program in Bethesda, Maryland. Supervisors also received additional training in various areas specific to their managerial responsibilities, including the use of Westat's Automated Survey Control System, a computer-based system for managing the data collection effort. Finally, supervisors and editors were trained to perform an item-by-item edit for each data collection instrument completed by the field interviewers.

188.... Technical Notes

After the centralized training session in Bethesda, interviewers attended a regional training session in either San Francisco or Dallas. At these sessions, four training groups were formed, each led by a Westat home office field manager. The trainees in each group were then divided into "learning communities," each consisting of approximately 18 interviewers. Each community was led by the field supervisor who would supervise the interviewers during the data collection phase.

The training program was closely modeled after Westat's general approach to training field staff. This approach uses a mix of techniques to present study material and focuses heavily on trainee participation and practice. Verbatim scripts and a detailed agenda were used to ensure comparability in training across the groups.

The majority of training time was devoted to instructions for administering the data collection instruments: the household screener, the background questionnaire, and the interview guide and literacy exercise booklet. Instructional materials on gaining respondent cooperation, keeping records of nonresponse cases, editing completed work, and completing administrative forms were also presented. A bilingual field supervisor trained Spanishspeaking interviewers on the Spanish translations of the screener and background questionnaires.

Interviewers without previous experience attended an additional one-half day of training on general interviewing techniques prior to the project-specific training, Interviewers chosen for the prison survey received an additional day of training on interview procedures unique to that sample.

Administering the Data Collection Instruments

The data collection effort began immediately after training was completed. Field supervisors assigned cases to the interviewers and mailed letters to sampled households about one week before the interviewers planned to contact them. Interviewers were given a call record folder and screener for each sampled dwelling unit assigned to them. A computer-generated label attached to the front of each folder and screener provided the case identification number, address, and assigned exercise booklet number. Interviewers were also given all other field materials necessary for them to conduct their interviews and meet reporting requirements.

For each household assigned, the interviewer first verified that the address was in the sample and the unit was an occupied dwelling. If the interviewer was unable to complete a screener at an assigned address, she or he documented the reasons in a non-interview report form. Upon contacting a sampled household that met the basic criteria, the interviewer introduced the study using a statement printed on the front of the screener and indicated that if someone from the household was selected for an interview, the respondent would be paid \$20 for participating. The interviewer then conducted the screening interview with any household member 16 years of age or older. If the household members spoke only a language other than Spanish or English, the interviewer could obtain the services of a translator to complete the screener interview. Once the screener was completed and a respondent or respondents were selected, the interviewer administered the background questionnaire and assigned exercise booklet. If the selected respondent was not available at the time the screener was conducted, the interviewer returned to administer the background questionnaire and exercise booklet.

The background questionnaire was completed first, and then the interviewer administered the exercise booklet. During the administration of the exercise booklet, the interviewer was required to create the proper setting — that is, ensure sufficient lighting and table space; read instructions specified in the interview guide; provide materials, such as almanac, calculator, or tape recorder, required to perform certain tasks; tactfully move the respondent to the next task when he or she had spent too much time on one task; and record observations about the respondent's ability to complete the exercise booklet and about any problems that may have affected her or his performance.

Response Rates

A sampled individual could refuse to participate in the survey during any of the three phases of the data collection process; that is, during the administration of the screener, the background questionnaire, or the exercise booklet. The response rates presented below reflect the percentage of those who responded to each survey instrument, of those who had the opportunity to respond (Table B.1).



		11 <u>States</u>	This <u>State</u>
Instrument	National		
Screener	88.8	89.4	95.4
Background Questionnaire	81.9	79.9	86.2
Exercise Booklet	95.3	96.5	98.0

Data Collection Quality Control

Several quality control procedures were undertaken to ensure the integrity of the data collected. These included an edit by the interviewer, a complete edit of all documents by a trained editor, validation of 10 percent of each interviewer's completed (or "closed out") work, and observations by home office staff of interviewers conducting interviews and supervisors managing the data collection effort.

During the interviewer training session, interviewers were instructed on procedures for performing an edit of all data collection documents. The purposes of this edit were to catch and correct or explain any errors or omissions in recording, to learn from mistakes so they were not repeated, and to remove stray marks and completely fill in bubbles on the documents that were to be optically scanned.

In addition to this process, a complete edit was performed on all documents by trained editors. An item-by-item review was performed on each document, and each error was fully documented on an edit form. The supervisor reviewed the results of the edit with the interviewer during a weekly telephone conference.

Validation is the quality control procedure used to verify that an interview was conducted, at the correct address, and according to specified procedures, and to ensure that nonresponse statuses (e.g., refusals, vacancies, language problems) were accurately reported by the interviewers. Interviewers knew that their work would be validated but did not know which cases or which data items. A 10 percent subsample of dwelling units was selected and flagged in the supervisor's log and in the automated survey control system. The supervisors performed validation interviews by telephone if a phone number was available. Otherwise, validation was performed in person by the supervisor or by another interviewer.

Appendices 191

Field observations of both supervisors and interviewers were performed by Westat field management staff. One purpose of the interviewer observation was to provide home office staff with an opportunity to observe the effectiveness of the field procedures and monitor respondents' reactions to the survey. Another purpose was to provide feedback to weak interviewers when there was concern about their skills and/or performance. In addition to inperson observations, interviewers were required to tape record one complete interview and assessment. The field supervisor selected the particular case in advance and listened to the tape to "observe" each interviewer.

Finally, nine of the 24 supervisors were visited by field management staff and evaluated on their editing, coding, office organization, ability to maintain up-to-date records on production data, and supervision of interviewers.

Weighting

Weighting procedures were carried out by Westat, Inc. Full sample and replicate weights were calculated for each record to facilitate the calculation of unbiased estimates and their standard errors. The full sample and replicate weights for the household components were calculated as the product of the base weight for a record and a compositing and raking factor. Demographic variables critical to the weighting were recoded and imputed, if necessary, prior to the calculation of base weights. The recoded versions of these variables are not included in the file.

The base weight was calculated as the reciprocal of the final probability of selection for a respondent, which reflected all stages of sampling. The base weight was then multiplied by a compositing factor which combined the national and state component data in an optimal manner, considering the differences in sample design, sample size, and sampling error between the two components. Twelve different compositing factors were used, one for each of the 11 participating states, and a pseudo factor (equal to one) for all national component records from outside the 11 participating states. The product of the base weight and compositing factor for a given record was the composite weight. The records appropriate for a particular state analysis, therefore, include data from respondents age 16 years and older, although the inclusion of records for respondents over the age of 64 in state estimates significantly increases the sampling error of these estimates. Comparisons using data for adults age 65 and older should therefore be interpreted with caution. This caveat does not apply to national estimates, however, as all records for persons over the age of 64 come from the national component.

192 Technical Notes

The composite weights were raked so that several totals calculated with the resulting full sample weights would agree with the 1990 Census totals, adjusted for undercount. Raking, a procedure similar to poststratification, ensures that particular weighted estimates reach known control totals. Raking is used in place of poststratification when the full intersection of control totals is unavailable.

The cells used for the raking were defined to the finest combination of age, education level, race, and ethnicity that the data would allow. Raking adjustment factors were calculated separately for each of the 11 states and then for the remainder of the United States. The above procedures were repeated for 60 strategically constructed subsets of the sample to create a set of replicate weights to be used for variance estimation using the jackknife method. The replication scheme was designed to produce stable estimates of standard errors for national estimates as well as for the 11 individual state estimates.

The full sample and replicate weights for the incarcerated component were calculated as the product of the base weight for a record and a nonresponse and raking factor. The base weight was calculated as the reciprocal of the final probability of selection for a respondent, which reflected both stages of sampling. The base weights were then adjusted for nonresponse to reflect both facility and inmate nonresponse. The resulting nonresponseadjusted weights were then raked to agree with independent estimates for certain subgroups of the population.

Scoring the Exercise Booklets

As the first shipments of exercise booklets were received at ETS, photocopies were made of actual responses to the tasks. These sample responses were then scored by various staff, including the test developer and scoring supervisor, using either the scoring guides developed for the young adult tasks or guides prepared during the development of the new tasks. As the sample responses were scored, the scoring guides for the new tasks were adjusted to reflect the kinds of answers that the respondents were providing.

The sample papers were then used to train the group of readers who would score the exercise booklets. The purposes of the training were to familiarize the readers with the scoring guides and to ensure a high level of agreement among them. Each task and its scoring guide were explained, and sample responses representative of the score points in the guide were discussed. The readers then scored and discussed an additional 10 to 30 responses. After group training had been completed, all the readers scored all

Appendices 193

the tasks in more than 100 booklets to give them practice in scoring actual booklets, and to provide an opportunity to score more responses on a practice basis. A follow-up session was held to discuss responses that were given different scores by different readers. The entire training process was completed in about four weeks.

Twenty percent of all the exercise booklets were subjected to a reader reliability check, which entailed a scoring by a second reader. To prevent the second reader from being influenced by the first reader's scores, the first reader masked the scores in every fifth booklet he or she scored. These booklets were then passed to a second reader to score. When the second reader had scored every task, the first reader's scores were unmasked. The scoring supervisor reviewed each response that received discrepant scores from the two readers and discussed it with the readers involved.

The statistic used to report inter-reader reliability is the percentage of exact agreement — that is, the percentage of times that the two readers assigned a task precisely the same score. There was a high degree of inter-reader reliability across all the tasks in the survey, ranging from a low of 88.1 percent to a high of 99.9 percent, with an average agreement of 97 percent. For 133 out of the 166 open-ended tasks, the agreement was above 95 percent.

Data Entry

The background questionnaire was designed to be "read" (or processed) by a computerized scanning device. For most of the questions in this instrument, interviewers filled in scannable ovals next to the respondent's answers. Responses to open-ended items in the background questionnaire were translated into codes and the ovals filled in by Westat editors. During the check-in process at ETS, the screener coding was reviewed and documents were batched and sent to the scanning department on a regular basis. Exercise booklet scores were transferred to scannable documents by the readers who scored the items, and these were also batched and sent to the scanning department at regular intervals. The scanned data from the screeners, background questionnaires, and exercise booklets were transmitted to magnetic tape, which was then sent to the ETS computer center. As each of the different instruments was processed, the data were transferred to a database on the main



Editing and Quality Control

The editing procedures undertaken in this survey included an assessment of the internal logic and consistency of the data received. For example, data were examined for nonexistent housing locations or booklets, illogical or inconsistent responses, and multiple responses where single responses were requested. Where indicated, an error listing was generated and sent back to the processing area, where the original document was retrieved and the discrepancies were corrected wherever possible. For example, in the infrequent cases in which field personnel provided more than one response to a single-response background question, specific guidelines were developed to incorporate these responses consistently and accurately. If a conflict in the data could not be resolved, the information was left in the form in which it was received.

The background questionnaires were also checked to make sure that the skip patterns had been followed, and all data errors were resolved. Finally, a random set of booklets was selected to provide an additional check on the accuracy of transferring information from booklets and answer sheets to the database.

Scaling

The results from the National Adult Literacy Survey are reported on three scales established in the 1985 Young Adult Literacy Survey conducted as part of the National Assessment of Educational Progress: prose literacy, document literacy, and quantitative literacy. Using methods grounded in item response theory (IRT), the performance of a sample of examinees can be summarized on a series of scales even when different respondents have been administered different items. Conventional scoring methods are not suited for surveys such as this one. Specifically, statistics such as proportion of correct responses are inappropriate for surveys like the NALS and SALS, in which respondents receive different sets of items. Moreover, item-by-item reporting ignores patterns across items in various population subgroups. Finally, using average percent correct to estimate the proficiency mean of examinees within subgroups does not provide any other information about the distribution of skills among the examinees.

IRT scaling overcomes these limitations of traditional scoring methods. When several items require similar skills, the response patterns should have some uniformity. Such uniformity can be used to characterize both examinees and items in terms of a common scale, even when examinees receive different sets of items. Comparisons of items and examinees can then be made in

Appendices 195

reference to a scale, rather than to percent correct. IRT scaling also allows the performance distributions for various groups of examinees to be compared.

Scaling was carried out separately for each of the three domains of literacy (prose, document, and quantitative). The NAEP reading scale, used in the young adult survey, was dropped because of its lack of relevance to the current reading scale. The scaling model used for the national survey is the threeparameter logistic (3PL) model from item response theory.¹ It is a mathematical model for estimating the probability that a particular person will respond correctly to a particular item from a single domain of items. This probability is given as a function of a parameter characterizing the proficiency of that person, and three parameters characterizing the properties of that item.

Statistical Procedures

The statistical comparisons in this report were based on the t statistic. Generally, statistical significance is determined by calculating a t value for the difference between a pair of means, or proportions, and comparing this value to published tables of values at certain critical levels, called alpha levels. The alpha level is an a priori statement of the probability of inferring that a difference exists when, in fact, it does not.

The formula used to compute the t statistic was as follows: $t = (P_1 - P_2)/\sqrt{(se_1^2 + se_2^2)}$, where P_1 and P_2 are the estimates to be compared and se_1 and se_2 are their corresponding standard errors.

In order to make proper inferences and interpretations from the statistics, however, several points must be kept in mind. First, comparisons resulting in large t statistics may appear to merit special note. This is not always the case, because the size of the t statistic depends not only on the observed differences in means or the percentage being compared, but also on the standard error of the difference. Thus, a small difference between two groups with a much smaller standard error could result in a large t statistic, but this small difference is not necessarily noteworthy.

Second, when multiple statistical comparisons are made on the same data, it becomes increasingly likely that an indication of a population difference is erroneous. Even when there is no difference in the population, at an alpha level of .05, there is still a 5 percent chance of concluding that an observed t

value representing one comparison in the sample is large enough to be

¹ A. Birnbaum. (1968). "Some Latent Trait Models." In F.M. Lord and M.R. Novick, Statistical Theories of Mental Test Scores. Reading, MA: Addison-Wesley. F.M. Lord. (1980). Applications of Item Response Theory to Practical Testing Problems. Hillsdale, NJ: Erlbaum.

196 Technical Notes

statistically significant. As the number of comparisons increases, the risk of making such an error in inference also increases.

To guard against errors of inference based upon multiple comparisons, the Bonferroni procedure to correct significance tests for multiple contrasts was used. This method corrects the significance (or alpha) level for the total number of contrasts made with a particular classification variable. For each classification variable, there are $(K \cdot (K - 1))/2$ possible contrasts (or nonredundant pairwise comparisons), where K is the number of categories. The Bonferroni procedure divides the alpha level for a single t test (for example, .05) by the number of comparisons in order to give a new alpha that is corrected for the fact that multiple contrasts are being made.

Readers of this report are advised to use statistical tests of this nature to make their own comparisons and interpretations of the data reported herein.

Appendices 197

APPENDIX C

Participants in the Development Process

Literacy Definition Committee

Ms. Barbara Clark **Regional Manager Central Region** Los Angeles Public Library

Ms. Nancy Cobb Manager Human Resources Development Department Nabisco Biscuit Company

Ms. Hanna Fingeret Director Literacy South

Ms. Evelyn Ganzglass Director **Employment and Social Services Policy Studies** Center for Policy Research National Governors' Association

Mr. Ronald Gillum Director Adult Extended Learning Services Michigan Department of Education

Mr. Karl Haigler President The Salem Company

Mr. Carl Kaestle **Professor of Educational Policy Studies** Wisconsin Center for Educational Research University of Wisconsin

Mr. Reynaldo Macías (Liaison to the Technical Review Committee) Professor of Education and Director UC Linguistic Minority Research Institute University of California, Santa Barbara

Appendices 199

Mr. David Neice Director of Research and Analysis Directorate Department of the Secretary of State Canada

Honorable Carolyn Pollan (ex-officio member) State Representative Arkansas State Legislature

Ms. Lynne Robinson **Director of Support Services** Division of ACE Sweetwater Union High School District

Mr. Anthony Sarmiento Director **Education Department** AFL-CIO

Ms. Gail Spangenberg Vice President and Chief Operating Officer **Business Council for Effective Literacy**

Technical Review Committee

Ms. Susan Embretson Professor Department of Psychology University of Kansas

Mr. Jeremy Finn Professor Graduate School of Education **SUNY Buffalo**

Mr. Robert Glaser Director Learning Research and Development Center University of Pittsburgh

Mr. Ronald Hambleton Professor School of Education Laboratory of Psychometric and Evaluative Research University of Massachusetts

Mr. Huynh Huynh Professor Department of Educational Psychology University of South Carolina at Columbia

Ms. Sylvia Johnson Professor Howard University

200 Participants in the Development Process
Mr. Frank Schmidt Professor Industrial Relations and Human Resources College of Business University of Iowa

Mr. Richard Venezky (Liaison to the Literacy Definition Committee) Professor Department of Educational Studies University of Delaware

Test Development Consultants

Ms. Valerie de Bellis Center for Mathematics, Science, and Computer Education Rutgers University

Mr. John Dawkins Language and Literature Department Bucks County Community College

Ms. Harriet L. Frankel Secondary and Higher Education Programs Educational Testing Service

Ms. Bonnie Hole The Bureau of Evaluation and Student Assessment Connecticut State Department of Education

Mr. Richard Lesh Division of Cognitive and Instructional Science Educational Testing Service

Ms. Ave M. Merritt

Secondary and Higher Education Programs Educational Testing Service

Mr. Peter Mosenthal Reading and Language Arts Center Syracuse University

Ms. Pam Smith Secondary and Higher Education Programs Educational Testing Service

Ms. Wallie Walker-Hammond Secondary and Higher Education Programs Educational Testing Service

Appendices 201

Acknowledgments



VV e extend our deep appreciation to the many individuals who contributed to this project. These acknowledgments cannot begin to repay the debt of thanks we owe.

Above all, we thank John Hartwig, Iowa Department of Education, who coordinated the Iowa State Adult Literacy Survey. This work benefited immeasurably from his commitment and involvement. We are also grateful to Hal Beder, Rutgers—The State University, New Brunswick, New Jersey, for providing helpful guidance.

We acknowledge the members of the Literacy Definition Committee and the Technical Review Committee, who guided the project throughout. We also extend thanks to our colleagues at Westat, who managed sampling, data collection, and composite weighting for the national and state surveys.

The State Adult Literacy Survey project depended on the contributions of many individuals within Educational Testing Service. Doug Rhodes orchestrated the project, assisted by Cathy Shaughnessy. Anne Campbell led test development and scoring activities; Dave Hobson provided financial wisdom; and Jules Goodison guided the operations process, assisted by Debbie Giannacio.

Special thanks go to Norma Norris, Bruce Kaplan, Jim Ferris, and Jennifer Nelson, who conducted the statistical work and data analyses under tight deadlines. Don Rock and Kentaro Yamamoto directed statistical and psychometric aspects of the project. Puff Rice deserves both thanks and cheers for editing the entire stack of state report manuscripts. Thanks are due to Mary Michaels for coordinating administrative aspects of the project, and to Beverly Cisney for assisting with desktop publishing. We also gratefully acknowledge the ETS Publications Division — in particular, Peter Stremic, Kathy Benischeck, Fred Marx, Kiyo Toma, Robin Matlack, and Joyce Martorelli for designing and producing the state reports. Finally, we wish to thank the thousands of adults who gave their time to

respond to the survey.

Lynn B. Jenkins Irwin S. Kirsch

202 Participants in the Development Process

About the Authors

Lynn B. Jenkins is a program administrator in the Literacy Learning and Assessment Group at Educational Testing Service.

Irwin S. Kirsch is project director of the National Adult Literacy Survey and executive director of the Literacy Learning and Assessment Group at Educational Testing Service.

Appendices 203

